

University of Pennsylvania ScholarlyCommons

Publicly Accessible Penn Dissertations

2015

Hospital Nursing Linked to Readmissions Following Total Hip and Knee Arthroplasty

Karen Blanchette Lasater University of Pennsylvania, karen.b.lasater@gmail.com

Follow this and additional works at: https://repository.upenn.edu/edissertations

Part of the Nursing Commons

Recommended Citation

Lasater, Karen Blanchette, "Hospital Nursing Linked to Readmissions Following Total Hip and Knee Arthroplasty" (2015). *Publicly Accessible Penn Dissertations*. 1080. https://repository.upenn.edu/edissertations/1080

This paper is posted at ScholarlyCommons. https://repository.upenn.edu/edissertations/1080 For more information, please contact repository@pobox.upenn.edu.



Hospital Nursing Linked to Readmissions Following Total Hip and Knee Arthroplasty

Abstract

Hospital readmissions pose negative health risks for older adults and reflect low quality, high cost healthcare. Efforts to reduce readmissions have focused on disease-specific interventions that target patients during the transition from the hospital or in the post-acute care setting. Less attention has spotlighted the role of hospital nursing. Staff nurses represent an around-the-clock surveillance system that is well-positioned to recognize patients' physical and social needs that may contribute to a readmission. This cross-sectional secondary data analysis explored the association between the working conditions of hospital nurses - staffing and the practice environment - and 30 day readmissions among older adults following elective total hip and total knee arthroplasty. Data sources from 2005-2006 included patient administrative data, nurse survey data, and hospital organizational data. Nurse survey responses were aggregated to construct hospital measures of nurse staffing and the practice environment. The main outcome of interest was 30 day unplanned readmission. The analytic sample consisted of 112,018 Medicare patients electively undergoing either total hip or total knee arthroplasty. and 23,089 registered nurses working in 495 acute care hospitals in four states (CA, FL, NJ, PA). The sample was descriptively analyzed using cross-tabulations, Kaplan-Meier plots, and histograms. Multivariable logistic regressions estimated the effect of nurse staffing and the practice environment on 30 day readmission, adjusting for patient and hospital covariates and accounting for clustering of patients within hospitals. The 30 day unplanned readmission rate was 5.6% and 5.7% for hip and knee patients, respectively. After adjusting for patient and hospital covariates, each additional patient was associated with an 8% increase in the patient's likelihood of readmission. Patients cared for in a better practice environment, as compared to a mixed or poor environment, had 13% lower odds of readmission; however, this relationship became insignificant once adjusting for nurse staffing. These findings reveal that a substantial percentage of older adults experience an unplanned rehospitalization following elective total hip and total knee arthroplasty. The evidence suggests that improving nurse staffing and the practice environment may be strategies for reducing readmissions among older adult orthopedic surgical patients.

Degree Type

Dissertation

Degree Name Doctor of Philosophy (PhD)

Graduate Group Nursing

First Advisor Matthew D. McHugh

Keywords arthroplasty, hospital, joint replacement, nursing, readmissions

Subject Categories Nursing

This dissertation is available at ScholarlyCommons: https://repository.upenn.edu/edissertations/1080



HOSPITAL NURSING LINKED TO READMISSIONS

FOLLOWING TOTAL HIP AND KNEE ARTHROPLASTY

Karen B. Lasater

A DISSERTATION

in

Nursing

Presented to the Faculties of the University of Pennsylvania

in

Partial Fulfillment of the Requirements for the

Degree of Doctor of Philosophy

2015

Supervisor of Dissertation

Matthew D. McHugh, PhD, RN Associate Professor of Nursing

Graduate Group Chairperson

Connie Ulrich, PhD, RN Professor of Nursing

Dissertation Committee:

Linda H. Aiken, PhD, RN Professor of Nursing and Sociology

Eileen V. Lake, PhD, RN Associate Professor of Nursing and Sociology



HOSPITAL NURSING LINKED TO READMISSIONS FOLLOWING TOTAL HIP AND KNEE ARTHROPLASTY

COPYRIGHT

2015

Karen Blanchette Lasater



DEDICATION

To my parents, who instilled a love for learning To my husband, who inspired a dream

"Far and away the best prize that life has to offer is the chance to work hard at work worth doing" -Theodore Roosevelt



ACKNOWLEDGEMENT

This dissertation is the culmination of years of mentorship and support from a community of scholars, colleagues, and loved ones. First, I would like to acknowledge my dissertation chair, Dr. Matthew McHugh, who provided me with a balance of autonomy and guidance throughout the dissertation process. His generosity with his time and expertise has substantially contributed to my thinking and this work.

Secondly, I am grateful for the opportunity to be a part of the Center for Health Outcomes and Policy Research (CHOPR) at the University of Pennsylvania, and would like to acknowledge Dr. Linda Aiken and all of the faculty and staff who have made this opportunity possible. Dr. Aiken has been a generous mentor with a sincere focus on growing a future generation of nurse health services researchers. I am appreciative of the continuous attention and support of Dr. Eileen Lake, who has been central to my development. Her thoughtful critiques have impacted not only the development of my dissertation, but also who I am as a researcher and an academic.

This dissertation is possible because of the faculty and staff at CHOPR. Thank you especially to my dissertation readers, Dr. Margo Brooks Carthon and Dr. Nancy Hanrahan, for your contributions. Thank you, Tim Cheney, for your statistical support and expertise. I would also like to acknowledge that the parent study used in this dissertation is supported by the National Institute of Nursing Research R01-NR0451 grant – an award that reflects the work of both current and past students and faculty at CHOPR.

Finally, I am indebted to my colleagues and friends at CHOPR and across the School of Nursing, who made this experience so wonderful. And to my family, and



iv

especially my husband, who has unconditionally supported me in every step of the journey.



ABSTRACT

HOSPITAL NURSING LINKED TO READMISSIONS FOLLOWING TOTAL HIP AND KNEE ARTHROPLASTY

Karen B. Lasater

Matthew D. McHugh

Hospital readmissions pose negative health risks for older adults and reflect low quality, high cost healthcare. Efforts to reduce readmissions have focused on disease-specific interventions that target patients during the transition from the hospital or in the postacute care setting. Less attention has spotlighted the role of hospital nursing. Staff nurses represent an around-the-clock surveillance system that is well-positioned to recognize patients' physical and social needs that may contribute to a readmission. This crosssectional secondary data analysis explored the association between the working conditions of hospital nurses – staffing and the practice environment – and 30 day readmissions among older adults following elective total hip and total knee arthroplasty. Data sources from 2005-2006 included patient administrative data, nurse survey data, and hospital organizational data. Nurse survey responses were aggregated to construct hospital measures of nurse staffing and the practice environment. The main outcome of interest was 30 day unplanned readmission. The analytic sample consisted of 112,018 Medicare patients electively undergoing either total hip or total knee arthroplasty, and 23,089 registered nurses working in 495 acute care hospitals in four states (CA, FL, NJ, PA). The sample was descriptively analyzed using cross-tabulations, Kaplan-Meier plots, and histograms. Multivariable logistic regressions estimated the effect of nurse staffing and the practice environment on 30 day readmission, adjusting for patient and hospital



vi

covariates and accounting for clustering of patients within hospitals. The 30 day unplanned readmission rate was 5.6% and 5.7% for hip and knee patients, respectively. After adjusting for patient and hospital covariates, each additional patient was associated with an 8% increase in the patient's likelihood of readmission. Patients cared for in a better practice environment, as compared to a mixed or poor environment, had 13% lower odds of readmission; however, this relationship became insignificant once adjusting for nurse staffing. These findings reveal that a substantial percentage of older adults experience an unplanned rehospitalization following elective total hip and total knee arthroplasty. The evidence suggests that improving nurse staffing and the practice environment may be strategies for reducing readmissions among older adult orthopedic surgical patients.



Table of Contents

Table of Contents	viii
List of Tables	x
List of Figures	xii
CHAPTER 1: INTRODUCTION	
The problem	1
Background	4
Significance	8
Study overview	10
Specific aim	10
Gaps	11
Summary	11
CHAPTER 2: BACKGROUND AND SIGNIFICANCE	
Introduction	13
Conceptual framework	13
Background	17
Significance	43
Outcome	46
Covariates	47
Summary	48
CHAPTER 3: METHODS	
Introduction	49
Overview	49
Parent study	50
Data sets	52
Sample	55
Measurement and instrument	59
Statistical analysis	66 72
Data integrity plan	/3
Human subjects	13
CHAPTER 4: RESULTS	
Introduction	75
Characteristics of the sample	75
Analysis of specific aim	93
Sensitivity analysis	103
Summary	103
CHAPTER 5: DISCUSSION	
Introduction	105



viii

Main findings	105
Limitations	110
Implications	112
Future research	117
Conclusion	119

APPENDIX

REFERENCES



List of Tables

Table 3.1 Readmission penalties to be applied in fiscal year 2015, by state	56
Table 3.2 Inclusion criteria and rationale for the study of elective THA/TKA	
readmissions	58
Table 3.3 Exclusion criteria and rationale for the study of elective THA/TKA	
readmissions	59
Table 4.1 Patient characteristics, by surgery type	76
Table 4.2 Common comorbidities of study patients, by surgery type	77
Table 4.3 Ten most common reasons for 30 day readmission, by surgery type	79
Table 4.4 Thirty day readmission by patient characteristics and by surgery type	81
Table 4.5 Characteristics of the readmission event, by surgery type	82
Table 4.6 Distribution of the hospital and patient study sample by hospital characteris	tics
	84
Table 4.7 Thirty day readmission by hospital characteristics and surgery type	87
Table 4.8 Distribution of the hospital and patient study sample by nursing characterist	tics
	89
Table 4.9 Thirty day readmission by nursing characteristics and surgery type	90
Table 4.10 Correlation matrix of hospital and nursing characteristics, hospital level	92
Table 4.11 Effects of hospital nursing on 30 day readmission, in hospitals with at leas	t
10 live THA/TKA discharges ($N = 112,018$)	95
Table 4.12 Effects of hospital nursing on 30 day readmission, in hospitals with at leas	t
50 live THA/TKA discharges ($N = 108,906$)	97
Table 4.13 Effects of hospital nursing on 30 day readmission, in hospitals with at leas	t
100 live THA/TKA discharges ($N = 103,080$)	97
Table 4.14 Effects of hospital nursing on 30 day readmission, 4 subscale practice	
environment (N = $112,018$)	99
Table 4.15 Effects of hospital nursing on 10 day readmission, in hospitals with at leas	t
10 live THA/TKA discharges ($N = 112,018$)	101
Table 4.16 Effects of hospital nursing on 10 day readmission, in hospitals with at leas	t
50 live THA/TKA discharges ($N = 108,906$)	101



Х

Table 4.17 Effects of hospital nursing on 10 day readmission, in hospitals with at le	east
100 live THA/TKA discharges ($N = 103,080$)	102
Table A.1 Patient level sensitivity analysis of the effects of hospital nursing on 30 c	lay
readmission ($N = 112,018$)	127
Table A.2 Hospital level sensitivity analysis of the effects of hospital nursing on 30) day
readmission ($N = 112,018$)	130
Table A.3 Hospital surgical care improvement project measure sensitivity analysis	of the
effects of hospital nursing on 30 day readmission ($N = 112,018$)	131
Table A.4 Nursing characteristics by patient race	133
Table A.5 Patient characteristics by patient race	133
Table A.6 Nursing characteristics by patient socioeconomic status	134



List of Figures

Figure 2.1 Donabedian's structure process outcome model	14
Figure 2.2 Theoretical framework adapted from the Quality Health Outcomes Model	15
Figure 3.1 Planned and unplanned readmission algorithm	61
Figure 3.2 Diagram of data sources and linkages	67
Figure 4.1 Time-to-readmission for THA/TKA patients who were readmitted, by surg	gery
type	83
Figure 4.2 Distribution of THA/TKA patients across hospitals	85
Figure 4.3 Readmission rate of THA/TKA patients across hospitals	85
Figure 4.4 Distribution of registered nurses across study hospitals	88
Figure A.1 Inclusion criteria for the study of elective THA/TKA readmissions	120
Figure A.2 Exclusion criteria for the study of elective THA/TKA readmissions	121



CHAPTER 1: INTRODUCTION

"The hospital environment should enable patients not disable, and expect their staff do the same. I didn't realize the importance of staying active and not laying in bed the whole time. Do the people who work in the hospital know this?"

– Anonymous patient (Boltz et al, 2010)

The problem

Half of all hospitalized older adults experience profound negative consequences for their health and quality of life, above and beyond the reason for hospitalization (Boltz et al, 2012). A common outcome of hospitalization for older adults includes functional decline, which can onset as early as 48 hours from admission and result in increased morbidity and mortality (Boltz, Capezuti, & Shabbat, 2011; Boltz et al, 2012; Kortebein et al, 2008). Physical deconditioning is associated with prolonged recovery, with many patients never regaining pre-hospitalization functional capacity (Covinsky et al, 2003; Ponzetto et al, 2003). Each time an older adult is hospitalized, there is an increased probability for rehospitalization and a decreased likelihood of living independently in the future (Boltz, Capezuti, & Shabbat, 2011; Kortebein et al, 2008; Ponzetto et al, 2003). With one in every five older adults readmitted within 30 days after an initial hospitalization, preventing avoidable readmissions has the potential to improve the overall health and quality of life for older adults (Jencks, Williams, & Coleman, 2009; RWJF, 2013).

Preventing readmissions has been a focus of clinicians' efforts for decades (Achté & Apo, 1967; Jenkins, Bermiss, & Lorr, 1953; Strauss et al, 1974; Wing, Denham, & Munro, 1959). However, with the enactment of the Affordable Care Act and the Hospital Readmission Reduction Program, initiatives to reduce readmissions have come to the forefront of public awareness. Today, readmissions are targeted by the Centers for



Medicare and Medicaid Services (CMS) and others, as an indicator of hospital quality. Hospitals are being held financially accountable for excessive readmissions.

Medicare – the primary third party payer for older adults – spends approximately one quarter of its annual outlay on inpatient hospital services (CBO, 2014). Of the \$139 billion spent on hospital services in 2013, Medicare spent approximately \$26 billion on readmissions (RWJF, 2013), \$17 billion of which are potentially avoidable (CBO, 2014; RWJF, 2013). Avoidable readmissions account for more than 12% of Medicare inpatient spending (CBO, 2014).

Since the Hospital Readmission Reduction Program was enacted, researchers have increased attention to understanding the incidence and risk factors for readmission, and to identifying strategies to reduce readmissions. While the causes of readmission are multifactorial and span from individual patient characteristics to environmental and system-level contributors, only a small portion of research has focused on the role of the most numerous healthcare providers – inpatient nurses. Hospital nurses' proximity to patients and their care provision in the hospital make nurses instrumental in understanding the complexity of patient needs and well-positioned to identify and address issues that may precipitate a readmission following discharge.

Specifically, little is known about the role of hospital nurses, such as staffing and the practice environment, on reducing readmissions. Hospital nurses provide around-theclock care at the patient's bedside, which positions nurses as the primary surveillance system to identify and intervene on early warning signs or changes in patients' clinical condition. When nursing care is stressed due to unmanageable staffing ratios, limited time resources prevent the nurse from providing complete and thorough care and



surveillance. The practice environment describes the social context where nurses provide care. Early studies suggest more favorable nursing characteristics are associated with reduced odds of readmission for some medical and surgical conditions (Ma, McHugh, & Aiken, 2015; McHugh & Ma, 2013).

However, to date, little is understood about how hospital nurses influence readmissions in older adults undergoing total hip and total knee arthroplasty (THA/TKA). THA/TKA is a common surgical treatment for arthritis, which is caused in part by normal wear and tear on joints related to aging and can be exacerbated by undue stress on joints due to obesity. Given an aging baby boomer generation and the rising incidence of obesity, the numbers of THA/TKA procedures are likely to increase over time (Kurtz et al, 2005). Approximately two-thirds of the 285,000 THA and 600,000 TKA that are performed annually in the United States, are paid for by Medicare (AAOS, 2014; Ong et al, 2006). These surgeries account for Medicare's largest procedural cost (Bozic et al, 2008). Estimates suggest that the 30 day readmission rate following THA/TKA is approximately 5%, in the Medicare population (Suter et al, 2014).

Readmissions following this common surgical procedure have gained increased attention among patients, third party payers, and providers, alike. Beginning in 2015, preventable readmissions following elective THA/TKA became the first surgical procedures to be targeted for financial penalization for hospitals that have worse than expected readmission rates (PPACA, 2010). As provider reimbursement becomes increasingly linked to patient outcomes, providers have a growing incentive to utilize evidence-based strategies to reduce readmissions. This study examines the association



www.manaraa.com

between modifiable hospital nursing structural characteristics – staffing and the practice environment – and readmissions in older adults following elective THA/TKA.

Background

Nurses are the most numerous healthcare providers and spend a significant amount of time directly caring for hospitalized patients. Staff nurses are responsible for providing, overseeing, and coordinating the care of patients. For many hospitalized patients, nursing care includes assessment and management of signs and symptoms of the disease process and response to treatments, coordination of care between providers both inside and outside of the hospital, administration of medications and treatments, and education of the patient and their caregivers about self-care practices and disease management.

Older adults undergoing major orthopedic surgery require intensive care, particularly in the postoperative period. In the immediate postoperative period, patients are at risk for serious complications including infection, blood clots, pulmonary emboli, falls and prosthetic dislocation. During the days and weeks following surgery, patients require mobility assistance, physical rehabilitation, wound management, and ongoing education about self-care following a major joint replacement. For these patients, the role of the hospital staff nurse extends beyond the care provided in the hospital to include coordinating follow-up care after discharge and ensuring patients can safely transition to independent self-care activities.

Nurses' essential role in patient care is borne out by the fact that approximately one-third of inpatient hospital spending is allocated to nursing staff (Kane & Siegrist, 2002). It is well established that nurses are vital healthcare providers in ensuring the safe



and effective delivery of high quality healthcare. However, due to high patient demands and a lack of time, hospital staff nurses are often forced to prioritize patient care needs in such a way that patients frequently and unknowingly forfeit necessary nursing care (Kalisch, 2006; Kalisch, Gay & Williams, 2009). One recent study linked missed nursing care to an increased odds of patient readmission (Brooks Carthon et al, 2015).

Nurse-driven interventions like care coordination, patient education, and discharge planning are believed to effectively reduce readmissions (Coleman et al, 2006; Jack et al, 2009; Naylor et al, 2004; Naylor et al, 2011). However, to date, such evidence has largely been based on research pertaining to disease-specific nurse-initiated interventions, such as telephone follow-up for patients with congestive heart failure (Dudas et al, 2001; Wheeler & Waterhouse, 2006; Woodend et al, 2003), nurse practitioner-led transitional care models, which involve a nurse practitioner who is devoted to coordinating patient care across settings (Naylor et al, 2004; Naylor et al, 2011), and other team-based approaches focused on the hospital discharge process (Coleman et al, 2006; Jack et al, 2009; Naylor et al, 2011). While the effectiveness of many of these interventions have been demonstrated in randomized control trials (Ahmed, 2002; Coleman et al, 2006; Dudas et al, 2001; Jack et al, 2009; Naylor et al, 2004; Naylor et al, 2006; Dudas et al, 2001; Jack et al, 2009; Naylor et al, 2004; Naylor et al, 2004; Naylor et al, 2004; Naylor et al, 2004; Naylor et al, 2006; Dudas et al, 2001; Jack et al, 2009; Naylor et al, 2004; Naylor et al, 2001; Jack et al, 2009; Naylor et al, 2004; Naylor et al, 2004; Naylor et al, 2011; Wheeler & Waterhouse, 2006; Woodend et al, 2003), their effectiveness is less clear once implemented in practice (Hansen et al, 2011).

One hypothesis for this dissonance is the additional demands these interventions place on hospital nurses. These additional interventions may detract from the essential care that nurses deliver at the patient's bedside – care which is commonly missed due to lack of time (Kalisch, 2006; Kalisch, Gay & Williams, 2009). In addition to the lack of



evidence that disease-specific readmission reduction approaches are effective in practice, they require additional interventions beyond essential nursing care, and are overly narrow in focus; thus begging the question if there is a more effective and efficient way to reduce readmissions among at-risk populations (Dharmarajan et al, 2013).

Interventions that require additional providers or place additional demands on staff nurses should be considered a supplement, rather than a substitute for underlying systemic problems with care delivery. In other words, hospital nurse work conditions, including staffing and the practice environment, should be considered the essential building blocks to which these population and disease-specific interventions can be added. Every hospitalized patient receives nursing care; while only some patients receive targeted readmission reduction interventions.

System level investments in hospital nursing are associated with positive health outcomes. Additionally, increasing nurse staffing simultaneously improves various health outcomes for all hospitalized patients cared for by a nurse – unlike disease-specific approaches, which are narrowly focused on target populations. A system level approach to promoting positive health outcomes is supported by decades of research that demonstrate an association between more favorable staffing ratios, better practice environments, and more educated nursing staff and better outcomes. A large cross-sectional analysis of retrospective secondary data of general, orthopedic, and vascular surgery patients demonstrated that each additional patient per nurse was associated with 7% higher odds of mortality and failure-to-rescue – a measure of death in patients who experience a complication – within 30 days of admission (Aiken, Clarke, Sloane, Sochalski, & Silber, 2002). A cross-sectional secondary data analysis of a similar patient



www.manaraa.com

population in Pennsylvania hospitals found that patients cared for in a less favorable work environment had higher odds of mortality and failure-to-rescue. These findings suggest that hospitals exhibiting signals of nursing excellence, as measured by nurses' positive assessment of their work environment, are associated with better outcomes of care (Aiken et al, 2008). The proportion of hospital nurses with a baccalaureate degree has also been demonstrated to be associated with lower odds of mortality and failure-torescue. In a study of surgical patients, odds of mortality and failure-to-rescue decreased 5% for each additional 10% increase in the proportion of nurses within a hospital holding a baccalaureate degree (Aiken et al, 2003). These seminal studies are bolstered by numerous other studies across various patient populations (Kane et al, 2007; Kazanjian et al, 2005; Shekelle, 2013).

Far fewer studies have examined the effect of nursing on 30 day readmissions, yet early studies suggest that hospital nursing matters. For Medicare patients with heart failure, myocardial infarction, or pneumonia, one study demonstrated that the odds of 30 day readmission decreased for patients cared for by nurses working in more favorable practice environments and with better staffing ratios (McHugh & Ma, 2013). These findings were consistent in a study of general, vascular, and orthopedic surgical patients (Ma, McHugh, & Aiken, 2015). An analysis of nursing hours per patient day conducted at the unit level within a single institution showed that increasing investments in nurse staffing could be cost-beneficial through reductions in post-discharge hospital utilization (Weiss, Yakusheva, & Bobay, 2011). Increasing staffing in less well-staffed hospitals has the potential to reduce readmissions while being cost-beneficial to the hospital (Weiss, Yakusheva, & Bobay, 2011). These findings did not account for the CMS penalties for



worse than expected readmission rates, suggesting that cost savings could be greater than estimated in the study.

Significance

Understanding if and to what extent hospital nursing can influence patient readmission is significant for multiple stakeholders. With one in five discharged Medicare patients rehospitalized within 30 days (Jencks, Williams, & Coleman, 2009; RWJF, 2013), avoidable readmissions have become widely recognized as a signal of high cost, low quality care. Estimates suggest that with over 9 million Medicare hospitalizations annually, Medicare could realize \$17 billion in cost savings through eliminating avoidable readmissions (RWJF, 2013).

The Hospital Readmission Reduction Program is one of many programs in the Affordable Care Act, which constitute a new reimbursement paradigm – pay-forperformance. Pay-for-performance is a hypernym and umbrella term for the initiatives of the Affordable Care Act that aim to improve healthcare value through high quality outcomes at lower cost (James, 2012). For example, the Hospital Readmission Reduction Program allows CMS to financially penalize hospitals with worse than expected readmission rates, by withholding a fraction of the hospital's annual base CMS reimbursements. Other such programs are known as value-based purchasing, which gives CMS authority to incentivize hospitals, through financial penalties, to perform well on an a priori set of performance and outcome measures (James, 2012). Third party payers, beginning with CMS, are closely monitoring various indicators of hospital quality and withholding payment for poor performance and outcomes of care.



www.manaraa.com

Understanding how hospital nursing impacts patient outcomes can help inform hospital administrators about the value of additional investments and the consequences of reduced investments in nursing. Hospitals have been known to make cost reductions to nursing resources during financially stressful times (Aiken, 2008), a practice that may increase in frequency as hospital budgets are further subjected to CMS reimbursement withholdings for poor performance on quality measures. With more robust evidence about the impacts of more favorable nurse staffing ratios and better practice environments, hospital administrators are better positioned to make a social and financial case for strategic investments in nursing.

This findings from this study may also have state and federal policy implications. More evidence on the far reaching effects of more favorable staffing ratios on numerous patient outcomes can help to substantiate policies that support better working conditions for nurses. One way policy can influence better nurse staffing in hospitals is through mandated staffing ratios. California is the first and only state to have implemented staffing laws, yet other states are considering similar legislation. Among the states analyzed in this study, California has the best staffing ratios, the lowest percent of hospitalized penalized, and the lowest average penalty (Table 3.1). Building a more robust evidence-base can support other states to more effectively lobby for safe staffing policies, such as mandated staffing ratios.

Additionally, readmission and the negative human consequences of hospitalization are important to patients and families, alike. Although the recent attention to readmissions has largely been driven by national policy efforts to reduce readmissions in an effort to drive down unnecessary spending, the human consequences of readmission



are profound. Functional decline and the potential for loss of independent living are concerning outcomes for older adults (Boltz, Capezuti, & Shabbat, 2011; Kortebein et al, 2008; Ponzetto et al, 2003). Today, there is public reporting of hospital performance on health outcomes through the Hospital Compare website. The website is publicly accessible and reports hospital readmission rates, among other quality indicators. Finally, healthcare providers assume an ethical responsibility for providing the best care possible for each individual, which makes understanding predictors of readmission an imperative for healthcare providers and the populations they serve.

Study overview

This study is a cross-sectional analysis of secondary retrospective data that used multiple linked datasets to build on existing literature about the effects of hospital nursing – staffing and the practice environment – on reducing the likelihood of readmission 30 days after discharge following elective THA/TKA

Specific aim

Aim: To examine the extent to which hospital nursing – staffing and the practice environment – are associated with odds of 30 day readmission in a Medicare population undergoing elective total hip and total knee arthroplasty.

Hypothesis 1: Patients undergoing elective total hip and total knee arthroplasty in hospitals with more favorable nurse staffing ratios and better practice environments will have lower odds of 30 day readmission, after controlling for patient and hospital characteristics.



Gaps

This study addressed a number of gaps in the existing literature. This is the first study to examine the impact of nursing on readmission in an older adult population undergoing elective THA/TKA. Of principal importance is the impact that this study's findings can have on the health outcomes of hospitalized patients. Prior research on reducing readmissions has offered disease-specific interventions, which are often narrowly targeted to a subset of the population. Attention to the structural characteristics of hospital nursing, such as staffing and the practice environment, may have the potential to impact health outcomes in this orthopedic surgical population and other hospitalized patients.

This study extends beyond describing readmissions among older adults undergoing THA/TKA and offers evidence for actionable strategies that hospitals can employ to reduce readmission rates among their vulnerable patients. Specifically, this study builds on research by Ma and colleagues (2015) to analyze older adults undergoing a particular orthopedic procedure that has timely policy and financial relevance for hospitals and insurers, alike.

Summary

Chapter 1 provided an introductory discussion of the questions to be explored in this study – the association of hospital nursing with 30 day readmission in a Medicare population undergoing elective THA/TKA. The background and significance section briefly explored the literature in this area and the social, financial and political importance of reducing readmissions following hospitalization. This study was a cross-



sectional analysis of retrospective data using multiple linked datasets to explore the association of nurse reports of staffing and practice environment with readmission.

Chapter 2 describes the conceptual framework underlying the study and explores the empirical and clinical context that will be enhanced by the knowledge developed from this study. This chapter provides a review of the literature related to the effects of nursing on readmission for various patient populations, as well as a review of the literature on readmission for patients undergoing elective THA/TKA. The study outcomes and covariates are defined.

Chapter 3 describes the methodology for the study, including a description of the datasets and parent study, clearly delineated sample criterion, as well as an explanation of the statistical analytic plan for the specific aim. The measurement approach for the outcomes and covariates used in the study are described. In this section, there is a description of the Practice Environment Scale of the Nurse Work Index (PES-NWI) – an instrument used for measuring the practice environment. Lastly, Chapter 3 includes a brief discussion of the methodological limitations and assumptions, a data integrity plan, and issues related to human subjects including Institutional Review Board requirements.

Chapter 4 describes the findings of the study including descriptions of the patients, nurses, and hospitals, the results from the analysis of the specific aim and an extensive sensitivity analysis. Finally, Chapter 5 provides a discussion of the main findings in the context of previous literature findings. Limitations of the study, as well as implications for policy, practice, and future research are described.



CHAPTER 2: BACKGROUND AND SIGNIFICANCE

Introduction

Chapter 2 sets the context for the study. The conceptual framework, the Quality Health Outcomes Model, is introduced and its adaptation for use in this study is explained. Next, an integrated review of prior literature related to the associations of nursing and readmissions among various inpatient populations and readmissions among older adult patients undergoing THA/TKA are explored. Substantive gaps in knowledge that this study and future studies could address are identified. The social, financial, and political implications of the hypothesized study findings are described, lending credence to the significance of the study. Finally, the outcomes and covariates of interest are specified.

Conceptual framework

The Quality Health Outcomes Model provides the conceptual framework for this study. This model describes the relationships between four concepts: system, intervention, client, and outcome (Mitchell, Ferketich, & Jennings, 1998). The Quality Health Outcomes Model has previously served as an important framework in seminal nursing health services research (Mitchell & Lang, 2004).

The Quality Health Outcomes Model has iteratively emerged from another framework, Donabedian's structure process outcome model (Figure 2.1). Developed to evaluate and study the quality of care, Donabedian's model is composed of three dimensions: structure, process, and outcome (Donabedian, 1966). Each dimension consecutively affects the succeeding dimension (Donabedian, 1966). Structure is defined by the setting in which the process of care occurs. Structure includes variables such as



hospital size and teaching status. Nursing related structure concepts, for example, include staffing and the practice environment. The linear nature of Donabedian's structure process outcome model and the successive alignment of the three dimensions assume that the quality of the structure prescribes the quality of the outcome. For example if nurse staffing (structure) is poor, then the succeeding dimensions - delivery of care (process) and readmission (outcome) – will likely also be poor.

The process dimension is the dimension in which the act or practice of caregiving occurs and is concerned with *how* the care is delivered. The process dimension includes concepts such as how information is communicated between patient and provider and which interventions are performed. The process dimension directly affects the outcome (Donabedian, 1966). Outcomes can be either negative, such as readmission, or positive, such as patient satisfaction.

Figure 2.1. Donabedian's structure process outcome model



Although it is a commonly used model in the evaluation of healthcare delivery and quality, the linear and unidirectional nature of Donabedian's structure process outcome model may not accurately reflect the complexities of healthcare delivery. The process dimension encompasses multiple variables that require unpacking in order to appreciate the complexity of care. Such deficits of this model have aided in the development of another framework, the Quality Health Outcomes Model.



Unlike Donabedian's structure process outcome model, the Quality Health Outcomes Model contains complex relationships using bi-directional arrows between concepts to represent the interrelationships in healthcare delivery (Figure 2.2). This complexity allows researchers to more readily test theoretical relationships, including patient outcomes that are sensitive to both nursing interventions and system characteristics. This has made the Quality Health Outcomes Model a popular theoretical model among some nurse health services researchers (Mitchell & Lang, 2004).

Figure 2.2. Theoretical framework adapted from the Quality Health Outcomes Model



The system is a concept that describes a range of organizational attributes, such as nurse staffing and hospital teaching status. In this study, multiple system attributes were explored. Figure 2.2 depicts how the Quality Health Outcomes model was adapted for use in this study. Within each concept, the variables of interest are listed. The primary



predictors of interest explored included nurse staffing and the practice environment. Hospital organizational features explored included: bed size, teaching status, and technology status. Other system level organizational attributes were explored in an extensive sensitivity analysis, which is detailed later in this chapter.

In the Quality Health Outcomes Model, the system concept is bi-directionally related to other concepts. Unlike the unidirectional arrows in Donabedian's structure process outcome model, the bi-directional arrows in this framework demonstrate how features in one concept reciprocally relate to features in another concept. For example, in this study I hypothesize that hospital nurse staffing (system) is associated with 30 day readmission (outcome). In light of the Hospital Readmission Reduction Program financial penalties, it could also be hypothesized that 30 day readmission rates (outcome) are associated with hospital revenue (system). This example illustrates how two concepts in the Quality Health Outcomes Model can be bi-directionally related.

Although the system concept is related to the intervention concept, the intervention concept was not specifically explored in this study. Generally, the intervention concept encompasses all of the care which nurses deliver to patients, as represented by the bi-directional arrow to the concept of client. The client or patient is conceptually defined as the person receiving the care intervention. The client could also be the family or support system of the person receiving care. While it may be theoretically and empirically important to understand the role of caregivers in preventing readmissions, the data available for this study limited such an analysis. Therefore, for the purposes of this study, I considered the client to be the patient. Client characteristics of interest in this study included age, sex, comorbidities, type of procedure, and number of



procedures during index admission. Other client characteristics were further explored in a sensitivity analysis, described later in this chapter. Although a mediation analysis was not conducted in this study, conceptually, the client is a mediating concept between interventions and outcomes of care. Theoretically, this means that the mediator variable (client) explains the relationship between the predictor variable (intervention) and the outcome variable (outcome).

The outcome of care that was of interest in this study was 30 day unplanned readmission following elective THA/TKA. The arrow from outcomes to the system was not explored, but has important implications for future research related to readmissions. With the implementation of pay-for-performance under the Affordable Care Act, patient outcomes will result in either monetary rewards or penalties to systems based on the desirability of outcomes (PPACA, 2010). The outcomes of care will directly affect the financial inputs to the system.

Lastly, the bi-directional arrow between the system and client demonstrates that the two concepts act as mediators of the effect of the intervention on the clinical outcome. Thus, the system and the client never operate independently of the other. The system explains part of the relationship between the intervention and the outcome.

Background

Overview

The following section provides a background of existing knowledge related to readmissions in a Medicare population undergoing THA/TKA. The review begins with the historical interest in readmissions, and then focuses specifically on readmissions



following THA/TKA. The various definitions of THA/TKA readmissions are described, as well as the reasons for and incidence of readmission following THA/TKA.

The remainder of this chapter includes a review of the various patient and hospital characteristics that are commonly risk adjusted for in the THA/TKA readmission literature. Hospital nursing (staffing and practice environment) has, to my knowledge, not been studied for its effect on readmissions in patients following THA/TKA. The effect of these features will be discussed in the context of other studies of readmission in different patient populations. As mentioned in the preceding chapter, nurse education has been associated with many patient outcomes, yet evidence of an association between education and readmissions is less clear. A discussion about why nurse education was not included as a primary predictor of interest in this study is elaborated.

Readmission

The study of readmissions initially became popular in the 1950s. During this time, the psychiatric population – specifically, patients with schizophrenia – were the focus of readmission research. In the mid to late 20th century, more than half of discharged patients with schizophrenia experienced rehospitalizations (Achté & Apo, 1967; Jenkins, Bermiss, & Lorr, 1953; Strauss et al, 1974; Wing, Denham, & Munro, 1959).

After the introduction of CMS in 1965 and the resulting rise in healthcare spending, concerns about frequent readmissions in the older adult population surfaced. In a seminal study of hospital readmissions in the Medicare population, it was found that the 60 day readmission rate for all Medicare admissions was 22% (Anderson & Steinberg, 1984). Moreover, Medicare was estimated to be spending approximately \$8 billion in 1984 dollars on readmissions (Anderson & Steinberg, 1984).



Clinical concerns about avoidable readmissions spurred researchers' interest in reducing readmissions, yet policy lagged until 2010 with the passage of the Affordable Care Act. The Hospital Readmission Reduction Program payment reform, a provision in the Affordable Care Act, allows CMS to withhold a percentage of base CMS reimbursements for hospitals with worse than expected readmission rates for certain medical and surgical conditions. In 2012, the first penalties were targeted at all-cause readmissions following an admission for congestive heart failure, acute myocardial infarction, or pneumonia. In 2015, THA/TKA and chronic obstructive pulmonary disease joined the list.

Total hip and total knee replacements are a relatively new orthopedic surgery, with the first procedures performed in 1960 and 1968, respectively (AAOS, 2014). Today, more than 285,000 THA and 600,000 TKA are performed annually in the United States (AAOS, 2014) – of which, Medicare pays for approximately two-thirds (Ong et al, 2006). The popularity of these procedures and their expense to Medicare have made THA/TKA the target of CMS reimbursement penalties for worse than expected readmission rates, beginning in 2015 (CMS, 2014). In response to these pay-forperformance policy initiatives, hospital administrators are increasingly interested in understanding how to reduce readmissions following these procedures. The following describes the current knowledge on readmission following THA/TKA, as well as the predictors of readmission following hospital administrators in medical and surgical populations.



Definition

Despite a significant and growing body of research on hospital readmissions, the literature lacks a consistent definition of what temporally distinguishes a hospitalization as a readmission. The majority of studies that examine readmissions in a THA/TKA population define a readmission event as a hospitalization occurring with 30 or 90 days from discharge; however, the timeframe in the literature ranges from 28 days (Cullen, Johnson, & Cook, 2006; Khan et al, 2012) to 180 days after discharge (Riggs et al, 2010).

The time from discharge to rehospitalization is important for defining a readmission event. CMS has determined a readmission within 30 days of discharge to be the timeframe for measuring hospital care quality (Suter et al, 2014). This decision appears arbitrary and lacks strong empirical evidence to suggest a reasonable timeframe for measuring hospital care quality. Although researchers and hospital administrators have argued that a 30 day timeframe extends beyond a hospital's capacity to control patient outcomes (Vaduganathan, Bonow, & Gheorghiade, 2013), CMS continues to use this measure as the basis for withholding reimbursements under the Hospital Readmission Reduction Program.

In addition to defining a timeframe that constitutes a readmission, CMS also distinguishes between planned and unplanned readmissions in the THA/TKA population. The first study to attempt to differentiate planned and unplanned readmissions did so using clinical reasoning (Jencks, Williams, & Coleman, 2009). An unplanned readmission may or may not be avoidable, but is theoretically indicative of poor care quality. A planned readmission is theoretically indicative of good care quality. For example, it is clinically reasonable to believe that a readmission following THA/TKA for



inpatient rehabilitation is planned because rehabilitation is typically warranted following an intensive orthopedic surgery, and can be scheduled in advance. On the other hand, a readmission for shortness of breath due to pulmonary emboli is likely unplanned. Pulmonary embolism is an acute and potentially fatal event for which there are prophylactic measures. Another study defined unplanned readmissions as "any subsequent admissions through the emergency department within 180 days of the index admission" (Riggs et al, 2010), thereby assuming that emergency department visits are by definition, unplanned events.

The THA/TKA Readmission Technical Report, which is prepared for CMS by a study team at the Yale New Haven Health Services Corporation/Center for Outcomes Research and Evaluation (YNHHSC/CORE) and validated by the National Quality Forum, outlines an algorithm (Figure 3.1) to identify a planned versus unplanned readmission in the calculation of a hospital's THA/TKA readmission rate (Suter et al, 2014). A planned readmission is believed to be an indicator of good care quality – such as a planned follow-up appointment for postoperative care. An unplanned readmission is one that could have potentially been avoided and is therefore indicative of poor quality care, such as the development of a postoperative infection (Suter et al, 2014). To my knowledge, there have been no studies in which researchers distinguish between a planned or unplanned readmission using the algorithm set forth in the THA/TKA Readmission Technical Report (Figure 3.1). This study used the readmission algorithm to distinguish between planned and unplanned readmissions. Unplanned readmission was the outcome of interest in this study.



Reasons

Identifying the reasons for readmission is important for understanding how to reduce the incidence of readmission in the future. In studies of patients undergoing THA, the main reasons for readmission were found to be more often related to medical issues than surgical issues. Among the most common medical reasons for readmission after THA are pneumonia, dehydration and renal dysfunction, deteriorating mobility, congestive heart failure, cardiac dysrhythmias, osteoarthritis, acute myocardial infarction, and diabetes (Khan et al, 2012; Vorhies et al, 2012). The literature lacks consensus about whether the most common medical causes are more likely to be cardiac or pulmonary in nature (Schairer et al, 2014; Vorhies et al, 2011). Common surgical reasons for readmission in THA include dislocation of the prosthesis, surgical site infection, wound disruption, and postoperative hematoma (Cullen, Johnson, & Cook, 2006; Pugely et al 2013; Saucedo, 2014; Schairer et al, 2014). Reasons for readmission following TKA are more likely related to surgical issues, such as surgical site infection, cellulitis, and arthrofibrosis (Schairer, Vail, & Bozic, 2014). A recent study found surgical site infection to be the most common reason for unplanned readmission following THA/TKA (Merkow et al, 2015). Overall, there appears to be no definitive consensus across the literature regarding the principal reasons why patients are readmitted following THA/TKA. This study will describe the top ten most common reasons for readmission for both THA and TKA.



www.manaraa.com
Incidence

THA/TKA are among the most common inpatient procedures in the Medicare population (CDC, 2014; Kocher et al, 2013). Yet readmissions following THA/TKA are relatively uncommon, as a percentage of procedures, when compared with readmission rates for medical conditions, such as congestive heart failure (1 in 4), acute myocardial infarction (1 in 5) and pneumonia (1 in 6) (Krumholz et al, 2009; McHugh & Ma, 2013). Estimates from the literature find that 30 day readmission rates for THA and TKA range from 4-12% and 4-6%, respectively (Cullen, Johnson, & Cook, 2006; Khan et al, 2012; Mahomed et al, 2003; Merkow et al, 2015; Schairer, Vail, & Bozic, 2014; Zmistowski et al, 2013). This wide variation in reported readmission rates may be explained by trends over time. THA readmission rates steadily declined from 1991 to 2006, followed by a sudden increase in 2007 and 2008, which has been attributed to increasing patient complexity and reductions in length of stay (Cram et al, 2011). A similar study of the change in TKA readmissions over time also contributed reductions in length of stay to increases in readmission over time (Cram et al, 2012).

The estimates across the literature are generally consistent with CMS estimates of a combined THA/TKA readmission rate of 5.4% (Suter et al, 2014). With approximately 885,000 THA/TKA procedures performed each year, nearly 50,000 patients are unnecessarily readmitted annually (AAOS, 2014).

Risk adjustment

Risk adjustment is important for allowing meaningful comparisons to be made between different groups (Iezzoni, 2013). In a perfectly randomized study, risk adjustment would not be necessary because the groups for comparison would be alike on



all observable and unobservable attributes. However, in a cross-sectional study of existing data, it is necessary to account for the observable attributes that differ among the comparison groups, using risk adjustment (Iezzoni, 2013).

In this study, a number of patient and hospital characteristics are controlled for through risk adjustment. The following section discusses which patient and hospital characteristics are commonly adjusted for in the literature. While the risk adjustment for the analysis of the specific aim will be guided by the THA/TKA Readmission Technical Report, a sensitivity analysis of additional variables will be informed by the literature. The THA/TKA Readmission Technical Report will serve as the guide for sample selection and risk adjustment for regression models to ensure the findings of this study are relevant to hospitals at risk for CMS reimbursement penalties.

Patient characteristics

Age

The current literature on THA/TKA readmissions as well as the THA/TKA Readmission Technical Report consistently adjusts for patient age, finding that older age is associated with an increase in a patient's odds of readmission (Hu, Gonsahn, & Nerenz, 2014; Pugely et al, 2013; Tsai et al, 2013; Whittle et al, 1993; Zmistowski et al, 2013). This relationship makes intuitive sense because with age comes increased frailty and comorbidity. However, another study found no significant effect of age on readmission (Schairer, Vail, & Bozic, 2014). Patient age was controlled for in this study.



The current literature on THA/TKA readmissions as well as the THA/TKA Readmission Technical Report consistently adjusts for patient sex. The effect of sex on readmission varies across studies, with some studies finding no association between readmission and sex (Schairer, Vail, & Bozic, 2014). However, the majority of study findings agree that being male is a stronger predictor of readmission, than being female (Hu, Gonsahn, & Nerenz, 2014; Pugely et al, 2013; Singh et al, 2013; Tsai et al, 2013; Zmistowski et al, 2013). It is unclear why male sex is a greater predictor of readmission; however, one theory suggests that males tend to hold off seeking necessary medical care such that once males present for treatment, they are sicker than their female counterparts. Males may also engage is greater risk-taking behavior which could put them at risk for falls or other adverse events related to self-care management. Patient sex was controlled for in this study.

Comorbidities

Sex

Adjusting for patient acuity or comorbidities allows for valid comparisons of patients across outcomes. Although studies have used a range of risk adjustment techniques to account for comorbidities, such as a Charlson score, Elixhauser risk adjustment, or selecting relevant or common comorbid conditions, studies consistently find that patients with more comorbidities are at higher risk for readmission (Khan et al, 2012; Mahomed et al 2003; Riggs et al, 2010; Saucedo et al, 2014; Schairer et al, 2014; Schairer, Vail, & Bozic, 2014). Another method of adjusting for patient acuity is the American Society of Anesthesiologist score which was also shown to be positively



associated with odds of readmission for THA/TKA patients (Bini et al, 2009; Pugely et al, 2013). Patient comorbidities, as detailed in the THA/TKA Readmission Technical Report, were controlled for in this study.

Type and number of procedures

In a review of the literature, no prior studies were found to adjust for the type (THA vs. TKA) or number of procedures. Although no research strongly suggests that THA patients compared to TKA patients are more or less likely to be readmitted, the THA/TKA Readmission Technical Report adjusts for procedure type. The number of procedures may be a meaningful predictor of readmission, based on the theory that an individual recovering from multiple surgeries on the same admission may be more likely to experience a complication warranting readmission. The THA/TKA Readmission Technical Report of THA/TKA procedures on the index admission. Therefore, both the type and number of procedures were controlled for in this study.

Discharge destination

Where patients convalesce following a hospital discharge is associated with their odds of readmission following THA/TKA. Studies find that patients discharged to skilled nursing facilities have higher odds of readmission, even after adjusting for patient acuity and frailty (Bini et al, 2009; Schairer et al, 2014). This is a well-described finding across patient populations (Tsai et al, 2013). Although, to my knowledge, no reasons for this finding have been hypothesized and tested in the literature, patients admitted to a skilled



nursing facility may be at higher risk of readmission because skilled nursing facilities have no disincentive to readmit a patient at the earliest warning signs of a potential issue (Mor et al, 2010).

Zmistowski and colleagues (2013) find that patients who are discharged to inpatient rehabilitation have higher odds of readmission, while another study reports opposing findings (Riggs et al, 2010). These findings may be the result of patients in inpatient rehabilitation being readmitted prematurely at the earliest warning sign of a potential issue, or as a result of the positive impacts rehabilitation can have on postoperative THA/TKA patients, such as mobility and strength training. It is also unclear to what extent discharge destination is a proxy measure for patient acuity, the complexity of the procedure, or the quality of care during the hospitalization.

Discharge destination is not included as a variable for risk adjustment in the THA/TKA Readmission Technical Report because discharge destination is a factor of the structure of the healthcare system and patients' comorbidities (Suter et al, 2014). Geographic variation in the availability of providers and practice patterns make discharge destination an unreliable factor on which to risk adjust (Suter et al, 2014). However, in keeping with prior literature, discharge destination was controlled for in a sensitivity analysis.

Length of stay

Much of the evidence suggests that longer lengths of stay in the hospital for patients following THA/TKA are associated with higher odds of readmission (Jencks, Williams, & Coleman, 2009; Riggs et al, 2010; Saucedo et al, 2014; Schairer et al, 2014;



Schairer, Vail, & Bozic, 2014; Zmistowski et al, 2013). Yet some studies find no association between length of stay and readmission (Mnatzaganian et al, 2012), while others find that shorter lengths of stay increase the risk for readmission (Heggestad, 2002). In sum, the association between length of stay and readmission remains unclear.

Length of stay is a complicated measure to risk adjust for when assessing hospital care quality. A patient's length of stay is a factor of (1) the patient's severity of illness and (2) the amount of care received. While it is appropriate to risk adjust for patient severity of illness, to allow for a more fair comparison of hospital quality, adjusting for the amount of care delivered may be correlated with hospital quality and performance, which is likely to be correlated with patient health outcomes. It is often unclear whether the length of stay is appropriate or inappropriate for any individual patient; therefore, it can be difficult to determine whether or not the hospital is treating patients efficiently. Because it is unclear if it is appropriate to adjust for length of stay, and because it is not risk adjusted for in the THA/TKA Readmission Technical Report, length of stay was controlled for in a sensitivity analysis.

Race/Ethnicity

Race/ethnicity is commonly adjusted for in the literature on THA/TKA readmissions; however, studies lack agreement about how race affects a patient's likelihood of experiencing a readmission. Mahomed and colleagues (2003) find that whites have higher odds of readmission as compared with blacks. Other studies find that blacks have higher odds of readmission as compared with whites (Hu, Gonsahn, & Nerenz, 2014; Jencks, Williams, & Coleman, 2009; Zmistowski et al, 2013).



Race/ethnicity is not risk adjusted for in the THA/TKA Readmission Technical Report, and was therefore not be adjusted for in the analysis of the specific aim. Rather, race/ethnicity was controlled for in a sensitivity analysis.

Socioeconomic status

Socioeconomic status, a measure of an individual's income, education, and employment status (Green, 1970), is a predictor of health outcomes, including a patient's risk of readmission and surgical mortality (Birkmeyer et al, 2008; Weissman, Stern, & Epstein, 1994). However, it is less clear whether, in the context of risk for readmission, socioeconomic status acts as a proxy for other factors that may influence readmissions, such as social support systems. In an effort to measure social support, Hu and colleagues (2014) accounted for the effect of marital status on likelihood of readmission, and found that patients who are unmarried have higher odds of readmission compared with patients who are married. This analysis assumes that marriage is correlated with increased social support. In another study, no significant relationship was found between readmission and measures of social support including marital status, living situation, and availability of help at home (Weissman, Stern, & Epstein, 1994).

Income alone has been found to be associated with readmissions, such that people with a lower income have higher odds of readmission following THA (Mahomed et al, 2003; Weissman, Stern, & Epstein, 1994). Income data is not typically reported in medical records, making this information difficult to obtain. One study used the receipt of supplemental security income as a proxy for low-income status (Jencks, Williams, & Coleman, 2009). Area-based measures of income created through geo-coding can be used



to link a patient's address with Census data (Brooks Carthon, 2012; Diez Roux et al, 2001); however, this approach is limited in areas with greater socioeconomic heterogeneity (NQF, 2014). Another proxy for measuring individual income status that is more often reported in patient health data is Medicaid eligibility. Medicaid eligibility or dual eligible status for the Medicare population has been verified as a valid indicator of low-income status (NQF, 2014). However, this measure is not without disadvantages, such as the variation in income among the Medicaid and dual eligible population (NQF, 2014). Uninsured patients are also less likely to be readmitted; however, this could be related more to the desire to avoid the financial cost of accessing healthcare, rather than the health status and needs of the individual (Weissman, Stern, & Epstein, 1994).

Education – a component of the socioeconomic status measure – is not well documented in patient health data (NQF, 2014). In theory, patients with higher educational attainment are less likely to be readmitted. Health literacy and the ability to navigate the healthcare system can positively impact a patient's ability to provide selfcare as they transition from the hospital.

Employment status/occupation is the third component of the socioeconomic status measure. However, like education, this information is difficult to obtain from patient health data (NQF, 2014). Weissman and colleagues (1994) found that patients working in unskilled occupations were more likely to be readmitted. Again, it is unclear whether employment status and occupation are proxy measures for individual characteristics that may influence an individual's health outcomes. Employment status is positively correlated with income, and occupation may be related to literacy.



Patient socioeconomic status is not adjusted for in the THA/TKA Readmission Technical Report and was therefore not adjusted for in this study. Rather, a patient level measure of socioeconomic status was controlled for in a sensitivity analysis.

Hospital characteristics

Bed size

Studies comparing readmission rates across hospitals often control for a number of hospital characteristics, including bed size. In theory, larger hospitals are more likely to have a greater volume of patients and therefore perform surgical procedures more frequently. The more experienced providers are with a particular procedure, the more likely patients are to have positive health outcomes. On the other hand, larger hospitals are more likely to attract sicker patients, which may negatively impact hospital performance on outcome measures. As described later in this section, the findings pertaining to the relationship between patient volume and outcomes in surgical patients are mixed. Specifically, one study found the larger hospitals were more likely to be penalized under the Hospital Readmission Reduction Program penalties (Joynt & Jha, 2013); while an earlier study by the same researchers, found that smaller hospitals had higher readmission rates (Joynt & Jha, 2011). In sum, the relationship between hospital bed size and readmission, is not well defined. Hospital bed size was controlled for in this study.

Teaching status

Teaching hospitals are defined by the presence or absence of medical fellows and/or residents providing medical care within a hospital. Little research has looked at the



relationship between a hospital's teaching status and patient readmission rates; however, some evidence suggests that teaching hospitals are associated with higher readmission rates and are more likely to be penalized under the Hospital Readmission Reduction program (Joynt & Jha, 2013; Press et al, 2013). Teaching hospitals tend to be located in urban cities, close to medical universities. They tend to be large in size, which provides medical trainees with an opportunity to get experience in a range of clinical fields. The association between teaching hospitals and greater readmissions may be an effect of the acuity of patients who are attracted to large, urban hospitals. Hospital teaching status was controlled for in this study.

Technology status

Hospital technology status is defined by a hospital's capacity to perform openheart surgery and/or major organ transplantation. High technology hospitals also tend to be large, urban, teaching hospitals, which attract a case mix of patients that are on average, sicker and more clinically complex. Little research has directly examined the association of hospital technology status on readmissions; however, theory suggests that high technology hospitals would be associated with higher readmissions. Hospital technology status was controlled for in this study.

Geographic location

The geographic location of hospitals may be associated with a patient's risk for readmission; however, little research examines this direct association. As previously mentioned, urban hospitals are more likely to be larger, teaching hospitals with high



technology status – and therefore, more likely to attract sicker, more clinically complex patients. Rural hospitals, on the other hand, are more commonly smaller community hospitals. Hospital geographic location was controlled for in a sensitivity analysis.

Caseload volume

The volume-outcome relationship has been well researched; yet studies exploring the effect of surgeon and hospital volume on readmission reveal mixed findings. No relationship between volumes and readmission outcomes was found in a study of TKA patients (Judge et al, 2006). However, the same study did find that high-volume trusts in England's National Health Service were associated with lower readmission rates for THA (Judge et al, 2006). To the contrary, another study in Finnish hospitals reported conflicting findings with evidence that lower volume trusts had lower readmission rates (Mäkelä et al, 2011). Another study found no association between hospital volume and readmissions following surgery, but a positive association between high-volume and mortality (Goodney et al, 2003). Interestingly, a study of mortality following abdominal aortic aneurysm repair found the relationship between lower patient mortality in highvolume hospitals was contingent upon better nurse staffing in those hospitals (Wiltse Nicely, Sloane, & Aiken, 2012). This suggests that if a volume-outcome relationship exists in this postsurgical population, it may be correlated with hospital nurse work conditions. In sum, there is no consensus across the literature on the effect of hospital caseload volume on readmissions. The effect of hospital nursing on readmission was tested at various hospital caseload volumes in a sensitivity analysis.



Ownership type

There is little evidence to suggest an association between a hospital's ownership status and patient readmission outcomes. While one study found that not-for-profit performed better on readmission rates as compared with government-owned hospitals (Joynt & Jha, 2011), there is little theory or a robust empirical evidence-base to support any strong associations. Hospital ownership type was controlled for in a sensitivity analysis.

Socioeconomic status profile

In addition to patient socioeconomic status, a hospital's socioeconomic status profile, may explain some of the relationship between nursing and postsurgical readmission. Some hospitals have been described as disproportionate share hospitals (DSH), a definition used by the CMS to provide additional payments to hospitals serving a significant percentage of individuals insured through Medicaid. A study by Blegen and colleagues (2011) used DSH status to identify safety-net hospitals. Although the average staffing ratios were not significantly different between safety-net and non-safety-net hospitals, patients in safety-net hospitals were more likely to have poorer outcomes, including higher rates of mortality (Blegen et al, 2011). This study suggests, that despite adjusting for nurse staffing, hospital and patient characteristics, patients who receive care in a hospital with higher proportions of low socioeconomic status might be at greater risk for adverse outcomes.

However, studies vary in how they define safety-net hospitals. Safety-net hospitals have been identified based on the amount of uncompensated care provided, the



caseload of Medicaid patients, or other hospital characteristics such as being a public or teaching hospital (McHugh, Kang, & Hasnain-Wynia, 2009). Although no single best approach to defining a safety-net hospital has been identified, the method for identifying a safety-net hospital varies across the literature, leading to mixed findings about the impact of safety-net status on health outcomes (McHugh, Kang, & Hasnain-Wynia, 2009). For this study, a hospital's socioeconomic status profile was calculated based on the proportion of study patients within the hospital with a low socioeconomic status. Hospital socioeconomic status profile was controlled for in a sensitivity analysis.

Magnet® designation

In response to a nursing shortage in the 1980s, a study was conducted by the American Academy of Nursing to attempt to entice people to pursue a career in nursing (McClure et al, 1983). This study identified organizational features of certain hospitals that were best able to recruit and retain nurses during the nursing shortage. These hospitals were described to have "magnet-like" properties as a result of their low nurse turnover rates and high nurse satisfaction (McClure et al, 1983). Beginning in 1994, the American Nurses Credentialing Center (ANCC) began awarding hospitals that met specified criteria demonstrating nursing excellence as Magnet® designated hospitals (ANCC, 2014). Magnet® designation has been considered a proxy measure for hospitals with exceptional nurse practice environments.

A large and growing body of research suggests that better nurse work environments are associated with positive patient outcomes and reductions in mortality and readmission (Aiken, Smith, & Lake, 1994; Aiken et al, 2008; Aiken et al, 2011; Ma,



McHugh, & Aiken, 2015; McHugh & Ma, 2013). Magnet® designation is often used by researchers who do not have a direct measure of the nurse practice environments. In other words, Magnet® designation is a proxy for a direct measure of the practice environment. Although this study uses a direct measure of the practice environment, as measured by the PES-NWI, adjusting for each hospital's Magnet® designation status in a sensitivity analysis ensures that the direct measure of the practice environment is appropriately and fully accounting for hospital working conditions.

Surgical care improvement project process measures

Surgical care improvement project process measures are hospital quality performance measures included as one of the value-based purchasing programs among the pay-for-performance initiatives in the Affordable Care Act. Hospital postsurgical readmission rates have been found to be only weakly correlated with performance on surgical care improvement project measures (Tsai et al, 2013). The authors hypothesized that this finding may be due to the modest variation in hospital surgical care improvement project measure performance. A prior study of surgical care improvement project process measure performance revealed no significant association with surgical site infection (Garcia et al, 2012). Given this evidence, these measures may not be a highly valid quality measure for predicting patient health outcomes. Nevertheless, the measures do hold valid theoretical weight in the relationship of hospital care and postsurgical readmission. Therefore, as part of a sensitivity analysis, the hospitals' performance on surgical care improvement project measures were controlled for in a sensitivity analysis.



Nursing characteristics

The following section describes hospital nursing characteristics – staffing and practice environment – which were the primary predictors in this study. These features of hospital nursing have been consistently shown to be associated with various patient health outcomes, including readmission (Kane et al, 2007; Kazanjian et al, 2005; Shekelle, 2013; Ma, McHugh, & Aiken, 2015; McHugh & Ma, 2013; Weiss, Yakusheva, & Bobay, 2011).

Staffing

Registered nurses in the hospital are responsible for the ongoing surveillance of changes in patient status in response to medications, treatments, and disease progression. In a sense, hospital nurses are an around-the-clock surveillance system that is essential for detecting early warning signs of patient decompensation and intervening in order to prevent or ameliorate an adverse event. Patients undergoing elective orthopedic surgeries are generally healthy since they are deemed medically stable to undergo a major surgery. However, even the healthiest patients are at risk for complications in the postoperative period. Complications can be quick in onset and fatal, requiring rapid and intelligent response to prevent worsening. Some potentially fatal complications progress slowly overtime, but have subtle early warning signs, such as infection. Other complications, such as physical deconditioning, occur over time when patients are not frequently assisted and encouraged to ambulate. Whether postoperative complications progress quickly or slowly, the nurse is often the first healthcare provider to identify a change in



the patient condition and initiate a response. For many complications, a response, or lack thereof, can mean the difference between life and death for a postoperative patient.

The more patients a nurse cares for, the less time a nurse is able to spend with each individual patient which inhibits the nurse's capacity to recognize the subtle early warning signs of patient decompensation. Heavier nurse workloads can also influence the nurse's capacity to perform other essential nursing care tasks, such as the timely administration of medications, education related to self-care and disease management, and ambulation. Failure to perform these essential nursing care tasks can compromise the safety of patients, both within the hospital and beyond discharge.

Patients undergoing THA/TKA require ongoing mobility assistance, pain management, wound management, and education about self-care following a major joint replacement. The more patients a nurse cares for, the less time a nurse is able to spend providing thorough and in-depth care to each individual patient. Specifically, for patients following THA/TKA, nurses must tailor education and care instructions to each patient individually. Some patients may have support systems to help them perform activities of daily living when they return home, while other patients may require additional assistance from home health aides, registered nurses, and/or physical therapists. It is essential for hospital nurses to learn about each patient's individual needs in order to ensure that the patient can safety transition from the hospital to home.

With over two decades of research on nurse staffing, researchers consistently find that more favorable nurse staffing ratios positively impact patient outcomes (Kane et al, 2007; Kazanjian et al, 2005; Shekelle, 2013). Across the literature and in various patient populations, more favorable nurse staffing has been found to be associated with



reductions in patient mortality and a number of adverse patient events, including readmission (Kane et al, 2007). In a study of adult patients admitted for general, orthopedic, and vascular surgeries in 168 hospitals, each additional patient per nurse was associated with 7% higher odds of mortality (Aiken, Clarke, Sloane, Sochalski, & Silber, 2002). Using unit level analysis, Needleman and colleagues (2011) examined patient exposure to nurse staffing when the unit was staffed below and at the staffing target. Staffing below the unit target was found to be associated with increased mortality (Needleman et al, 2011). The effect of more favorable staffing levels has also been shown to attenuate the odds of unplanned readmission and in-hospital mortality in a cardiac surgical population (Diya et al, 2011). A study by Ma and colleagues found every additional surgical patient per nurse increased the odds of readmission by 3% (Ma, McHugh, & Aiken, 2015).

Other studies have examined the association in medical patients. An analysis of national claims data for patients in all United States hospitals categorized nursing staffing into quartiles and found that patients discharge from hospitals in the lowest staffing quartile experienced higher readmission rates compared with patients discharged from hospitals in the highest staffing quartile (Joynt & Jha, 2011). These findings were further supported in a study of 375,681 patients with a primary diagnosis of congestive heart failure, acute myocardial infarction, or pneumonia, in 412 hospitals. Each additional patient in the nurse's average workload was associated with 7%, 6%, and 9% higher odds of readmission for congestive heart failure, pneumonia, and acute myocardial infarction patients, respectively (McHugh & Ma, 2013). Another study found that hospitals with more favorable staffing ratios were less likely to face CMS readmission penalties as



compared with hospitals with less favorable staffing ratios (McHugh, Berez, & Small, 2014).

Practice environment

In addition to having the time to perform essential nursing care tasks, the social context in which nurses work can impact the amount and quality of care nurses are able to deliver. The nurse practice environment describes this social context as the perceived role nurses have within their institution (Lake, 2002). It includes the relationships nurses have with other healthcare providers, including physicians and their direct supervisors (Lake, 2002). A positive social context in which nurses are able to effectively communicate with other providers can influence the care process and patient outcomes. A growing body of research finds that hospitals with better work environments are associated with fewer adverse patient outcomes.

When the clinical resources needed to perform nursing tasks are readily available, positive collegial relationships among physicians and nurses exist, and nurses experience clinical and political autonomy in hospital affairs, then nurses are better positioned to provide high quality care. Formally referred to as the practice environment, the organizational culture and resources available to nurses have been found to have predictive validity for patient outcomes, including readmission (Ma, McHugh, & Aiken, 2015; McHugh & Ma, 2013).

An early study of the practice environment matched 39 "magnet" hospitals known for good nursing care with 195 control hospitals and found that, controlling for hospital characteristics, patients in the "magnet" hospitals experienced a 7.7% lower mortality



rate than the control hospitals. After adjusting for patient characteristics in the "magnet" and control hospitals, the results remained significant, with the "magnet" hospitals, known for better nursing care, experiencing a 4.6% lower mortality rate (Aiken, Smith, & Lake, 1994). These findings were replicated in a study of surgical patients in 168 Pennsylvania hospitals using the PES-NWI to measure the nurse practice environment. In conjunction with more favorable nurse staffing and higher proportions of baccalaureate educated nurses, better nurse practice environments were associated with lower 30 day mortality rates (Aiken et al, 2008). A later study demonstrated the moderating relationship of the practice environment on the nurse staffing effect on 30 day mortality such that the effect of decreasing nurse workloads was greater in hospitals with better practice environments (Aiken et al, 2011).

The practice environment's effect on readmission outcomes is less well understood. One study of medical patients found that patients cared for in the best practice environments compared with the poorest, experienced 7%, 6%, and 10% lower odds of 30 day readmission for congestive heart failure, acute myocardial infarction, and pneumonia, respectively (McHugh & Ma, 2013). A study of general, vascular, and orthopedic surgical patients, found that patients cared for by nurses working in better work environments had 3% lower odds of readmission (Ma, McHugh, & Aiken, 2015). A systematic review of ten studies revealed that a lack of standardized measures and methodological rigor prevents a clear understanding of the influence of the practice environment on readmissions (Ma, Shang, & Stone, 2014). This study attempts to build on previous findings of an association between readmission and the practice environment to bolster the evidence.



Education

Nurse education, or the hospital proportion of nurses holding a bachelors degree, is associated with various health outcomes. A rigorous body of empirical evidence suggests that hospitals with lower proportions of bachelors degree nurses are linked with poorer patient outcomes (Aiken et al, 2012; Aiken, Clarke, & Sloane, 2002; Blegen et al, 2013; Estabrooks et al, 2005; Kendall-Gallagher, et al 2011; Kutney-Lee, Sloane, & Aiken, 2013; Needleman et al, 2006; Tourangeau, Cranley, & Jeffs, 2006; Yakusheva, Lindrooth, & Weiss, 2014a).

The association of nurse education with readmission is less well understood. McHugh and Ma (2013) explored the relationship of nurse education and readmission in Medicare patients with congestive heart failure, acute myocardial infarction, and pneumonia but did not find a statistically significant relationship for patients with congestive heart failure or acute myocardial infarction. For patients with pneumonia, patients had 3% lower odds of readmission for each 10% increase in the hospital proportion of nurses with a bachelors degree (McHugh & Ma, 2013). Another study of readmissions among surgical patients found no effect of nurse education (Ma, McHugh, & Aiken, 2015). Because prior evidence does not suggest a robust effect of nurse education on readmissions, nurse education was not examined as a primary predictor of interest in this study.

Summary

While the effects of nurse staffing and practice environment on postsurgical outcomes including mortality and failure-to-rescue are well understood, much less is known about their effects on postsurgical readmissions. To my knowledge, this study was



the first to examine how hospital nursing is associated with readmissions in Medicare patients following elective THA/TKA.

Significance

Social

Hospitalization is a significant event for patients and their families. In an attempt to understand the patient experience of hospital readmission, the Robert Wood Johnson Foundation published a report, *Hospital Readmissions from the Inside Out: Stories from Patients and Health Care Providers*, which tells the stories of 16 readmitted patients (RWJF, 2013). Although patients believed their readmission event to be a unique experience, common themes emerged from the patient stories. Many patients felt they were discharged from the hospital prematurely and many did not understand their discharge instructions or felt their care instructions were too vague (RWJF, 2013). Many patients were concerned about being home alone upon discharge and reported they did not have a support system when they returned home (RWJF, 2013). One patient recounted his emotions about being home alone on the first night after being discharged, "I was real nervous; I didn't know if I would make it. I thought this might be it" (RWJF, 2013). These vignettes reveal the difficulty of transitions for patients following hospitalization.

The hospitalization itself, as well as the transition from the hospital are emotionally, mentally, and physically taxing for many older patients. Insufficient education about self-care management, care coordination, and post-discharge support exacerbate the difficulties patients face upon hospital discharge (RWJF, 2013). Older adult patients can be especially frail and deconditioned when leaving the hospital, making



activities of daily living more difficult to accomplish (Kortebein, 2009). An inability to perform activities of daily living, such as preparing meals and basic self-care like taking medications, put these patients at risk for readmission.

Financial

The CMS quality initiative to reduce costly and avoidable readmissions has fundamentally complicated hospital administrators' financial incentives. Hospital administrators are now forced to grapple with a perverse financial incentive to earn additional money for the second hospital stay of a readmitted patient, or prevent readmissions and avoid CMS withholding a small fraction of reimbursements for all Medicare patient stays (Burton, 2012; Joynt et al, 2014). As the readmission penalties increase annually, the ambiguity of this decision for administrators may evaporate.

In the second year of implementation, 2,217 hospitals across the country experienced a readmission penalty (Burton, 2012). In total, these hospitals forfeited more than \$280 million in Medicare reimbursements (Rau, 2013). While this amount is only 0.3% of total Medicare reimbursements made to hospitals annually, the penalties are expected to become more severe over time, placing more hospital revenue at stake (Rau, 2013). After the third year of penalties, 2,610 hospitals were penalized, totaling approximately \$428 million in fines (Rau, 2014). Although data suggest national readmission rates are dropping on average, the penalties increased to 3% of base CMS reimbursements during fiscal year 2015 (Rau, 2014).

To date, it remains unclear whether the financial incentives are encouraging hospitals to provide higher quality care. Between the first and second years of



implementation, the average national penalty decreased from 0.42% of base CMS reimbursements to 0.38% (Rau, 2013). The average penalty in the third year of implementation was 0.68%, slightly less than double the penalties from the prior year (Rau, 2014). It is too early to know if changes in the average penalty are the result of actual care improvements, or modifications in the readmission penalty criteria made by CMS, such as the inclusion of more clinical conditions in the risk adjustment and more refined criteria to differentiate planned and unplanned readmissions (Rau, 2013). Only 129 hospitals which were penalized in the second year were able to avoid a penalty in the third year (Rau, 2014).

Hospitals are unnecessarily forfeiting a portion of reimbursement. Understanding how modifiable characteristics of hospital nursing affect readmission rates is fundamental to understanding the economic interplay of pay-for-performance incentives and investments in hospital nursing.

Political

The enactment of the Affordable Care Act in March 2010 has brought heightened attention to the link between avoidable hospital readmissions and healthcare costs as a signal for poor quality care. Historically, this health reform legislation and the pay-forperformance initiatives, mark the first time that payers are legally able to reimburse providers based on the quality of care, rather than the volume of care delivered.

The first year of implementing the Hospital Readmission Reduction Program began in October 2012. In the initial year of implementation, CMS withheld up to 1% of base CMS reimbursements for those hospitals that had worse than expected readmissions



rates. Expected readmission rates are risk standardized based on the hospital's case mix index as measured by patient demographics (age and sex) and comorbidities (CMS, 2014). In a report to Congress in 2011, CMS stated their aim to reduce readmissions by 20% by the end of 2013, which would result in the prevention of 1.6 million avoidable rehospitalizations and \$15 billion in savings to CMS (Kocher & Adashi, 2011).

Beginning in October 2012, the readmission penalty applied to all cause 30 day readmissions for Medicare beneficiaries with a principal index admission diagnosis of congestive heart failure, acute myocardial infarction, or pneumonia. In the first year of the program, readmission penalties were set for as much as 1% of annual base CMS reimbursements, with penalties increasing to 2% and 3% in fiscal year 2014 and 2015, respectively (CMS, 2014). Beginning in fiscal year 2015, penalties for 30 day unplanned readmissions for THA/TKA, as well as chronic obstructive pulmonary disease, took effect (CMS, 2014).

This study directly addresses modifiable nursing predictors of unplanned 30 day readmission for Medicare patients following elective THA/TKA. Understanding the extent to which nursing contributes to reductions in the likelihood of readmission, can guide hospital administrators and policymakers to more appropriately invest in nursing resources.

Outcome

The outcome of interest in this study was 30 day unplanned readmission. The 30 day readmission outcome was pragmatically chosen based on the Hospital Readmission Reduction Program penalty. In a study of older adult surgical patients, readmission was found to be associated with measures of surgical care quality, including mortality (Tsai et



al, 2013). A meta-analysis of inpatient care quality and 30 day readmission concluded that readmissions are associated with substandard inpatient care quality (Ashton et al, 1997). In a sensitivity analysis, 10 day unplanned readmission was also studied given the common criticism that 30 days from discharge is too far to accurately assess hospital quality (Joynt & Jha, 2012; Vaduganathan, Bonow, & Gheorghiade, 2013).

Covariates

The patient level covariates to be examined in this study were chosen a priori based on the risk adjustment measures used in the THA/TKA Readmission Technical Report, which is prepared for CMS by a study team at the Yale New Haven Health Services Corporation/Center for Outcomes Research and Evaluation (YNHHSC/CORE) (Suter et al, 2014).

Patient level covariates included: age, sex, comorbidities, type of procedure, and number of procedures during admission. Hospital level covariates were included to risk adjust for differences across hospital characteristics. Hospital level covariates in this study included: hospital size, teaching status, and technology status. Hospital level primary predictor variables of interest were used to test the effects of nursing on the odds of readmission. These nurse covariates included: staffing and the practice environment. Other covariates adjusted for in a sensitivity analysis included patient level covariates: discharge destination, length of stay, race/ethnicity, socioeconomic status; and hospital level covariates: geographic location, caseload volume, ownership type, socioeconomic profile, Magnet® designation, surgical care improvement project performance measures.



Summary

In Chapter 2, the conceptual framework, the Quality Health Outcomes Model, was introduced and the relationships tested in this study were described. A review of the literature on readmissions and nursing elucidated the gaps that this study addressed. The social, financial, and political significance of this research was argued. Finally, the outcome and the covariates of interest in this study were described. In the chapters to follow, Chapter 3 describes the methods of the study, an explanation of the parent study, the datasets, the sample, and the analytic plan for the specific aim and the sensitivity analysis; Chapter 4 describes the study findings; and Chapter 5 contains a discussion of the main study findings as well as study limitations, implications, and areas for future inquiry.



CHAPTER 3: METHODS

Introduction

This section addresses the methodological aspects of the study, which aimed to understand the effects of nursing on the likelihood of 30 day readmission for older adults undergoing elective THA/TKA. After a general overview of the study methodology, this chapter describes the parent study from which the nurse survey data was collected, the hospital, nurse, and patient datasets, the procedure for identifying the study sample, and the plan for measuring variables of interest. The analytic plan for the specific aim and sensitivity analysis as well as the methodological limitations and assumptions are discussed. Finally, the concern for and attention to issues related to data integrity and human subjects are addressed.

Overview

This study was a cross-sectional analysis of three 2005-2006 secondary data sources including an annual hospital administrative survey, a survey of nurses, and patient data related to hospitalizations. The hospital administrative survey provided data on hospital characteristics including bed size, teaching status, and technology status. The nurse survey provided demographic data about the nurses as well as nurse reports of staffing and the organizational climate in which the nurse worked. The patient data included patient demographic data as well as diagnoses and procedures during each hospitalization. Three additional data sources were used in the sensitivity analysis: United States Census data, Hospital Compare Surgical Care Improvement Project data, and ANCC Magnet® designation data.



Parent study

The nurse survey data was retrieved from the parent study, the Multi-State Nursing Care and Patient Safety Study survey. The parent study was conducted in 2006, led by Principal Investigator, Dr. Linda Aiken, at the Center for Health Outcomes and Policy Research (CHOPR) at the School of Nursing at the University of Pennsylvania. Prior to data collection, the University of Pennsylvania Institutional Review Board approved the parent study. The data from this study are securely maintained at CHOPR.

Random sampling of a percentage of registered nurses in California (40%), Florida (25%), New Jersey (50%), and Pennsylvania (40%) were surveyed using a modified Dillman approach in which nurses received mailed surveys to their home addresses (Dillman, 1978; Dillman, 2000). These states were a convenience sample based on funding, but they represent nearly one-quarter of the national population and are geographically diverse (U.S. Census Bureau, n.d.). All surveys were mailed in 2005-2006, with the exception of surveys mailed to nurses in Florida, which were mailed in 2007-2008. In total, 272,783 surveys were mailed, with 106,532 surveys mailed to California nurses, 49,385 surveys mailed to Florida nurses, 52,545 surveys mailed to New Jersey nurses, and 64,321 surveys mailed to Pennsylvania nurses. Surveys were mailed to the nurses' home address, which was obtained from his/her state board of nursing. Mailing to the nurses' home, rather than their place of employment, helped to reduce hospital selection bias, which was a potential threat to the study's validity (Aiken et al, 2011). Hospitals with poor nurse working conditions or poor patient outcomes may have discouraged nurses from answering the survey, thereby biasing the sample. This



approach prevented hospitals from influencing whether or not nurses at their hospital responded to the survey.

The nurses received a mailed survey that could be completed with a pencil and paper or on a secured website. Each survey was labeled with a unique barcode and number. Returned surveys were then scanned with these unique identification numbers and no other personal information, thus maintaining the confidentiality of the survey respondents. The nurses were asked to identify the name and location of the hospital, homecare agency, or nursing home in which they were employed. This allowed responses from nurses who reported working in the same facility to be aggregated to create facility level measures (Aiken et al, 2011).

The overall response rate was 39% (Aiken et al, 2011). A follow-up survey involving double sampling was conducted to determine if and to what extent response bias was present in the sample of responders (Smith, 2008). Response bias, a term used to describe the cognitive biases that influence whether a survey recipient will participate, can weaken the validity of the findings. To ensure there was no response bias in the original survey, a follow-up study of 1,300 non-responders in California and Pennsylvania from the original survey was conducted. These non-responders were mailed a shortened survey, a modest monetary incentive, and received telephone reminders to complete the survey. This follow-up survey achieved a 91% response rate. Although there were demographic differences between those who responded to the original survey (responders) and those who responded to the follow-up survey (non-responders), such as age, years of experience, and race/ethnicity, there were no differences in the nurse



reported measures of interest to this study, such as nurse reports of staffing (Smith, 2008).

Data sets

Multi-State Nursing Care and Patient Safety Study Survey, 2005-2006

The Multi-State Nursing Care and Patient Safety Study survey conducted in 2005-2006 is comprised of nurse reports of the nurse's demographics as well as information about the environment in which he or she works. Specifically, the nurse survey data contains information about staffing and the practice environments and culture within the hospital. The survey contains the Practice Environment Scale of the Nursing Work Index Revised (PES-NWI) which is composed of five subscales: 1) nurse participation in hospital affairs; 2) nursing foundations for quality of care; 3) nurse manager ability, leadership, and support of nurses; 4) staffing and resource adequacy; and 5) collegial nurse–physician relations (Lake, 2002).

American Hospital Association Annual Survey of Hospitals, 2006

The American Hospital Association (AHA) is a national organization dedicated to representing hospitals nationally and to engaging in health policy development as well as legislative and regulatory matters pertaining to hospital and healthcare networks. The 2006 AHA Annual Survey of Hospitals used primary survey data to review over 6,500 United States hospitals. The survey provides data on hospital organizational structure, facility and service lines, inpatient and outpatient utilization, expenses, physician arrangements, staffing, corporate and purchasing affiliations, as well as geographic indicators (AHA, 2014). Variables of interest for this study that were sourced from this



data set include: bed size, teaching status, technology status, geographic location, and ownership type.

Medicare Provider Analysis and Review (MedPAR) Research Identifiable File, 2006

The Medicare Provider Analysis and Review (MedPAR) Research Identifiable File (RIF) contains information for all Medicare beneficiaries for every event in which beneficiaries accessed hospital inpatient services (RESDAC, 2013). A given Medicare beneficiary can have multiple MedPAR records, as each record is representative of one inpatient hospital stay (RESDAC, 2013). The MedPAR RIF includes information for each Medicare beneficiary inpatient stay including: ICD-9 diagnoses and procedure codes, diagnosis-related groups (DRGs), dates of service, hospital provider numbers, and demographic information (RESDAC, 2013). This file also appends the date of death, from the National Center for Health Statistics Linked Mortality Files, if death occurred within three years of the date of discharge (CMS, 2013).

Sensitivity analysis datasets

Information about patient socioeconomic status was obtained from United States Census data, which is publically available and collected by the United States Census Bureau. The data has previously been used to create a neighborhood socioeconomic summary index score for each patient's ZIP code (Brooks Carthon et al, 2012; Diez Roux et al, 2001). The socioeconomic index score is a summary of six measures related to an individual's wealth/income (log of the median household income, log of the median value of housing units, the percentage of households receiving interest, dividend or net



rental income), education (percentage of adults 25 years or older who completed high school, percentage of adults 25 years or older who completed college), and occupation/employment (percentage of employed persons 16 years or older in executive, managerial, or professional specialty occupations) (Diez Roux et al, 2001). These six measures were identified through a factor analysis and are reported in the United States Census data in 2000 (Diez Roux et al, 2001). The socioeconomic summary index score was derived from the sum of the z-scores for each of the six measures, with higher scores indicating greater socioeconomic status, within the ZIP code level (Diez Roux et al, 2001). This data was used to explore patient socioeconomic status and hospital socioeconomic status profiles in the sensitivity analysis.

Data about hospital surgical care improvement project performance was collected from Hospital Compare – a website created by CMS to provide publically available data to healthcare consumers about the quality of care in hospitals across the country. In 2006, three measures of hospital surgical care were collected: hospital percentage of surgical patients who/whose (1) received preventative antibiotic(s) one hour before incision, (2) received the appropriate preventative antibiotic(s) for their surgery, and (3) preventative antibiotic(s) were stopped within 24 hours after surgery.

Hospital Magnet[®] designation statuses are publically reported by the ANCC, the organization responsible for recognizing hospitals that demonstrate nursing excellence and awarding Magnet[®] designation (ANCC, 2014). The ANCC website provides publically available data indicating the years in which hospitals have been Magnet[®] designated and re-designated. Hospitals holding Magnet[®] designation during 2005, 2006 or 2007, were identified as Magnet[®] hospitals in the analytic sample.



Sample

Hospitals

Eligibility criteria for inclusion of hospitals in the study included 10 or more nurse respondents and 10 or more patients discharged alive following elective THA/TKA. Previous empirical work suggests that 10 or more nurse survey respondents is sufficient for providing reliable estimates of the hospital's organizational features (Aiken et al, 2003). In the sensitivity analysis, the effects of nursing on readmission were assessed at different thresholds for patient volume, including at least 50 and 100 live THA/TKA discharges per hospital.

The percent of all adult non-federal acute care hospitals penalized and the average penalty by state during fiscal year 2015 are listed in Table 3.1. This table represents data from all adult non-federal acute care hospitals in California, Florida, New Jersey and Pennsylvania, as well as all states nationally. Although not representative of surgical readmissions, the data reported here provide evidence for the generalizability of this study's sample to adult non-federal acute care hospitals nationally. During fiscal year 2015, hospital penalties ranged from 0% - 3%. On average, the hospitals in the states under study were penalized more than hospitals in all states nationally. The average penalties for hospitals in California (0.41%) and Florida (0.58%) were less than and equal to the average penalty of hospitals nationally (0.58%). Pennsylvania hospitals' average penalty (0.63%) was slightly higher than all states. New Jersey hospitals' average penalty (0.82%) was far greater than the average national penalty (0.58%). Despite state-to-national variation in the percent of hospitals penalized and the severity of the average penalty, when considered together, the four states in this study are not unlike all states



nationally. Of note is that California is the only state to have mandated staffing legislation, and also has the fewest percent of penalized hospitals and the lowest average

penalty compared with other states in the sample.

State	Percent of hospitals penalized	Average penalty
California	64%	0.41%
Florida	79%	0.58%
New Jersey	98%	0.82%
Pennsylvania	72%	0.63%
All States*	51%	0.58%

Table 3.1. Readmission penalties to be applied in fiscal year 2015, by state

Source: (Kaiser Health News, 2014); *excluding Maryland, which has a unique reimbursement agreement with Medicare

Nurses

The nurse sample was limited to registered nurses who worked in an adult nonfederal acute care hospital and who reported being a direct care staff nurse. Direct care nurses have the most frequent and direct contact with patients, placing them in a position to have a significant impact on patient care quality and safety.

Patients

The inclusion and exclusion criteria for patients in this study was based on the THA/TKA Readmission Technical Report, which is prepared for CMS by a study team at the Yale New Haven Health Services Corporation/Center for Outcomes Research and Evaluation (YNHHSC/CORE). This report is re-evaluated annually and updated as needed to refine the measures. The National Quality Forum has endorsed the THA/TKA readmission measure as an evidence-based and valid measure for performance assessment (QualityNet, 2014a).



Patients were included in the study if they were admitted to one of the study hospitals and met the following inclusion and exclusion criteria set forth in the THA/TKA Readmission Technical Report (Suter et al, 2014). These specified inclusion and exclusion criteria were followed because they are the established criteria used to calculate excess readmissions beyond the expected risk adjusted readmission rates for hospitals under the Hospital Readmission Reduction Program. Following these guidelines makes these study findings more relevant to hospital administrators and policymakers. The inclusion and exclusion criteria as well as the rationale for each criterion are described in Table 3.2 and Table 3.3. The number of THA/TKA patients excluded with each criterion can be found in the Appendix A.



Inclusion Criteria		Rationale	
1.	Enrolled in Medicare fee-for-	Hospital claims data are regularly available for	
	service	only for Medicare fee-for-service beneficiaries.	
2.	Age 65 or older	Medicare patients younger than age 65 qualify	
		for Medicare due to severe disability, making	
		them distinctly different from the elderly	
		Medicare population.	
3.	Discharged from a non-federal	Only those patients who are alive at time of	
	acute care hospital alive	hospital discharge are eligible for a	
		readmission.	
4.	Enrolled in Part A and Part B	Including Medicare Part A beneficiaries	
	Medicare for the 12 months prior	ensures there are no Medicare Part C	
	to the date of the index admission	(Medicare Advantage patients) in the data.	
		Enrollment in Medicare in the 12 preceding	
		months ensures one year of administrative data	
		for risk adjustment purposes.	
5.	Have a qualifying elective primary	Elective primary THA/TKA is the procedure	
	THA/TKA procedure, without any	of interest in this study.	
	of the following:		
	a. Femur, hip, or pelvic fractures	Procedures to correct an orthopedic fracture	
	coded in the principal or	are considered non-elective. Patients with	
	secondary discharge diagnoses	orthopedic fracture tend to have higher	
	fields of the index admission	mortality, complication, and readmission rates.	
	b. Partial hip arthroplasty	Partial hip arthroplasty are primarily indicated	
	procedures with concurrent	for hip fractures.	
	THA/TKA		
	c. Revision procedures with a	Few hospitals perform THA/TKA revision	
	concurrent THA/TKA	procedures and are associated with higher	
		mortality, complication, and readmission rates.	
	d. Resurfacing procedures with a	Resurfacing procedures are distinctly different	
	concurrent THA/TKA	than THA/TKA and are primarily indicated for	
		younger, healthier patients.	
	e. Mechanical complication of	A mechanical complication was likely present	
	the pelvis, sacrum, coccyx,	on admission and may require more	
	lower limbs, or bone/bone	technically complex procedures to correct.	
	marrow or disseminated	Patients with malignant neoplasms undergoing	
	malignant neoplasm coded in	a THA/TKA are likely not elective and the	
	the principal discharge	patients are more likely to have a readmission.	
	diagnosis field		
	f. Removal of implanted	Removal of an implanted device/prostheses	
	devices/prostheses	may be more complicated.	
	g. Transfer from another acute	Transfers from another acute care facility for	
	care facility for THA/TKA	THA/TKA is likely not elective	

Table 3.2. Inclusion criteria and rationale for the study of elective THA/TKA readmissions


Ex	clusion Criteria	Rationale				
1.	Without at least 30 days post-discharge	Since readmissions are identified using				
	enrollment in fee-for-service Medicare	claims data, 30 days of post-discharge				
		enrollment in Medicare fee-for-service is				
		required.				
2.	Discharged against medical advice	Patients leaving AMA may not allow				
	(AMA)	providers to deliver complete and full care				
		to prepare the patient for discharge.				
3.	Admitted for the index procedure and	Including these cases in the readmission				
	subsequently transferred to another	measure makes it difficult to determine to				
	acute care facility	which hospital the readmission outcome				
		should be attributed.				
4.	With more than two THA/TKA	More than two THA/TKA procedures				
	procedure codes during the index	likely reflects an error in coding.				
	hospitalization					

Table 3.3. Exclusion criteria and rationale for the study of elective THA/TKA readmissions

For patients who had multiple admissions characterized as a qualifying admission in the study timeframe, one qualifying admission was randomly selected for analysis (Suter et al, 2014). A qualifying admission was characterized as an index admission if an unplanned readmission occurred within 30 days of the index admission discharge (Suter et al, 2014). Randomly selecting one qualifying admission allowed for statistical independence of observations, such that a patient's prior readmission would not influence a future hospitalization. After this selection process, the sample included a total of 124,300 patients (36,745 THA patients; 87,555 TKA patients).

Measurement and instrument

Outcome

The outcome of interest in this study was unplanned readmission 30 days from discharge. 30 days from discharge is a widely used timeframe in the literature and is designated as the clinically meaningful timeframe in which hospitals should be held accountable for the care outcomes of patients beyond hospitalization, according to the



THA/TKA Readmission Technical Report (Suter et al, 2014). In light of criticism that a 30 day timeframe may not be an appropriate assessment of hospital quality (Joynt & Jha, 2012; Vaduganathan, Bonow, & Gheorghiade, 2013), a sensitivity analysis of unplanned readmission 10 days from discharge was also performed.

The THA/TKA Readmission Technical Report outlines what qualifies as a readmission that can be used to calculate a hospital's readmission rate. A qualifying readmission is defined by a readmission that occurs within 30 days of discharge from an index admission. Any admission that occurs within 30 days of discharge cannot be considered as an index admission. An index admission is defined based on the inclusion and exclusion criteria as previously described (Suter et al, 2014).

Unlike prior readmission policies for medical conditions, which penalized hospitals for all cause readmissions, readmissions for THA/TKA are delineated as either planned or unplanned readmissions. Based on the theory that hospitals should not be penalized for having a patient return to the hospital for a planned readmission, only unplanned readmissions were used to calculate a hospital's readmission rate. The THA/TKA Readmission Technical Report provides a detailed algorithm (Figure 3.1) to distinguish planned and unplanned readmissions. The algorithm is based on condition categories which are clusters of International Classification of Diseases, Ninth Revision, Clinical Modification (ICD-9-CM) codes, as well as individual ICD-9-CM codes (Suter et al, 2014; QualityNet, 2014b). The condition categories are derived from the Hierarchical Condition Categories system, which classifies condition categories into clinically meaningful categories at a more aggregated level (Pope et al, 2000). The Clinical Classifications Software is a tool developed by the Agency for Healthcare



60

Research and Quality and is used to clinically group diagnoses and procedures to aid in creating manageable condition categories (HCUP, 2012).



Figure 3.1. Planned and unplanned readmission algorithm

Algorithm obtained from: Suter et al, 2014



For the patients in the sample that had multiple unplanned readmissions within 30 days, the first readmission was identified as the unplanned readmission.

After merging the patient sample with the nurse and hospital data, the final analytic sample included 112,018 patients (33,155 THA patients; 78,863 TKA patients). The final count of unplanned readmissions was 7,524 (2,204 THA patients; 5,320 TKA patients). 28 patients who had both a THA and TKA on the same admission were excluded. The final sample also included 23,089 direct care registered nurses who reported working in the 495 study hospitals.

Covariates

Nurse staffing and the practice environment were the main predictors of interest in this study. Other covariates discussed in this section relate to hospital characteristics and patient demographics and were included for risk adjustment in the regression models for the specific aim and/or the sensitivity analysis.

Predictors of interest

Nurse staffing

The Multi-State Nursing Care and Patient Safety Study survey asked nurse respondents to report how many patients were on the nurse's unit and how many registered nurses provided direct patient care on the last shift the nurse worked. Using these data, I created a continuous variable of the average number of patients per registered nurse within each study hospital. For descriptive purposes, this continuous variable was categorized into the following categories representing the number of patients per registered nurse: <4; $\geq 4 \& <5$; $\geq 5 \& <6$; $\geq 6 \& <7$; ≥ 7 . For the regression



analyses, a continuous measure of the hospital aggregated number of patients per nurse was used. This approach has been previously used in the literature on nursing's effect on readmissions following admission for medical conditions and mortality outcomes for postsurgical patients (Brooks Carthon et al, 2012; McHugh & Ma, 2013).

Practice environment

The Multi-State Nursing Care and Patient Safety Study survey included the Practice Environment Scale of the Nurse Work Index (PES-NWI) a survey instrument to measure attributes that comprise the nurse practice environment. The PES-NWI was developed using exploratory factor analysis, which identified five subscales representing the five domains of the practice environment: 1) nurse participation in hospital affairs (9 items); 2) nursing foundations for quality of care (10 items); 3) nurse manager ability, leadership, and support of nurses (5 items); 4) staffing and resource adequacy (4 items); 5) collegial nurse–physician relations (3 items) (Lake, 2002). These five subscales, both independently and as a composite measure, were found to be psychometrically sound and have been tested on data from staff nurses in Pennsylvania hospitals (Lake, 2002).

In this study, the responses from each nurse respondent were aggregated within hospitals to create hospital level composite measures of the practice environment. This composite measure was created as the mean of the five subscale scores (Lake, 2002). A composite quartile approach was used to categorically describe the practice environments across hospitals. Hospitals ranking in the top 25th percentile on the practice environment scale were referred to as a "good" practice environment hospital. Hospitals in the middle 50th percentile on the practice environment scale were referred to as a "mixed" practice



63

environment hospital. The bottom 25th percentile were referred to as a "poor" practice environment hospital. In the regression analyses, I modeled the effect of a "good" practice environment, leaving "mixed" and "poor" practice environments as the reference category. This approach allows for ease in interpretation such that the good practice environment hospitals are compared to all other hospitals.

Hospital covariates

Bed size

The AHA Annual Survey provides data on hospital bed size. For descriptive and analytic purposes, bed size was categorized as follows: small (≤ 100 beds); medium (101-250 beds); large (>250 beds) (Brooks Carthon et al, 2012; Friese et al, 2008). For the analysis, each bed size category was dichotomized, with large hospital bed size as the reference category.

Teaching status

Teaching status was defined by the ratio of medical fellows and medical residents to the number of hospital beds. This data was derived from the AHA Annual Survey. Teaching status was categorized as follows: non-teaching hospital (no medical fellows or residents); minor teaching hospital (\leq 1:4); major teaching hospital (>1:4) (Ayanian & Weissman, 2002; McHugh & Ma, 2013).

Technology status

For this study, a high technology hospital was defined as a hospital that had the capacity to perform open-heart surgery and/or major organ transplantation (McHugh &



Ma, 2013). If a hospital was not able to perform either of these services, they were categorized as a low technology hospital. This data was available from the AHA Annual Survey. For analytic purposes, technology status was an indicator variable, with low technology status as the reference category.

Patient covariates

Age

Data on patient age was available from the MedPAR RIF. Patient age (years) was descriptively and analytically modeled as a continuous variable.

Sex

Data on patient sex was available from the MedPAR RIF. Patient sex was descriptively and analytically modeled as a categorical variable, with male sex as the reference category in the analyses.

Comorbidities

The MedPAR RIF provided information on patient comorbidities. The THA/TKA Readmission Technical Report delineates which specific comorbidities are meaningful indicators of patient frailty and have strong relationships with readmission (Suter et al, 2014). Each comorbidity was created as a dichotomous variable, with "0" indicating the patient did not have the comorbidity and "1" indicating the presence of the comorbidity.



Type of procedure

The THA/TKA Readmission Technical Report, controls for the type of procedure the patient undergoes (Suter et al, 2014). For the purposes of this study, the procedure types were categorized by the procedure codes on admission: THA or TKA. For descriptive purposes, the patient, hospital and nursing characteristics were described by procedure type. For analytic purposes, the effects of the predictors of interest on readmission included both THA and TKA patients together. In other words, separate regression analyses were not run for each procedure type. This is consistent with the THA/TKA Readmission Technical Report and Hospital Readmission Reduction Program policy (Suter et al, 2014).

Number of procedures

The THA/TKA Readmission Technical Report controls for the number of procedures the patient undergoes on the index admission (Suter et al, 2014). Because patients were excluded from the analysis if they had more than two procedures on the index admission, the number of procedures was dichotomized as either one or two procedures. The number of procedures undergone during the hospitalization was available from the MedPAR RIF.

Statistical analysis

The following describes the data software and statistical analytic plan that were used to meet the specific aim of the study, as well as to conduct the sensitivity analysis.



Data analysis software

The data obtained from the Multi-State Nursing Care and Patient Safety Study survey was received by CHOPR in a STATA Version 10.0 file. Within this file, the AHA Annual Survey data had been merged with the nursing data. The file was then converted into a SAS file to systematically check for duplication errors. The MedPAR RIF patient data was received in a SAS file and restricted to include only patients who had a THA and/or TKA procedure. The exploration and analysis of the data was conducted in STATA Version 13.0 (StataCorp, 2013).

Analytic plan

Figure 3.2 diagrams how the three data sources were linked together and from which dataset each of the variables for the specific aim was obtained.







Aim: To examine the extent to which hospital nursing – staffing and the practice environment – are associated with odds of 30 day readmission in a Medicare population undergoing elective total hip and total knee arthroplasty.

First, the study sample was described. In the following descriptive statistics, continuous variables were presented as the mean and standard deviation. Categorical variables were presented as the frequency and percent of the total sample. The characteristics of patients in the sample were described demographically by surgery type (THA, TKA). The most common comorbidities, calculated based on condition categories, were described. Condition categories rather than ICD-9-CM codes were used because they categorize ICD-9-CM codes into more clinically meaningful groups, thus enabling a more comprehensive understanding of the most common comorbidities. The ten most common reasons for readmission, defined by ICD-9 diagnoses and procedures, were described. The sample was compared on patient demographics between patients who were and were not readmitted within surgery types. Comparisons of the variables were made using analysis of variance and t-tests. Among the patients who were readmitted, the characteristics of the readmission event, such as days until readmission and length of stay upon readmission, were described. Additionally, Kaplan-Meier plots and a log-rank test for equality of survivor function were used to identify if there were significant differences in the median time-to-readmission between surgery types.

Additionally, the patient sample, including those patients who died during the initial hospitalization, was examined to assess for competing risk. Dying reduces an individual's probability of readmission to zero; therefore, solely examining readmissions as an indicator of hospital care quality may produce biased findings due to the competing



68

risk of mortality on readmissions (Gorodeski, Starling, & Blackstone, 2010; Press et al, 2013; Satagopan et al, 2004). Moreover, evidence suggests that hospital readmission rates and mortality rates are negatively correlated such that hospitals with low readmission rates tend to have high mortality rates and vice versa (Gorodeski, Starling, & Blackstone, 2010; Press et al, 2013). These correlations do not necessarily suggest conflicting evidence of hospital quality, but rather elucidate the issue of competing risk. Less than 0.5% of patients died within 30 days of admission, suggesting that mortality did not pose a meaningful competing risk to readmission in this study.

Hospital characteristics by hospital and patient were examined, including histograms of the distribution of patients across hospitals and readmission rates across hospitals. Finally, a comparison of patients who were readmitted versus not readmitted, within surgery type, were compared on the characteristics of the hospitals in which they were initially admitted.

Next, the distribution of registered nurses across the study hospitals were depicted in a histogram. The distribution of both hospitals and patients (by surgery type) were described by nursing characteristics (staffing and the practice environment). Then, a comparison of patients who were readmitted versus not readmitted, within surgery type, were compared on the nursing characteristics in the hospitals to which they were initially admitted.

The final descriptive statistics included a correlation matrix of the hospital and nursing characteristics used in the analysis of the specific aim. A weak correlation was defined by r = -0.3 - 0.3; a moderate correlation was defined by r = -0.6 - -0.3 or r = 0.3 - 0.6; a strong correlation was defined by r = -1.0 - -0.6 or r = 0.06 - 1.0.



69

The specific aim – an analysis of the effect of hospital nursing on 30 day readmission – was accomplished by estimating logistic regression models. Robust standard errors (or Huber-White sandwich estimators) were used to account for the clustering of patients within hospitals (Fitzmaurice, Laird, Ware, 2011; Vittinghoff et al, 2012). Clustering of patients within hospitals is important to account for because patients treated in the same hospital are more likely to be demographically alike and more likely to be treated similarly than patients in different hospitals. Failing to account for this correlation between patients in the same hospital could result in: poorly estimated standard errors (too small), confidence intervals (too narrow), p-values (too small) (Fitzmaurice, Laird, Ware, 2011; Vittinghoff et al, 2012). The level of significance at which the null hypothesis was rejected was $\alpha < 0.05$ for a two-tailed test.

The first model consisted of an unadjusted bivariate model or simple logistic regression. No covariates were included in the analysis. This bivariate analysis provides a baseline understanding of the likelihood of the event in relation to the predictor (either staffing or the practice environment), prior to accounting for patient or hospital characteristics that may affect an individual's odds of the event. These unadjusted separate models, estimate the independent effects of each of the primary predictor variables on the readmission outcome, without controlling for the other primary predictor variable.

Equation 1 depicts the form of a simple logistic regression:

(1)
$$\log\left(\frac{p_i}{1-p_i}\right) = \beta_0 + \beta_1 x_{i1}$$



where β_0 represents the intercept of the slope of the odds ratio, and where β_1 represents the odds ratio, or likelihood of experiencing the outcome, given explanatory variable *x*.

The second and third levels of analyses were multivariate logistic regression models, controlling for patient covariates, and then hospital and patient covariates, respectively. The effects of the main explanatory variables of interest – staffing and the practice environment – were modeled independently of each other, or separately. Sequentially building the models in this way allowed for the ability to identify the extent to which each level of analysis had an impact on the outcome.

Equation 2 depicts the form of multivariable logistic regression:

(2)
$$\log\left(\frac{p_i}{1-p_i}\right) = \beta_0 + \beta_1 x_{i1} + \beta_2 x_{i2} + \dots \beta_p x_{ip}$$

where each β_i represents a covariate in the risk adjustment. In the first three models, the effects of the nursing primary predictor variables were modeled separately.

In the fourth and final model, the effects of staffing and the practice environment were examined jointly. In the joint model, the effect of primary predictors of interest on the readmission outcome control for the patient and hospital covariates in addition to adjusting for the effects of each other.

All of the models were assessed for model goodness-of-fit by computing the area under the receiver operating characteristic (ROC) curve (c-statistic). C-statistics range from 0.5 (probability of predicting the outcome is no better than chance) to 1.0 (predicts the outcome perfectly), with 0.7 indicating a reasonable fit and 0.8 indicating a strong fit (Hosmer & Lemeshow, 2001).



Sensitivity analysis plan

An extensive sensitivity analysis was performed to address conflicting findings in the literature related to which patient and hospital characteristics have an effect on readmission odds. One set of sensitivity analyses explored whether variation in the hospital volume of live THA/TKA discharges affected the findings. This analysis was warranted because the THA/TKA Readmission Technical Report does not specify a minimum number of discharges needed per hospital to be at risk for penalty by CMS. Analyses for the specific aim were conducted for hospitals with at least 10 live THA/TKA annual discharges, to be conservative. The sensitivity analysis explored the extent to which restricting the analytic sample to hospitals with at least 50 annual discharges and at least 100 annual discharges changed the findings.

Additional sensitivity analyses used the outcome of 10 day unplanned readmission, rather than 30 day unplanned readmission, to address the common criticism that 30 days is not an appropriate timeframe for assessing hospital quality (Joynt & Jha, 2012; Vaduganathan, Bonow, & Gheorghiade, 2013). 10 days was chosen as the cut-off point because it was approximately the median time-to-readmission for both THA and TKA patients.

Finally, the sensitivity analyses examined the effect of specific patient and hospital covariates on readmission. The patient covariates included: discharge destination, length of stay on the index admission, race/ethnicity, and patient socioeconomic status. The hospital covariates included: geographic location, caseload volume, ownership type, socioeconomic status profile, Magnet® designation, and surgical care improvement project performance measures. These covariates were chosen



72

based on the conflicting findings in the literature about the effects of the variable on readmission and/or because of controversial debate about whether the variable should be adjusted for in the readmission rate calculations. The sensitivity analyses show the unadjusted effect of the specific additional covariate on the outcome and the effects of the primary predictors of interest, adjusting for all patient and hospital covariates, including the additional covariate. Descriptions of how each of these additional covariates were analyzed in the sensitivity analysis can be found in the Appendix B.

Data integrity plan

The data analyzed for this study were retained on a secured computer server maintained by the Office of Technology and Information Systems (OTIS) at the University of Pennsylvania in the School of Nursing. OTIS was responsible for the nightly and weekly backing up of all computer-generated information, which is stored in a secure off-site location. Firewalls, antiviral software, patches, and other software updates were routinely updated on the School of Nursing's computer system. Specifically, the data used in this study were analyzed on my password protected computer in a locked office located in CHOPR.

Human subjects

This study was a secondary analysis of existing data that had previously received Institutional Review Board (IRB) approval. Although the nurse survey contains sensitive information regarding the nurse's perception of their hospital quality and safety, the nurse's identifying information had been detached from the survey before returning to the sender. Thus, the nurse survey data are de-identified. The nurse respondents were able to consent to the study by completing and returning the survey. A detailed consent form, a



Participants Bill of Rights, and the contact information of the study team were printed on the front of the survey explaining the consent process. The Multi-State Nursing Care and Patient Safety Study (IRB Protocol #176400) received exempt status upon evaluation by the University of Pennsylvania IRB. The study continues to undergo continuing review and was most recently approved on March 20, 2014. Additional IRB approval for this particular study was obtained and considered exempt (IRB Protocol #821910).



CHAPTER 4: RESULTS

Introduction

The purpose of this study was to examine the extent to which hospital nursing – staffing and the practice environment – was associated with 30 day readmission in a Medicare population undergoing elective THA/TKA. This chapter describes the characteristics of the sample, including the patients, hospitals, and nurses. The incidence of readmission 30 days from discharge is described. Then, the findings from multiple logistic regression analyses which address the specific aim of this study are described. The main findings from the sensitivity analysis will be briefly described. A detailed explication of the sensitivity analysis is available in the Appendix C.

Characteristics of the sample

Patients

The final sample included 112,018 Medicare patients (33,155 patients underwent THA; 78,863 patients underwent TKA). Characteristics of the patient sample, by surgery type, are shown in Table 4.1. Both the THA and TKA patients had a similar incidence of readmission within 30 days of discharge (5.61% and 5.66%, respectively). THA patients were slightly older on average (75.9 years) than TKA patients (75.3 years). There were more females in the TKA patient sample (64.30%) than the THA patient sample (61.69%); and overall, the patient sample was substantially more female than the general population. TKA patients underwent two procedures on the same admission (6.59%) markedly more often than did the THA patients (0.61%). TKA patients had slightly more comorbid conditions on average (1.78), than THA patients (1.75). THA patients had



75

marginally longer lengths of stay (3.84 days) compared with TKA patients (3.75 days) and were less often discharged home (40.21%) compared to TKA patients (44.67%).

Detions above stavistics	TH	A	TKA n = 78,863		
Fatient characteristics	n = 33	,155			
	N/Mean	%/SD	N/Mean	%/SD	
No. of readmission 30 days after discharge	1,859	5.61	4,463	5.66	
Age (years), (mean, SD)	75.9	6.05	75.3	5.82	
Sex					
Male	12,702	38.31	28,153	35.70	
Female	20,453	61.69	50,710	64.30	
No. of procedures					
1	32,953	99.39	73,662	93.41	
2	201	0.61	5,201	6.59	
No. of comorbidities, (mean, SD)	1.75	1.29	1.78	1.27	
Length of stay (days), (mean, SD)	3.84	2.07	3.75	1.95	
Discharge destination					
Home or home with home healthcare	13,331	40.21	35,225	44.67	
Other facility	19,824	59.79	43,638	55.33	

Table 4.1.	Patient	characteristics,	by	surgery type
------------	---------	------------------	----	--------------

Note. Percentages may not sum to 100% due to rounding. Discharge to "other facility" includes e.g. inpatient rehabilitation and skilled nursing facility.

The common comorbidities of the study patients are shown in Table 4.2.

Hypertension was the most prevalent comorbid condition for both THA (63.33%) and

TKA patients (67.51%), followed by chronic arthrosclerosis / angina (THA 22.43%;

TKA 21.72%). The prevalence of the comorbid conditions varied only marginally

between the surgery types. The largest variation is noted for diabetes or diabetes mellitus

complications (THA 14.44%; TKA 20.27%).



Comorbidity	TH	A	ТКА		
Comorbidity	n = 33	,155	n = 78	,863	
	Ν	%	Ν	%	
Hypertension	20,998	63.33	53,242	67.51	
Chronic atherosclerosis or angina	7,437	22.43	17,129	21.72	
Diabetes or diabetes mellitus complications	4,790	14.44	15,986	20.27	
Specified arrhythmias	4,896	14.77	11,405	14.46	
Disorders of fluid/electrolyte/acid-base	3,305	9.97	7,080	8.98	
Chronic obstructive pulmonary disease	3,253	9.81	6,372	8.08	
Vascular or circulatory disease	3,160	9.53	6,353	8.06	
Major symptoms, abnormalities	1,671	5.04	3,629	4.60	
Congestive heart failure	1,416	4.27	3,423	4.34	
Renal failure	1,411	4.26	3,183	4.04	
Rheumatoid arthritis and inflammatory	1.0.40	275	2 000	250	
connective tissue disease	1,242	5.75	2,808	3.30	
Polyneuropathy	599	1.81	1,571	1.99	
Cancer	551	1.66	1,037	1.31	
History of infection	518	1.56	932	1.18	
Morbid obesity	483	1.46	1,633	2.07	
Other injuries	492	1.48	1,086	1.38	
Dementia or other specified brain disorders	481	1.45	947	1.20	
Major psychiatric disorders	223	0.67	701	0.89	
Pneumonia	235	0.71	565	0.72	
Hemiplegia, paraplegia, paralysis, functional disability	200	0.60	458	0.58	
Decubitus ulcer or chronic skin ulcer	160	0.48	181	0.23	
Skeletal deformities	129	0.39	≤10*		
Cellulitis, local skin infection	85	0.26	259	0.33	
Post traumatic osteoarthritis	76	0.23	127	0.16	
Severe hematological disorders	66	0.20	94	0.12	
Protein-calorie malnutrition	62	0.19	112	0.14	
Metastatic cancer or acute leukemia	56	0.17	70	0.09	
Stroke	46	0.14	74	0.09	
End-stage renal disease or dialysis	25	0.08	39	0.05	

Table 4.2. Common comorbidities of study patients, by surgery type

Note. Patient comorbidities are defined by condition categories (Pope et al, 2000). *Due to CMS' cell size suppression policy, numbers of 10 or less are not displayed.



The ten most common reasons for readmission, by surgery type, are shown in Table 4.3. Patients were readmitted for both medical and surgical reasons. The most common reason patients were readmitted following THA was for a packed cell transfusion (5.81%), followed by dislocation of the prosthetic joint (5.76%), and closed reduction of the dislocated hip (5.43%). Following TKA, patients were most commonly readmitted for a postoperative infection (6.39%), followed by localized osteoarthritis (5.60%) and packed cell transfusion (5.33%). Both THA and TKA patients were more often readmitted with a primary medical diagnosis than a procedural diagnosis. In sum, the ten most common reasons for readmission accounted for 39.76% of readmissions for THA and 38.94% for TKA patients.



Most common	ТНА	тка			
reason for	n = 1.859	n = 4.463			
readmission					
	N (%)	N (%)			
1	Packed cell transfusion (p)	Postoperative infection (m)			
1	108 (5.81)	285 (6.39)			
	Dislocation of prosthetic				
2	joint (m)	Osteoarthritis (m)			
	107 (5.76)	250 (5.60)			
	Closed reduction of				
3	dislocated hip (p)	Packed cell transfusion (p)			
	101 (5.43)	238 (5.33)			
4	Osteoarthritis (m)	Atrial fibrillation (m)			
4	81 (4.36)	175 (3.92)			
=	Postoperative infection (m)	Venous catheterization (p)			
5	73 (3.93)	151 (3.38)			
		Infection due to internal			
6	Atrial fibrillation (m)	joint prosthesis (m)			
	63 (3.39)	147 (3.29)			
	Intestinal infection due to				
7	C. diff (m)	EGD with closed biopsy (p)			
	57 (3.07)	139 (3.11)			
	Hematoma complicating				
8	procedure (m)	Arthrocentesis (p)			
	56 (3.01)	132 (2.96)			
		Intestinal infection due to			
9	Venous catheterization (p)	C. diff (m)			
	47 (2.53)	112 (2.51)			
10	EGD with closed biopsy (p)	Congestive heart failure (m)			
10	46 (2.47)	109 (2.44)			
Total	683 (36.76)	1,738 (38.94)			

Table 4.3. Ten most common reasons for 30 day readmission, by surgery type

Note. Reasons for readmission are defined by ICD-9 diagnoses and procedures. (m) and (p) designate medical ICD-9 codes and ICD-9 procedure codes, respectively. C. diff, *Clostridium difficile*; EGD, esophagogastroduodenoscopy



Approximately 6% of the study sample experienced an unplanned readmission within 30 days following discharge. The patients who experienced a readmission were significantly different than the patients who did not experience a readmission, as shown in Table 4.4. For both THA and TKA patient populations, patients who were older in age, male, had more comorbidities, longer lengths of stay on the index admission, and were discharged to a facility, were all significantly more likely to experience a readmission. TKA patients who underwent bilateral knee replacements on the same admission were significantly more likely to be readmitted than TKA patients who had only one knee replacement procedure. The number of procedures were not significantly different between THA patient who were and were not readmitted; however, this may be an effect of how few THA patients had bilateral hip replacements.



			THA			TKA					
Patient	Not						Not				
characteristics	Readr	nitted	Readm	nitted		Readr	nitted	Readm	nitted		
	N = 1	,859	N = 31	1,296		$\mathbf{N} = 4$,463	N = 74	,400		
	N/	%/	N/	%)	p-	N/	%/	N/	%/	p-	
	Mean	SD	Mean	SD	value	Mean	SD	Mean	SD	value	
Age (years)	77.55	6.35	75.81	6.02	< 0.001	76.58	6.06	75.23	7.80	< 0.001	
Sex											
Male	758	40.77	11,943	38.16	0.0100	1,770	39.66	26,383	35.46	-0.001	
Female	1,101	59.23	19,352	61.84	0.0122	2,693	60.34	48,017	64.54	<0.001	
No. of											
1	1,844	99.19	31,109	99.41	0.2514	4,005	89.74	69,657	93.63	-0.001	
2	15	0.81	186	0.59	0.2514	458	10.26	4,743	6.38	<0.001	
No. of comorbidities	2.23	1.38	1.73	1.28	< 0.001	2.21	1.37	1.76	1.26	< 0.001	
Length of stay (days)	4.50	3.45	3.81	1.95	< 0.001	4.33	3.36	3.71	1.83	< 0.001	
Discharge											
destination											
with home	448	24.10	12.883	41.17		1.249	27.99	33.976	45.67		
healthcare				,	< 0.001	-,>				< 0.001	
Other facility	1,411	75.90	18,412	58.83		3,214	72.01	40,424	54.33		

Table 4.4. Thirty day readmission by patient characteristics and by surgery type

Note. Percentages may not sum to 100% due to rounding. Analysis of variance and t-tests were used to compare differences across patients who were and were not readmitted.



The characteristics of the readmission event for those patients who were readmitted are described in Table 4.5. On average, THA patients remained out of the hospital roughly one day longer (11.30 days) than TKA patients (10.50 days). The length of stay on readmission was approximately 2 days longer than the average length of the stay on the index admission for both THA and TKA patients. The majority of THA (52.23%) and TKA (60.70%) patients were discharged home with or without home healthcare following the readmission. Notably, some of the patients who were readmitted died in the hospital upon readmission.

Patient abaractoristics	THA	4	ТКА		
I attent characteristics	n = 1,8	859	n = 4,463		
	N/Mean	%/SD	N/Mean	%/SD	
Days from discharge to	11.30	8.83	10.50	8.71	
readmission, (mean, SD)					
Length of stay (days), (mean, SD)	5.44	4.89	5.10	3.95	
Discharge destination					
Home or home with home healthcare	971	52.23	2,709	60.70	
Skilled nursing facility	586	31.52	1,125	25.21	
Inpatient rehabilitation	202	10.87	470	10.53	
Other facility	89	4.79	140	3.14	
Against medical advice	≤10*		<i>≤</i> 10 *		
Expired in hospital	≤10 *		13	0.29	

Table 4.5. Characteristics of the readmission event, by surgery type

Note. Percentages may not sum to 100% due to rounding. *Due to CMS' cell size suppression policy, numbers of 10 or less are not displayed.

Figure 4.1 shows Kaplan-Meier time-to-readmission curves for THA/TKA patients who were readmitted within 30 days of discharge. The median time-toreadmission was 11 days for THA patients who were readmitted and 10 days for TKA patients. A log-rank test for equality of survivor function revealed a statistically



significant difference in median time-to-readmission between THA and TKA patients (p

= 0.0030).



Figure 4.1. Time-to-readmission for THA/TKA patients who were readmitted, by surgery type

Note. Median time-to-readmission for patients experiencing a readmission event was 11 days from discharge following THA (n = 1,859) and 10 days following TKA (n = 4,463).

Hospitals

As described in Table 4.6, the final study sample included 495 acute care nonfederal hospitals in four states (California, New Jersey, Pennsylvania, and Florida). The study hospitals were distributed among the four states with approximately 37% of hospitals in California, 12% in New Jersey, 25% in Pennsylvania, and 26% in Florida. The majority of hospitals were large, urban, not-for-profit, non-teaching hospitals. Less than half (47.68%) of the study hospitals had a high-technology status, meaning they had the capacity to perform open-heart surgery and/or major organ transplantation. The patients were similarly distributed within the hospitals by hospital characteristics, except



that both THA and TKA patients more often went to a high technology status hospital on the index admission (THA 66.55%; TKA 64.40%).

Hospital	Hosp	ital	TH	A	TKA		
characteristics	n = 4	95	n = 33,155		n = 78	,863	
State	Ν	%	Ν	%	Ν	%	
California	182	36.77	10,446	31.51	23,948	30.37	
New Jersey	57	11.52	3,569	10.76	7,677	9.73	
Pennsylvania	126	25.45	7,115	21.46	18,284	23.18	
Florida	130	26.26	12,025	36.27	28,954	36.71	
Geographic location							
Urban	444	89.70	31,162	93.99	73.080	92.67	
Rural	51	10.30	1,993	6.01	5,783	7.33	
Ownership type							
For profit	93	18.79	5,270	15.90	13,440	17.04	
Not-for-profit	361	72.93	25,663	77.40	60,099	76.21	
Government, nonfederal	41	8.28	2,222	6.70	5,324	6.75	
High technology	236	47.68	22,063	66.55	50,785	64.40	
Hospital size							
Small	48	9.70	1,197	3.61	4,116	5.22	
Medium	222	44.85	9,978	30.10	25,517	32.36	
Large	225	45.45	21,980	66.29	49,230	62.42	
Teaching status							
Non-teaching	261	52.73	16,930	51.06	40,930	51.90	
Minor	203	41.01	13,195	39.80	32,207	40.84	
Major	351	6.26	3,030	9.14	5,726	7.26	

Table 4.6. Distribution of the hospital and patient study sample by hospital characteristics

Note. Percentages may not sum to 100% due to rounding.

Figure 4.2 shows the distribution of the combined THA/TKA patients across the 495 study hospitals. The mean number of THA/TKA patients admitted to a hospital on an index admission was 226 patients (SD 253) and ranged from 11-1,733 patients.



Figure 4.2. Distribution of THA/TKA patients across hospitals



Note. The sample includes 112,018 patients in 495 hospitals

The combined readmission rate for THA and TKA patients, varied across hospitals. The mean hospital readmission rate was 6.4% (SD 4.1%) and ranged from 0%-33.33%, as shown in Figure 4.3.



Figure 4.3. Readmission rate of THA/TKA patients across hospitals

Note. The sample includes 112,018 patients in 495 hospitals



As shown in Table 4.7, patients who experienced a readmission event were significantly different than the patients who did not experience a readmission, on a number of different hospital characteristics. For both THA and TKA patient populations, patients who were readmitted tended to be in urban hospitals. Patients in California and Florida hospitals were significantly less likely to be readmitted; while the opposite was observed for patients in New Jersey hospitals. Patients were also slightly more likely to be readmitted in Pennsylvania hospitals; however the difference was only marginally significant.

Hospital ownership type had only marginal differences between readmitted and not-readmitted THA patients. The differences were greater for TKA patients, who were more likely to be readmitted in a not-for-profit hospital and less likely to be readmitted in a for-profit hospital. TKA patients cared for in a high technology hospital were slightly more likely to avoid being readmitted. Finally, no differences between THA patients who were and were not readmitted were observed based on hospital bed size. Only marginal differences were observed in the TKA sample, with patients in smaller hospitals less likely to be readmitted. Finally, there was a significant trend in the association between hospital teaching status and readmission, such that patients in non-teaching hospitals were less likely to be readmitted, compared with patients in major teaching hospitals.



	THA						ТКА			
Hospital			No	t				No	ot	
characteristics	Readmitted		Readm	Readmitted		Readr	nitted	Readm	nitted	
	N = 1	,859	N = 31	,296		N = 4	1,463	N = 74	1,400	
	N	0/	N	0/	p-	N	0/	N	0/	p-
	IN	/0	1	/0	value	IN	/0	19	/0	value
California	497	26.73	9,948	31.79	< 0.001	1,115	24.98	22,833	30.69	< 0.001
New Jersey	327	17.59	3,242	10.36	< 0.001	721	16.16	6,956	9.35	< 0.001
Pennsylvania	430	23.13	6,685	21.36	0.0710	1,095	24.54	17,189	23.10	0.0277
Florida	605	32.54	11,420	36.49	0.006	1,532	34.33	27,422	36.86	0.007
Geographic location										
Urban	1,768	95.10	29,393	93.92	0.0270	4,190	93.88	68,890	92.59	0.0012
Rural	91	4.90	1,902	6.08	0.0372	273	6.12	5,510	7.41	0.0013
Ownership type										
For profit	266	14.31	5,004	15.99	0.0541	660	14.79	12,780	17.18	< 0.001
Not-for-profit	1,484	79.83	24,178	77.26	0.0101	3,526	79.01	56,573	76.04	< 0.001
Government, nonfederal	109	5.86	2,113	6.75	0.1366	277	6.21	5,047	6.78	0.0678
High technology	1,212	65.20	20,851	66.63	0.2040	2,796	62.65	47,989	64.50	0.0120
Hospital size										
Small	63	3.39	1,134	3.62	0.5982	179	4.01	3,937	5.29	0.0002
Medium	560	30.12	9,418	30.09	0.9785	1,442	32.31	24,075	32.36	0.9461
Large	1,236	66.49	20,743	66.28	0.8557	2,842	63.68	46,388	62.35	0.0748
Teaching status										
Non-teaching	904	48.63	16,027	51.21	0.0304	2,178	48.80	38,752	52.09	< 0.001
Minor	780	41.96	12,414	39.67	0.0499	1,946	43.60	30,261	40.67	< 0.001
Major	175	9.41	2,855	9.12	0.6722	339	7.60	5,387	7.24	0.3744

Table 4.7. Thirty day readmission by hospital characteristics and surgery type

Note. Percentages may not sum to 100% due to rounding. Analysis of variance and t-tests were used to compare differences across patients who were and were not readmitted.



Nurses

As shown in Figure 4.4, the final study sample included 23,089 registered nurses who reported working in one of the 495 nonfederal acute care hospitals included in this study. The mean number of registered nurses per hospital was 47 (SD 38), and ranged from 10-282 nurses.



Figure 4.4. Distribution of registered nurses across study hospitals

Each registered nurse survey respondent reported on the working conditions in the hospital where the nurse worked. Responses from nurses working in the same hospital were aggregated to create hospital level measures of hospital nursing characteristics. The results of this analysis are reported in Table 4.8 at the hospital and patient level. On average, nurses in this study cared for 5 patients on their last shift (SD 1). Of the 495 study hospitals, 183 hospitals (36.97%) had a staffing ratio of 4-5 patients per registered nurse. Nearly half of THA patients (44.57%) and TKA (42.07%) patients were cared for in hospitals with a staffing ratio of 4-5 patients per registered nurse. Although far fewer,



Note. The sample includes 23,089 registered nurses in 495 hospitals

some hospitals (4.85%) had staffing ratios of 7 or more patients per registered nurse, accounting for approximately 3% of the THA and TKA patients in the sample.

The largest proportion of hospitals was characterized as having a mixed (47.07%) or poor (33.54%) practice environment. The majority of THA (51.94%) and TKA (49.58%) patients were cared for in hospitals with a mixed practice environment. Approximately 20% of the hospitals in the sample were characterized as having a good practice environment. Approximately one quarter of the study patients were cared for in a hospital regarded as having a good practice environment.

Tuble net Distribution of the nospital and patient study sample by harsing enductoristics										
Nursing	Hospi	ital	TH	A	TKA					
characteristics	n = 4	95	n = 33	,155	n = 78,863					
	Ν	%	Ν	%	Ν	%				
Staffing										
<4	74	14.95	4,583	13.82	10,071	12.77				
4-<5	183	36.97	14,777	44.57	33,175	42.07				
5-<6	162	32.73	10,707	32.29	27,349	34.68				
6-<7	52	10.51	2,328	7.02	6,258	7.94				
7 +	24	4.85	760	2.29	2,010	2.55				
Practice environment										
Good	96	19.39	8,450	25.49	19,673	24.95				
Mixed	233	47.07	17,222	51.94	39,098	49.58				
Poor	166	33.54	7,483	22.57	20,092	25.48				

Table 4.8. Distribution of the hospital and patient study sample by nursing characteristics

Note. Percentages may not sum to 100% due to rounding. Staffing indicates a ratio of the number of patients to the number of registered nurses.

As shown in Table 4.9, patients who experienced a readmission were significantly different than the patients who did not experience a readmission, in terms of hospital nursing characteristics. In both THA and TKA patient samples, patients cared for by nurses with lower workloads were less often readmitted. Patients cared for in good practice environments were also less often readmitted; however the difference was not statistically significant for THA patients.



			THA					ТКА				
Nursing		Not						Not				
characteristics	Readmitted		Readn	Readmitted		Readr	nitted	Readmitted				
	N = 1	1,859	N = 31	1,296		N = 4	1,463	N = 74	1,400			
	N	%	N	%	p- value	N	%	N	%	p- value		
Staffing	11	70	11	70	vulue	11	70	11	70	vulue		
<4	235	12.64	4,348	13.89		530	11.88	9,541	12.82			
4-<5	766	41.20	14,010	44.77		1,779	39.86	31,396	42.20			
5-<6	620	33.35	10,087	32.23	< 0.001	1,684	37.73	25,665	34.50	0.0029		
6-<7	177	9.52	2,151	6.87		362	8.11	5,896	7.92			
7 +	61	3.28	699	2.23		108	2.42	1,902	2.56			
Practice environment												
Good	535	28.78	9,255	29.57	0.4657	1,144	25.63	21,501	28.90	< 0.001		
Mixed	1,019	54.81	17,002	54.33	0.6826	2,570	57.58	40,603	54.57	< 0.001		
Poor	305	16.41	5,038	16.10	0.7255	749	16.78	12,296	16.53	0.6554		

Table 4.9. Thirty day readmission by nursing characteristics and surgery type

Note. Percentages may not sum to 100% due to rounding. Analysis of variance and t-tests were used to compare differences across patients who were and were not readmitted.



The correlation matrix of nursing and hospital characteristics is shown in Table 4.10. Each of the five subscales of the PES-NWI were strongly to moderately correlated with each other, with correlations ranging from r = 0.54 (nurse participation in hospital affairs and collegial nurse-physician relations), to r = 0.88 (nurse participation in hospital affairs and foundations for quality of care). The staffing and resource adequacy subscale of the PES-NWI was moderately correlated with the direct measure of nurse staffing (r = -0.40). The moderate correlation and the conceptual overlap between the staffing subscale and the direct measure of staffing, suggest a reasonable argument for excluding the staffing subscale from the practice environment measure. Therefore, the analysis of the specific aim included all five subscales of the PES-NWI, and, in a sensitivity analysis, the staffing and resource adequacy subscale was excluded. The correlations among the other nursing and hospital characteristics were weak, with the exception of a moderate correlation between technology status and hospital bed size (r = 0.47).



		1	2	3	4	5	6	7	8	9
1.	Practice									
	environment									
2.	Staffing &									
	resource	0.71								
	adequacy									
3.	Foundations for	0.84	0.72							
	quality of care	0.01	0.72							
4.	Collegial									
	nurse-physician	0.70	0.61	0.61						
~	relations									
5.	Nurse	0.02	0.66	0.00	0.54					
	participation in	0.83	0.66	0.88	0.54					
c	hospital affairs									
0.	Nurse manager									
	addreship &	0.80	0.71	0.76	0.55	0.75				
	support of	0.80	0.71	0.70	0.55	0.75				
	nurses									
7.	Staffing	-0.26	-0.40	-0.29	-0.27	-0.26	-0.29			
8.	Bed size	0.09	-0.03	0.19	-0.03	0.14	0.03	-0.16		
9.	Teaching status	-0.03	-0.10	-0.04	-0.04	-0.05	-0.09	-0.08	0.27	
10.	Technology	0.00	0.06	0.17	0.05	0.14	0.06	0.27	0.47	0.19
	status	0.09	0.00	0.17	0.05	0.14	0.00	-0.27	0.47	0.10

 Table 4.10. Correlation matrix of hospital and nursing characteristics, hospital level (N=

 495)

Note. Weak correlation: -0.3 - 0.3; moderate correlation: -0.6 - -0.3 or 0.3 - 0.6; strong correlation: -1.0 - -0.6 or 0.06 - 1.0



Analysis of specific aim

The associations between hospital nursing and readmission were explored using logistic regression analysis, accounting for clustering of patients within hospitals. First, analyses were conducted in hospitals with at least 10 live THA/TKA patient discharges (Table 4.11). The analytic sample was then restricted to hospitals with (1) at least 50 (Table 4.12) and (2) at least 100 live THA/TKA patient discharges (Table 4.13), to test if the results were sensitive to patient volume effects.

Four levels of analysis were conducted. First, the bivariate, or unadjusted, association between 30 day readmission and each of the hospital nursing features (staffing and practice environment) were explored. Second, the associations were examined adjusting for patient characteristics, including: age, sex, comorbidities, type of surgery (THA vs. TKA), and number of procedures (1 vs. 2). In the third level of analysis, each association additionally adjusted for hospital characteristics, including: bed size, teaching status, technology status. Finally, the fourth model examined the effects of nurse staffing and the practice environment jointly.

The effects of nurse staffing and the practice environment on 30 day readmission in the 495 hospitals, before and after controlling for patient and hospital characteristics are described in Table 4.11. The unadjusted, separate model for staffing describes the effect of one additional patient per nurse on the odds of 30 day unplanned readmission. The unadjusted, separate model for practice environment describes the effect of being cared for in a good practice environment (as compared to a mixed or poor environment) on the odds of readmission.

The bivariate association of staffing was significantly associated with readmission in the hypothesized direction (OR 1.10, 95% CI 1.04-1.17). The relationship remained



93

significant after controlling for patient and hospital covariates (OR 1.08, 95% CI 1.02-1.15) and after controlling for the practice environment in the joint model (OR 1.07, 95% CI 1.00-1.14).

The bivariate association of the practice environment was significantly associated with readmission in the hypothesized direction (OR 0.83, 95% CI 0.73-0.94) and the relationship remained significant after controlling for patient and hospital covariates (OR 0.87, 95% CI 0.76-0.99). In the joint model, after adjusting for nurse staffing, the effect of the five subscale practice environment is rendered insignificant (OR 0.90, 95% CI 0.78-1.04). The c-statistic for the specific aim was 0.64, which is similar to the c-statistic of 0.65 reported in the THA/TKA Readmission Technical Report (Suter et al, 2014).


Odds of 30 day Readmission	Unadjusted, Separate	Adjusted for patient characteristics, Separate †	Adjusted for hospital & patient characteristics, Separate	Adjusted for hospital & patient characteristics, Joint
	OR (95% CI)	OR (95% CI)	OR (95% CI)	OR (95% CI)
- Staffing	1.10**	1.08**	1.08*	1.07*
Stannig	(1.04-1.17)	(1.02-1.15)	(1.02-1.15)	(1.00-1.14)
Practice	0.83**	0.86*	0.87*	0.90
environment	(0.73-0.94)	(0.76-0.97)	(0.76-0.99)	(0.78-1.04)
n < 0.05* n < 0.0	1** n < 0.001***			

Table 4.11. Effects of hospital nursing on 30 day readmission, in hospitals with at least 10 live THA/TKA discharges (N = 112,018)

p <0.05*; p<0.01**; p<0.001 † C-statistic 0.64

Patient characteristics include: age, sex, comorbidities, type of surgery, number of procedures. Hospital characteristics include: bed size, teaching status, technology status. All of the models account for clustering of patients within hospitals. This analysis contains 112,018 patients in 495 hospitals.



The analysis was restricted to hospitals with at least 50 live THA/TKA discharges, as shown in Table 4.12. This analytic sample consists of 108,906 patients in 396 hospitals. The association between staffing and readmission remain significant in the separate model adjusting for patient and hospital covariates, but are rendered insignificant, in the joint model adjusting for practice environment (OR 1.06, 0.99-1.14). Although statistically insignificant, the joint effect of staffing approaches significance and is in the hypothesized direction.

The bivariate effect of practice environment is statistically significant (OR 0.85, 95% CI 0.75-0.96), even after controlling for patient covariates (OR 0.87, 95% CI 0.77-0.99). The association is insignificant after controlling for hospital characteristics, and in the joint model; however, the odds ratio remains in the hypothesized direction.

The analysis was further restricted to hospitals with at least 100 live THA/TKA discharges, as shown in Table 4.13. This analytic sample consists of 103,080 patients in 318 hospitals. The findings for the practice environment reveal a similar pattern, such that after adjusting for hospital and patient characteristics, the effect becomes insignificant (OR 0.88, 95% CI 0.77-1.01). The magnitude of the staffing effect in the bivariate analysis is large and significant (OR 1.11, 95% CI 1.03-1.19) and remains significant after controlling for patient and hospital covariates (OR 1.09, 95% CI 1.02-1.17) and the practice environment (OR 1.08, 95% CI 1.08, 95% CI 1.00-1.16).



Odds of 30 day Readmission	Unadjusted, Separate	Adjusted for patient characteristics, Separate †	Adjusted for hospital & patient characteristics, Separate	Adjusted for hospital & patient characteristics, Joint
	OR (95% CI)	OR (95% CI)	OR (95% CI)	OR (95% CI)
	1.09**	1.07*	1.07*	1.06
Staring	(1.02-1.16)	(1.02-1.14)	(1.01-1.14)	(0.99-1.14)
Practice	0.85*	0.87*	0.88	0.99
environment	(0.75-0.96)	(0.77-0.99)	(0.77-1.01)	(0.79-1.05)
n < 0.05 * : n < 0.0	$1 * * \cdot n < 0 001 * * *$			

Table 4.12. Effects of hospital nursing on 30 day readmission, in hospitals with at least 50 live THA/TKA discharges (N = 108,906)

p <0.05*; p<0.01**; p<0.001

† C-statistic 0.64

Patient characteristics include: age, sex, comorbidities, type of surgery, number of procedures. Hospital characteristics include: bed size, teaching status, technology status. All of the models account for clustering of patients within hospitals. This analysis contains 108,906 patients in 396 hospitals.

Table 4.13. Effects of hospital nursing on 30 day readmission, in hospitals with at least
100 live THA/TKA discharges ($N = 103,080$)

Odds of 30 day Readmission	Unadjusted, Separate	Adjusted for patient characteristics, Separate †	Adjusted for hospital & patient characteristics, Separate	Adjusted for hospital & patient characteristics, Joint
	OR (95% CI)	OR (95% CI)	OR (95% CI)	OR (95% CI)
	1.11**	1.09*	1.09*	1.08*
Stannig	(1.03-1.19)	(1.01-1.16)	(1.02-1.17)	(1.00-1.16)
Practice	0.84*	0.87*	0.88	0.91
environment	(0.74-0.96)	(0.76-0.99)	(0.77-1.01)	(0.78-1.06)
·0.05* ·0.0	1 *** .0.001 ****			

p <0.05*; p<0.01**; p<0.001***

† C-statistic 0.64

Patient characteristics include: age, sex, comorbidities, type of surgery, number of procedures. Hospital characteristics include: bed size, teaching status, technology status. All of the models account for clustering of patients within hospitals. This analysis contains 103,080 patients in 318 hospitals.



Additional analyses were conducted to examine the effect of the practice environment with the staffing and resource adequacy subscale excluded from the measure. The results in Table 4.14 reflect the analyses of 495 hospitals with at least 10 live THA/TKA discharges. As anticipated, the effect of staffing remains significant in the joint model (OR 1.07, 95% CI 1.01-1.14). The four subscale practice environment is significantly associated with readmission in the bivariate association (OR 0.85, 95% CI 0.75-0.96) and after adjusting for patient covariates (OR 0.88, 95% CI 0.78-0.99). The effect of the practice environment becomes insignificant once adjusting for hospital characteristics; however, the association approaches significance in the hypothesized direction (OR 0.89, 95% CI 0.78-1.01).



Odds of 30 day Readmission	Unadjusted, Separate	Adjusted for patient characteristics, Separate †	Adjusted for hospital & patient characteristics, Separate	Adjusted for hospital & patient characteristics, Joint
	OR (95% CI)	OR (95% CI)	OR (95% CI)	OR (95% CI)
Staffing.	1.10**	1.08**	1.08*	1.07*
Stannig	(1.04-1.17)	(1.02-1.15)	(1.02-1.15)	(1.01-1.14)
Practice	0.85*	0.88*	0.89	0.92
environment	(0.75-0.96)	(0.78-0.99)	(0.78-1.01)	(0.79-1.06)
n < 0.05* n < 0.05	1** n < 0 001***			

Table 4.14. Effects of hospital nursing on 30 day readmission, 4 subscale practice environment (N = 112,018)

p <0.05*; p<0.01**; p<0.001 † C-statistic 0.64

Patient characteristics include: age, sex, comorbidities, type of surgery, number of procedures. Hospital characteristics include: bed size, teaching status, technology status. All of the models account for clustering of patients within hospitals. This analysis contains 112,018 patients in 495 hospitals.



Additional analyses were conducted to examine the association of staffing and the practice environment on readmissions occurring within 10 days of discharge. Although the Hospital Readmission Reduction Program penalties define 30 days from discharge as the meaningful timeframe for measuring hospital care quality, many readmission studies examine shorter timeframes from discharge to isolate whether the influence of readmission predictors is more pronounced in earlier versus later readmission. The following analyses explore the effects of hospital nursing on 10 day readmission because the median time-to-readmission for the readmitted patients in this sample was approximately 10 days. 967 THA patients (2.92%) and 2,538 TKA patients (3.22%) were readmitted within 10 days of discharge.

As shown in Table 4.15, there is a pronounced association of nurse staffing and 10 day readmission, even after adjusting for hospital and patient covariates (OR 1.12, 95% CI 1.02-1.23). This relationship remains significant after restricting to the 396 hospitals with at least 50 live discharges (Table 4.16; OR 1.11, 95% CI 1.01-1.23) and after restricting to the 318 hospitals with at least 100 live discharges (Table 4.17; OR 1.14, 95% CI 1.02-1.27). After controlling for the practice environment (five subscales), the staffing effect becomes insignificant in each of the models (Table 4.15, Table 4.16, Table 4.17). The effect of the practice environment is significant in the bivariate associations; however, adjusting for patient and hospital covariates renders the effects insignificant.



Odds of 10 day Readmission	Unadjusted, Separate	Adjusted for patient characteristics, Separate †	Adjusted for hospital & patient characteristics, Separate	Adjusted for hospital & patient characteristics, Joint
	OR (95% CI)	OR (95% CI)	OR (95% CI)	OR (95% CI)
Staffing	1.12* (1.03-1.22)	1.10* (1.01-1.20)	1.12* (1.02-1.23)	1.10 (1.00-1.22)
Practice	0.81*	0.84	0.85	0.89
environment	(0.68-0.98)	(0.70-1.10)	(0.70-1.03)	(0.72-1.09)

Table 4.15. Effects of hospital nursing on 10 day readmission, in hospitals with at least 10 live THA/TKA discharges (N = 112,018)

p <0.05*; p<0.01**; p<0.001***

† C-statistic 0.63

Patient characteristics include: age, sex, comorbidities, type of surgery, number of procedures. Hospital characteristics include: bed size, teaching status, technology status. All of the models account for clustering of patients within hospitals. This analysis contains 112,018 patients in 495 hospitals.

Table 4.16. Effects of hospital nursing on 10 day readmission, in hospitals with at least 50 live THA/TKA discharges (N = 108,906)

Odds of 10 day Readmission	Unadjusted, Separate	Adjusted for patient characteristics, Separate †	Adjusted for hospital & patient characteristics, Separate	Adjusted for hospital & patient characteristics, Joint	
	OR (95% CI)	OR (95% CI)	OR (95% CI)	OR (95% CI)	
Staffing	1.11*	1.09	1.11*	1.10	
	(1.01-1.23)	(0.99-1.20)	(1.01-1.23)	(0.99-1.22)	
Practice	0.82*	0.85	0.86	0.89	
environment	(0.68-0.99)	(0.71-1.03)	(0.70-1.04)	(0.72-1.10)	

p <0.05*; p<0.01**; p<0.001***

† C-statistic 0.64

Patient characteristics include: age, sex, comorbidities, type of surgery, number of procedures. Hospital characteristics include: bed size, teaching status, technology status. All of the models account for clustering of patients within hospitals. This analysis contains 108,906 patients in 396 hospitals.



Odds of 10 day Readmission	Unadjusted, Separate	Adjusted for patient characteristics, Separate †	Adjusted for hospital & patient characteristics, Separate	Adjusted for hospital & patient characteristics, Joint
	OR (95% CI)	OR (95% CI)	OR (95% CI)	OR (95% CI)
Stoffing	1.14*	1.12*	1.14*	1.12
Staring	(1.03-1.27)	(1.01-1.24)	(1.02-1.27)	(1.00-1.26)
Practice	0.82	0.85	0.85	0.90
environment	(0.67-1.00)	(0.70-1.03)	(0.69-1.05)	(0.72-1.12)
n < 0.05 * n < 0.01	$1 * * \cdot n < 0 0 0 1 * * *$			

Table 4.17. Effects of hospital nursing on 10 day readmission, in hospitals with at least 100 live THA/TKA discharges (N = 103,080)

p <0.05*; p<0.01**; p<0.001 † C-statistic 0.64

Patient characteristics include: age, sex, comorbidities, type of surgery, number of procedures. Hospital characteristics include: bed size, teaching status, technology status. All of the models account for clustering of patients within hospitals. This analysis contains 103,080 patients in 318 hospitals.



Sensitivity analysis

A sensitivity analysis was conducted, controlling for patient and hospital level covariates. All of the sensitivity analyses were conducted using the analytic sample of 112,108 patients in 495 hospitals with at least 10 live THA/TKA discharges. Patient level covariates used in the sensitivity included: discharge destination (home with/without homecare vs. other), length of stay on the index admission, race/ethnicity (white non-Hispanic vs. other), and socioeconomic status (high socioeconomic status vs. other). Hospital level covariates, including: geographic location (urban vs. rural), patient caseload volume, ownership type (profit, not-for-profit, vs. government nonfederal), socioeconomic status profile (hospitals in the highest decile of the proportion of patients with low socioeconomic status), Magnet® designation, and surgical care improvement project performance measures. Controlling for patient and hospital level covariates did not have a clinically meaningful impact on the associations between staffing and the practice environment on 30 day unplanned readmission. The tables and a more detailed explication of the findings are located in the Appendix C.

Summary

Nearly 6% older adults experience an unplanned rehospitalization following elective total hip and total knee arthroplasty. For every additional patient a nurse cared for, each patient had an 8% increase in the likelihood of readmission, even after adjusting for patient and hospital characteristics. Patients cared for in a better practice environment, as compared to a mixed or poor environment, had 13% lower odds of readmission. The effect of staffing was more pronounced for readmissions occurring with 10 days compared with 30 days, suggesting that hospital nursing has an important role to play in



reducing avoidable readmissions. The main findings from this study point to hospital nursing as a potential strategy for reducing readmissions among older adult orthopedic surgical patients.



CHAPTER 5: DISCUSSION

Introduction

This study found that hospital nurses who provide care at the bedside serve an important role in lowering the risk for readmission following discharge. This study examined the association between characteristics of hospital nurses – staffing and the practice environment – and older patients' odds of unplanned readmission following elective THA/TKA. This chapter discusses the main findings of the study and the limitations of the research. Finally, the implications of these findings and recommendations for future research are considered.

Main findings

Nurse staffing and the practice environment were significantly associated with 30 day unplanned readmissions for older adults following elective THA/TKA. Each additional patient per nurse was associated with an 8% higher odds of readmission (OR 1.08, 95% CI 1.02-1.15). This staffing effect was substantially larger than previous findings for surgical patients (Ma, McHugh, & Aiken, 2015), and similar to the staffing effects found among medical patients (McHugh & Ma, 2013). The effect of nurse staffing was more pronounced in regards to 10 day readmission as compared with 30 day readmission, suggesting that many readmissions following THA/TKA are strongly related to hospital care quality – specifically nursing care. Patients cared for in the best practice environments were 13% (OR 0.87, 95% CI 0.76-0.99) less likely to experience a readmission compared to patients in either a mixed or poor practice environment. Previous studies report an effect of the practice environment ranging between 3% for surgical patients and 10% for pneumonia patients (Ma, McHugh, & Aiken, 2015). The



main findings of this study suggest that better staffing and better practice environments are associated with lower odds of readmission – findings that are consistent with prior work (Joynt & Jha, 2011; Ma, McHugh, & Aiken, 2015; McHugh & Ma, 2013).

Although the effect of the practice environment was large (OR 0.87, 95% CI 0.76-0.99), once adjusting for nurse staffing and the practice environment simultaneously in the joint model, the effect of the practice environment became insignificant. Due to the moderate correlation between the staffing and resource adequacy subscale and the direct measure of staffing (r = -0.40), the subscale was excluded from the practice environment measure to test whether that particular subscale was driving the effect. The four subscale practice environment (without staffing and resource adequacy) was not significantly associated with readmission and did not detract from the pronounced effect of staffing in the joint model. This provided some evidence that the significance of the practice environment may be driven, in part, by the staffing and resource adequacy subscale.

An extensive sensitivity analyses revealed that even after controlling for additional patient and hospital characteristics the associations between hospital nursing and readmissions remained. Controlling for patient discharge destination had the most pronounced effect on the association of both staffing and the practice environment. The effect of staffing was reduced by two percentage points (OR 1.06, 95% CI 1.00-1.12) and the practice environment was rendered insignificant (OR 0.89, 95% CI 0.80-1.01). These findings are not entirely unexpected in consideration of prior evidence (Bini et al, 2009; Schairer et al, 2014; Tsai et al, 2013; Zmistowski et al 2013). These findings are interesting given that the majority of these postoperative patients go to either skilled nursing facility or inpatient rehabilitation upon discharge. Future research should



highlight the role that these facilities have on readmission among this population. In sum, the relationship between hospital nursing and readmission was found to be robust, even after adjusting for potentially confounding patient and hospital characteristics. A more detailed discussion of the findings from the sensitivity analysis can be found in the Appendix D.

Discussion of corollary findings

While readmissions in this population were prevalent, mortality following this elective surgery was found to be extremely rare. In this study, less than 0.5% of THA/TKA patients in the sample died within 30 days of admission. These findings are similar to those reported in the literature, suggesting that even in an older adult population, mortality following elective surgery is rare (Singh et al, 2011; Katz et al, 2004). These findings are not entirely surprising given that patients who undergo elective surgery are deemed sufficiently healthy to tolerate an invasive operation followed by arduous rehabilitation. Given the low incidence in this study, mortality was not considered to be a meaningful competing risk with readmission; however, it remains an important consideration for future research.

Among readmitted patients, comorbid conditions were more prevalent, compared to the patients who were not readmitted. This finding is consistent across the literature for older patients following THA/TKA (Khan et al, 2012; Mahomed et al 2003; Riggs et al, 2010; Saucedo et al, 2014; Schairer et al, 2014; Schairer, Vail, & Bozic, 2014). This finding yields important implications for clinical practice, including provider awareness of the risk factors for readmission. Knowing what characteristics predispose patients to



readmission helps providers identify the patients at greatest risk and allocate resources accordingly.

Patients were more frequently rehospitalized for a primary medical diagnosis than a procedural diagnosis. Although there is no consensus in the literature regarding the common reasons for readmission, many studies found that THA patients were more often readmitted for medical conditions (Cullen, Johnson, & Cook, 2006; Khan et al, 2012; Pugely et al 2013; Saucedo, 2014; Schairer et al, 2014; Vorhies et al, 2011; Vorhies et al, 2012). Interestingly, the most common reason THA patients in this study were readmitted was for a packed cell transfusion (5.81%) – a procedural diagnosis, which is likely indicative of postsurgical anemia due to hemorrhage. A recent study cited surgical site infection to be the most common reason for unplanned readmission following THA/TKA (Merkow et al, 2015). While postoperative infection was the most common reason for TKA patients (6.39%) it was the fifth most common reason THA/TKA patients (3.93%). In sum, the literature lacks a consensus on the reasons THA/TKA patients are readmitted.

In this study, the top ten reasons for readmission accounted for less than 40% of the observed readmissions, suggesting a diverse array of post-discharge issues. While many of the top reasons for readmission appear to be related to the surgery, readmissions for atrial fibrillation and congestive heart failure may be indicative of uncontrolled comorbid conditions. Given that readmitted patients have more comorbidities on average compared with non-readmitted patients, and patients are readmitted for conditions unrelated to the prior hospitalization, it is plausible that comorbid conditions may not be adequately addressed and monitored during the hospitalization. This begs the question



whether the care delivered in hospitals is too narrowly focused on the acute reason for admission, and whether this narrow focus contributes to readmissions.

Additionally, future research should examine how the care delivered in post-acute care facilities, such as inpatient rehabilitation and skilled nursing facilities, may contribute to the incidence of readmission. Considering the major reasons for readmission, it is reasonable to suspect that some of the postoperative complications, such as dislocation of the prosthetic joint, postoperative infection, or intestinal infection due to *Clostridium difficile*, could be attributed to the care quality is post-acute care facilities.

On average, patients who were readmitted had approximately two days longer lengths of stay on the readmission as compared to the average length of stay on the index admission. It is unclear to what extent reimbursement associated with diagnosis-related groups (DRGs) is related to patient length of stay, in this study. Hospitals are reimbursed on a DRG system, whereby hospitals are given a pre-specified reimbursement per diagnosis/procedure. Hospitals that discharge patients sooner, generally make a profit on the admission, compared to when patients are hospitalized for longer. Despite pay-forperformance policies that are meant to incentivize hospitals to prevent readmissions, hospitals may struggle with the perverse incentive of not preventing a readmission in order to earn an additional DRG payment for the return hospitalization. Many have argued that the current penalties are too small to meaningfully incentivize behavior, which implies that policymakers should consider increasing the penalty (Jha, 2013; Werner & Dudley, 2012; Werner et al, 2011). Alternatively, the reason for readmission



may warrant a larger DRG payment than the THA/TKA admission DRG, which may explain the longer lengths of stay on readmission.

Patients who were readmitted were also more likely to be discharged home than to a facility, as compared with their index admission. This suggests that hospitals may keep readmitted patients in the hospital longer to ensure they are medically stable enough to return home, thereby reducing the odds of a second readmission. Patients who are able to go home (with or without home care) at discharge, tend to be more medically stable and have adequate support systems to help the patient safely transition to independent living and self-care. In sum, it remains unclear how hospitals are responding to pay-forperformance penalties to reduce readmissions.

Various characteristics of the hospitals where patients received care were found to be associated with higher odds of readmission; however, these hospital characteristics are largely unmodifiable and therefore lack feasible strategies for improving health outcomes. Despite this study's findings and the literature, it remains unclear if and to what extent certain hospital characteristics are associated with readmissions. At this time, sufficient evidence is lacking to support any recommendations for risk adjusting on hospital structural characteristics for the purposes of reimbursement penalties.

Limitations

This study was not without limitations. The cross-sectional design limits the understanding of causal relationships. A longitudinal approach would be needed to support causal relationships. Although cross-sectional, the carefully selected covariates and extensive sensitivity analyses aided in risk adjusting for differences across patients and hospitals, allowing for more valid comparisons across groups. Moreover, the



hospitals in the four states included in this study are generalizable to nonfederal acute care hospitals, nurses, and patients nationally.

Another limitation is that the data used for this study were collected in 2005-2006, which is not contiguous with the introduction of readmission penalties under the Affordable Care Act, which began in fiscal year 2013. This limitation, while worth mentioning, is negligible because the predictor variables of interest – nurse staffing and the practice environment – have likely not changed significantly since the data collection. The penalties levied against hospitals only represent a small fraction of CMS reimbursements and, for many hospitals, are too small to incentivize immediate and meaningful changes in practice (Jha, 2013; Werner & Dudley, 2012; Werner et al, 2011).

One concern regarding the nurse survey is that nurses may not give reliable reports of their working conditions. It is reasonable to assume that the nurse respondents had no incentive to be dishonest on the survey because it was confidential and mailed to the nurse's home address rather than place of work. Furthermore, evidence suggests that nurse reported quality of care is a valid predictor of hospital performance (McHugh & Witkoski Stimpfel, 2012). The nurse survey used in this study constitutes a unique data set incomparable to other existing data about registered nurses working in hospitals.

Both the incidence of readmissions and their association with hospital nursing may be underestimated in this study. In the 12 month study period, only one index admission per individual and only the first readmission to occur within 30 days of the index admission were selected for analysis. This was done to ensure statistical independence of observations, because a patient who was readmitted previously would be more likely to be readmitted in the future.



A final point to consider when interpreting the study findings is that logistic regression assumes a linear relationship between the predictor variable and the log odds, such that the log-odds represents a one-unit increase in the predictor variable. In other words, the odds ratio increases exponentially with a one-unit increase in the predictor variable. This is a strong assumption because, for example, adding another patient to a nurse's workload of six patients may have less of an effect than adding another patient to a nurse's workload of four patients, or vice versa. Prior work related to nurse staffing's effect on mortality found no evidence that the relationship is nonlinear (Aiken et al, 2002). In this study, staffing ratios ranged from 2.93 to 9.79 patients per nurse (mean 5.1; SD 1.04). Two additional patients per nurse was associated with a 17% increase in the patients' likelihood for readmission (1.08*1.08 = 1.17). Extrapolation of the odds ratios, in this way, should not extend beyond the staffing ranges observed in this study and should be interpreted with caution.

Implications

These findings have important implications for patients, providers, and policymakers, alike. For many older adults, recurrent hospitalization reduces functional independence and quality of life (Boltz, Capezuti, & Shabbat, 2011; Boltz et al, 2012; Kortebein et al, 2008; Ponzetto et al, 2003). Although the recent attention to readmissions has been brought about as a matter of the financial implications of the Hospital Readmission Reduction Policy, the human consequences of hospitalization should also be stressed.

The study findings that many hospital nursing characteristics are associated with readmission suggest that readmissions following THA/TKA can be reduced. Nurse



staffing had a more pronounced effect on readmission within 10 days as compared to 30 days, suggesting that greater attention to the patient care delivered *in the hospital* may significantly reduce an older adult's likelihood of readmission.

These findings are supported by decades of research that better hospital working conditions, including staffing and the practice environment, are associated with better patient outcomes (Kane et al, 2007; Kazanjian et al, 2005; Shekelle, 2013). More recently, studies have begun to show that better hospital nursing conditions are associated with fewer readmissions (Joynt & Jha, 2011; Ma, McHugh, & Aiken, 2015; Ma, Shang, & Stone, 2014; McHugh, Berez, & Small, 2014; McHugh & Ma, 2013; Weiss, Yakusheva, & Bobay, 2011). These study findings add to the existing evidence and have important implications for hospitals which are now at risk for CMS reimbursement penalties. Despite robust evidence, many hospitals have not moved the needle on increasing investments in hospital nursing.

To date, hospitals have invested in numerous disease-specific interventions aimed at reducing readmissions, yet these interventions yield mixed results in practice (Hansen et al, 2011). While few of these interventions have proven successful in particular patient populations, most are short-sighted and may be unsustainable over the long term. Strategies that are dependent upon existing staff to carryout additional care tasks (eg. telephone follow-up calls), or specialized personnel (eg. patient care coordinators), add additional complexity to an already complex system. Staff nurses report missing aspects of nursing care due to a lack of time (Kalisch, 2006; Kalisch, Gay & Williams, 2009) which is associated with readmissions (Brooks Carthon et al, 2015). While specialized personnel may theoretically reduce the number of tasks a staff nurse is responsible for,



adding specialized caregivers further fragments an already fragmented system. Without robust evidence that such strategies are effective across various patient populations, it is plausible that these strategies may actually impede care delivery efficiency and negatively impact patient outcomes.

Hospital administrators, in particular, should be cautious of the effects of reducing investments in hospital nursing in an effort to cut expenses. Although there has been no public discussion about whether the CMS readmission penalties will continue to increase over time, it seems as though reimbursement from both public and private payers has made an irreversible shift towards reimbursing for value, rather than volume. Whether the financial incentives are strong enough to encourage hospitals to reconsider their structural organization of care delivery has yet to be seen; however, these findings add to the growing evidence that improving investments in hospital nursing characteristics may be a reasonable strategy to reduce readmissions.

As pay-for-performance becomes more pervasive in healthcare, providers will be increasingly incentivized to deliver value across the entire spectrum of health conditions and care settings. For example, the Bundled Payments for Care Improvement initiative, which began in January 2013, incentivizes better care quality at a lower cost to Medicare by allowing hospitals to voluntarily bear the financial risk for a specified episode of care. Among these episodes of care is joint replacement. This initiative represents an early sign that provider reimbursement will increasingly move beyond penalties and incentives for preventing negative outcomes, such as unplanned readmissions, and target high value comprehensive care across the entire spectrum of healthcare delivery.



In light of the changing reimbursement climate, it is important to recognize that hospital care is often myopically focused on the acute issue requiring hospitalization. Yet for many older adults, comorbid conditions and complicated social needs demand a holistic approach to care. Strategies that consider patient conditions in isolation fail to appreciate the complex context of the individual patient. Registered nurses are uniquely trained in the "protection, promotion, and optimization of health and abilities" and "prevention of illness and injury" (ANA, 2015). The essence of nursing requires a holistic approach that seeks to identify and address the needs and goals specific to each individual. In doing so, nurses are able to help individuals optimize quality of life and wellness.

However, hospital investments to improve patient outcomes often fail to optimally utilize their most numerous healthcare providers – nurses. For example, investments in system level hospital nursing features would likely yield better outcomes across healthcare settings and clinical conditions, as compared to the commonly implemented disease-specific approaches. With hospitals already investing a significant proportion of their budget in nursing staff (Kane & Siegrist, 2002), there should be heightened attention to ensure that staff nurses have the time and resources to optimize their effectiveness. Moreover, every hospitalized patient has a nurse; whereas only certain patients receive the additional care interventions characteristics of many of the current readmission reduction efforts. Hospital nursing care should be considered as a building block upon which additional interventions could be added. The impact of any additional interventions beyond hospital nursing care is contingent on having a solid foundation of care delivery. Early evidence suggests that hospital nursing may be a high-



value investment (Martsolf et al, 2014; Weiss, Yakusheva, & Bobay, 2011; Yakusheva, Lindrooth, & Weiss, 2014a; Yakusheva, Lindrooth, & Weiss, 2014b). As pay-forperformance initiatives encompass more aspects of care across the healthcare setting, staff nurses may further prove to be a valuable asset.

Hospitals can capitalize on the opportunity to improve nursing care in multiple ways. This study provides evidence that increasing nurse staffing may reduce readmissions. This can be achieved in numerous ways, such as an individual hospital's commitment to better staffing, or through national or state legislation, such as mandated staffing ratios. Hospitals can improve the nurse practice environment by earning Magnet® recognition through the ANCC or the ANCC Pathway to Excellence® (ANCC, 2014). The Pathway to Excellence®, like Magnet®, has a focus on nurse autonomy, safety, quality, professional development, and leadership; which are closely aligned with the domains of the PES-NWI. The Pathway to Excellence® has less of a focus on nursedriven research, which makes it a more approachable and affordable choice for many hospitals looking to benefit from a better practice environment (ANCC, 2014).

Despite an increased focus on improving post-acute care in the community where overhead is considerably lower, the hospital will continue to be a safety-net of care for many invasive surgeries, acute health needs, and the nation's most vulnerable people. While the causes of readmission are multifactorial, these study findings demonstrate that hospital nursing care has important implications for patient outcomes. Administrators should consider investing in evidence-based value strategies that simplify care delivery and enhance the effectiveness of existing providers.



In light of these study findings, policymakers should continue to reward providers for high value care. Early evidence suggests that hospital pay-for-performance is not associated with improvements in care (Ryan et al, 2015), yet some contend that this may be due too small financial incentives and the overly intricate design of the incentive system (Jha, 2013; Werner & Dudley, 2012; Werner et al, 2011). Whether or not the payfor-performance incentives have been successful in improving care delivery and patient outcomes, they represent initial steps toward promoting high value care. As hospitals reimagine how to deliver high value care effectively, they should consider how to strategically use their most numerous providers – nurses.

Patients are also becoming increasingly aware of hospital quality as a result of public reporting of hospital performance measures on the Hospital Compare website and public knowledge about hospital Magnet® accreditation. Access to information about hospital quality on various performance measures, including readmission rates, gives patients tools to make more informed choices about where to seek healthcare. The public also has a responsibility to engage in political decisions including nurse staffing laws, which could make hospitals safer environments for patients and providers alike.

Future research

Future research in the areas of hospital nursing and readmissions is warranted. To date, research in this area has been cross-sectional. The study of causal relationships could be supported through longitudinal studies that account for patterns and changes over time. Such an analysis would provide stronger evidence of the relationship between hospital nursing and readmissions.



Future research should explore the processes through which better staffing and practice environments are linked to fewer readmissions. A recent study showed that missed nursing care – that is, care left undone due to a lack of time – may be one mechanism by which hospital nursing is linked to readmissions (Brooks Carton et al, 2015). This study conceptualized nurse staffing as a measure of patient surveillance. Additional research is needed to understand if and to what extent nursing surveillance, or other important tasks central to nursing – such as patient education and care coordination – , explain nursing's relationship to readmissions.

Comparative effectiveness studies contrasting various current practice models – such as telephone follow-up calls, care coordinators, or transitional care programs –, with improvements in nurse staffing and/or the practice environment, may yield additional insights into the efficacy and generalizability of various readmission reduction strategies. A comparative effectiveness approach may provide a reasonable method to compare costs associated with each intervention. Evidence about the comparative costs and benefits of investments in hospital nursing compared with current disease-specific approaches may add to the growing value-case for investments in hospital nursing.

Additional lines of research on readmissions should explore strategies for reducing readmissions among the most vulnerable patients at greatest risk of readmission. For example, this study and others have found that racial minorities are at higher risk for readmission compared with their white counterparts (Hu, Gonsahn, & Nerenz, 2014; Jencks, Williams, & Coleman, 2009; Joynt, Orav, & Jha, 2011; Zmistowski et al, 2013). Interestingly, in this study and others, racial minorities tend to be in better staffed hospitals compared with white patients (Brooks Carthon et al, 2012). Future research



should explore whether there is a moderating effect of race and staffing on readmission outcomes. One hypothesis is that better nurse staffing may have a more pronounced effect on readmission outcomes for minority patients. Such an analysis would provide evidence for more targeted readmission reduction strategies. Given the financial pressures faced by many hospitals, system level investments targeted at the most vulnerable populations may be one initial strategy to reduce health disparities and improve healthcare quality overall.

Conclusion

The findings of this study demonstrate an association between hospital nursing and readmissions in an older adult population undergoing elective THA/TKA. These findings have important implications that extend from patients, to providers, payers, and policymakers. Due to pay-for-performance initiatives in the Affordable Care Act, hospitals are pressured from payers and policymakers to improve health outcomes and processes of care. As hospital care quality becomes more transparent to the public, patients may also become more conscientious of where they seek healthcare, which will have important implications for hospitals. Indeed, the priority of healthcare should be providing safe care to patients. Despite increased attention to care quality, hospitals remain a dangerous environment for many older adults. Hospital readmissions are largely avoidable. Investments in hospital nursing may be one strategy to reduce readmissions. Failure to reduce readmissions remains a signal for low quality, high cost healthcare and jeopardizes the health, safety, and quality of life of patients.



APPENDIX A

	in merusion enterna for the study of elective fifth fifth feadimissions
	 Enrolled in Fee-For-Service Medicare
	THA: 44.043
	TKA: 06.536
	57
	2. Age 65 or older
	THA: 43,910
	TKA: 96,181
	\sim
	Discharged from a non-federal acute care hospital alive
	THA: 43,624
	TKA: 95.681
	<u>_</u>
	\sim
4. Enro	lled in Part A and Part B Medicare for the 12 month prior to the date of index admission
	THA: 42,920
	TKA: 95 021
	$\overline{\nabla}$
	×
5. Having	g a qualifying elective primary THA/TKA procedure; defined as those procedures without
	any of the following:
аF	emur, hip, or netwic fracture coded as principal or secondary discharge diagnosis fields of
4100	index admission
me	mdex admission
	1HA: 39,995
	TKA: 94,813
b. P	artial hip arthroplasty (PHA) procedures with concurrent THA/TKA
	THA: 39,995
	TKA: 94.810
	c. Revision procedures with a concurrent THA/TKA
	TUA 20 0005
	TEA. 04.650
	1KA: 94,039
d. F	tesurfacing procedures with a concurrent THA/TKA
	THA: 39,888
	TKA: 94,659
e. N	Achanical complication coded in the principal discharge diagnosis
	THA: 39 470
	TKA-04 594
	115.7. 74,204
£ 3.	folignant peoplesm of the policie geogram account lower limbs or here there were not
f N	falignant neoplasm of the pelvis, sacrum, coccyx, lower limbs, or bone/bone marrow or a
f. N diss	falignant neoplasm of the pelvis, sacrum, coccyx, lower limbs, or bone/bone marrow or a seminated neoplasm coded in the principal discharge diagnosis field
f. N diss	falignant neoplasm of the pelvis, sacrum, coccyx, lower limbs, or bone/bone marrow or a seminated neoplasm coded in the principal discharge diagnosis field THA: 39,446
f. N diss	falignant neoplasm of the pelvis, sacrum, coccyx, lower limbs, or bone/bone marrow or a seminated neoplasm coded in the principal discharge diagnosis field THA: 39,446 TKA: 94,570
f. M diss g. R	falignant neoplasm of the pelvis, sacrum, coccyx, lower limbs, or bone/bone marrow or a seminated neoplasm coded in the principal discharge diagnosis field THA: 39,446 TKA: 94,570 temoval of implanted devices/prostheses
f. N diss g. F	falignant neoplasm of the pelvis, sacrum, coccyx, lower limbs, or bone/bone marrow or a seminated neoplasm coded in the principal discharge diagnosis field THA: 39,446 TKA: 94,570 temoval of implanted devices/prostheses THA: 38.821
f. N diss g. F	falignant neoplasm of the pelvis, sacrum, coccyx, lower limbs, or bone/bone marrow or a seminated neoplasm coded in the principal discharge diagnosis field THA: 39,446 TKA: 94,570 temoval of implanted devices/prostheses THA: 38,821 TKA: 94,107
f. M diss g. F	falignant neoplasm of the pelvis, sacrum, coccyx, lower limbs, or bone/bone marrow or a seminated neoplasm coded in the principal discharge diagnosis field THA: 39,446 TKA: 94,570 temoval of implanted devices/prostheses THA: 38,821 TKA: 94,197
f. M diss g. F h. T	falignant neoplasm of the pelvis, sacrum, coccyx, lower limbs, or bone/bone marrow or a seminated neoplasm coded in the principal discharge diagnosis field THA: 39,446 TKA: 94,570 temoval of implanted devices/prostheses THA: 38,821 TKA: 94,197 fransfer from another acute care facility for THA/TKA
f. M diss g. F h. T	falignant neoplasm of the pelvis, sacrum, coccyx, lower limbs, or bone/bone marrow or a seminated neoplasm coded in the principal discharge diagnosis field THA: 39,446 TKA: 94,570 Removal of implanted devices/prostheses THA: 38,821 TKA: 94,197 Fransfer from another acute care facility for THA/TKA THA: 38,566

Figure A.1. Inclusion criteria for the study of elective THA/TKA readmissions





Figure A.2. Exclusion criteria for the study of elective THA/TKA readmissions



APPENDIX B

Methods: Sensitivity analysis

Patient characteristics

Discharge destination

Data on where the patient was discharged following admission was available from the MedPAR RIF. Discharge destination was descriptively and analytically modeled as a dichotomous variable. Discharge destination was specified as "discharged to home" or "not discharged to home". "Discharged to home" included being discharged to a private residence either with or without home healthcare services. "Not discharged to home" included, for example, a skilled nursing facility or inpatient rehabilitation facility.

Length of stay

Length of stay was calculated from the date of admission and date of discharge, which was provided in the MedPAR RIF. Length of stay was descriptively and analytically modeled as a continuous variable. Length of stay describes the number of days the patient was hospitalized on the initial admission.

Race/ethnicity

Data on patient race/ethnicity was available from the MedPAR RIF. The vast majority of patients in the sample were identified as white non-Hispanic. For analytic purposes, race/ethnicity was dichotomously coded as "white non-Hispanic" versus "other", with "other" as the reference category.



Socioeconomic status

Patient socioeconomic status was defined using the neighborhood socioeconomic summary index score for each ZIP code, available from the United States Census data. Socioeconomic status was categorized into tertiles: low (bottom 25%), average (26%-74%), and high (top 25%). For analytic purposes, each of the three categories was created as indicator variables, with average and high socioeconomic status as the reference category, to ease interpretation.

Hospital characteristics

Geographic location

The AHA Annual Survey includes a measure of population density in the surrounding area where each hospital is located. This Census-derived measure is known as the core-based statistical area and was categorized as follows: division (>2.5 million); metropolitan (50,000 - 2.5 million); micropolitan (10,000 - <50,000); rural (<10,000) (Brooks Carthon et al, 2012). To ensure consistency with other studies and to simplify the analysis, population density of the hospital's location was restricted to urban (\geq 50,000) and rural (<50,000). Rural was the reference category in the regression analysis.

Caseload volume

The caseload volume was derived from the MedPAR RIF as the number of live discharges per hospital of patients who met the inclusion and exclusion criteria for this study. Each study hospital's caseload volume was used to create tertiles such that the lowest volume tertile represented hospitals in the lowest third for caseload volume. A high volume hospital was represented as a hospital in the highest third of the study



hospitals for caseload volume. The volume tertiles were created as dummy variables and the high volume variable was the reference category in the regression analyses.

Ownership type

Hospital ownership type was available from the AHA Annual Survey. Hospitals were categorized as being for-profit, not-for-profit, or government-non-federal. Government-non-federal was the reference category in the analysis, so that the effect of each ownership type on readmission could be assessed.

Socioeconomic status profile

The hospital socioeconomic status profile was defined by the proportion of low socioeconomic status patients within the hospital. Hospitals serving higher proportions of low socioeconomic status patients may deliver a larger share of uncompensated care, or be less well reimbursed by Medicaid, than commercial third party payers. This lack of reimbursement through uncompensated care, and minimal reimbursement through Medicaid, reflects the hospital's ability to invest in structural and organizational features that may influence patient outcomes.

After identifying the individual socioeconomic score of patients by linking their home addresses with the United States Census data, a rough measure of patient socioeconomic status was identified. The top 10% of hospitals in the study sample with the highest proportion of low socioeconomic status patients were classified as a hospital serving a disproportionate share of low socioeconomic status patients (Joynt, Orav, & Jha, 2011). Hospitals defined as serving a disproportionate share of low socioeconomic



status patients was an indicator variable in the regression analysis, with nondisproportionate share hospitals as the reference category.

Magnet® designation

Hospital Magnet® designation status data was available from the ANCC website. An indicator variable was created to identify whether a hospital was designated as a Magnet® hospital in the years 2005, 2006, or 2007.

Surgical care improvement project process measures

Hospital performance on 2006 surgical care improvement project process measures, were analytically modeled as a continuous variable of the percentage of patients within the hospital for whom the hospital met each of the three quality measures. Additionally, an average of the hospital's performance on all three measures was used to create a total score. The total score on surgical care improvement project quality performance was also modeled in the sensitivity analysis.



APPENDIX C

Results: Sensitivity analysis

Patient characteristics

The findings reflect: (1) the unadjusted effect of the additional patient level covariate on readmission, and (2) separate models adjusting for patient and hospital covariates, including the additional patient level covariate being tested in the sensitivity analysis. The patient level covariates, as shown in Table A.1, included: discharge destination (home with/without homecare vs. other), length of stay on the index admission, race/ethnicity (white non-Hispanic vs. other), and socioeconomic status (high socioeconomic status vs. other).

The unadjusted effects are consistent with prior findings that white non-Hispanic race/ethnicity and discharge to home are associated with lower odds of readmission and longer lengths of stay are associated with greater odds of readmission. Patient socioeconomic status was not significantly associated with readmission in the unadjusted analysis (OR 1.08, 95% CI 0.95-1.23).

The association of staffing and 30 day readmission proved to be robust, after controlling for patient level covariates in the sensitivity analysis. Indeed, even after separately controlling for discharge destination, length of stay, race/ethnicity, and socioeconomic status, the effects of staffing remain significant. Where patients rehabilitate following discharge appears to be somewhat predictive of the patients' odds of readmission. Except when controlling for discharge destination, the effects of practice environment remain significant.



Odds of 30		Length of stay		
day	Discharge	on index	Race/	Socioeconomic
Readmission	destination	admission	Ethnicity	status
	OR (95% CI)	OR (95% CI)	OR (95% CI)	OR (95% CI)
Unadjusted	0.46***	1.10***	0.86*	1.08
-	(0.42-0.50)	(1.09-1.11)	(0.77-0.97)	(0.95-1.23)
Staffing	1.06*	1.09**	1.09**	1.09**
-	(1.00-1.12)	(1.03-1.16)	(1.02-1.16)	(1.03-1.16)
Practice	0.89	0.87*	0.87*	0.84*
environment	(0.80-1.01)	(0.76-0.99)	(0.76-0.99)	(0.74-0.96)

Table A.1. Patient level sensitivity analysis of the effects of hospital nursing on 30 day readmission (N = 112,018)

p <0.05*; p<0.01**; p<0.001***

Patient characteristics include: age, sex, comorbidities, type of surgery, number of procedures. Hospital characteristics include: bed size, teaching status, technology status. All of the models account for clustering of patients within hospitals. This analysis contains 112,018 patients in 495 hospitals.



Hospital characteristics

Sensitivity analyses were also conducted for hospital level covariates, including: geographic location (urban vs. rural), patient caseload volume, ownership type (profit, not-for-profit, vs. government nonfederal), socioeconomic status profile (hospitals in the highest decile of the proportion of patients with low socioeconomic status), Magnet® designation, and surgical care improvement project performance measures. Controlling for each of these hospital level covariates does not have a clinically meaningful impact on the associations between staffing and the practice environment on 30 day readmission (Table A.2). In fact, the association between practice environment and readmission actually became more pronounced after controlling for specific hospital covariates, such as ownership type (OR 0.84; 95% CI 0.73-0.95).

In the unadjusted models, hospital caseload volume, socioeconomic status profile, and Magnet® designation were not significantly associated with readmission. Urban location was associated with greater odds of readmission (OR 1.25; 95% CI 1.06-1.49). For-profit hospitals had lower odds of readmission (OR 0.86; 95% CI 0.76-0.98) and notfor-profit had greater odds of readmission (OR 1.17; 95% CI 1.04-1.32).

Table A.3 shows the hospital nursing effects adjusted for hospital level surgical care improvement project performance measures, in addition to other patient and hospital covariates. The following surgical care improvement project measures from 2006 were examined separately and then combined into an overall average score: percentage of surgery patients who/whose (1) received preventative antibiotic(s) one hour before incision, (2) received the appropriate preventative antibiotic(s) for their surgery, and (3) preventative antibiotic(s) are stopped within 24 hours after surgery. The surgical care



improvement project measures were not associated with readmission in the unadjusted models. The staffing and practice environment effects remained significant even after adjusting for each of the surgical care improvement project measures.



Odds of 30 day Readmission	Geographic location	Patient caseload volume	Ownership type	Socio- economic status profile	Magnet® designation
	OR	OR	OR	OR	OR
	(95% CI)	(95% CI)	(95% CI)	(95% CI)	(95% CI)
Unadjusted	1.25** (1.06-1.49)	1.16 (0.63-2.13)	Profit: 0.86* (0.76-0.98) Not-For- Profit: 1.17* (1.04-1.32)	1.10 (0.86-1.42)	0.98 (0.13-7.33)
Staffing	1.09** (1.03-1.17)	1.08** (1.02-1.15)	1.08** (1.02-1.15)	1.08* (1.02-1.15)	1.08* (1.02-1.15)
Practice	0.87*	0.86*	0.84**	0.86*	0.87*
environment	(0.76-0.99)	(0.75-0.98)	(0.73-0.95)	(0.76-0.98)	(0.76-0.99)

Table A.2. Hospital level sensitivity analysis of the effects of hospital nursing on 30 day readmission (N = 112,018)

p <0.05*; p<0.01**; p<0.001***

Patient characteristics include: age, sex, comorbidities, type of surgery, number of procedures. Hospital characteristics include: bed size, teaching status, technology status. All of the models account for clustering of patients within hospitals. This analysis contains 112,018 patients in 495 hospitals.


Odds of 30 day Readmission	Antibiotic(s) 1 hour before incision	Appropriate antibiotic(s) for surgery	Antibiotic(s) stopped within 24 hours after surgery	Average score
	OR (95% CI)	OR (95% CI)	OR (95% CI)	OR (95% CI)
Unadjusted	1.00	1.01	1.00	1.01 (1.00-
	(1.00-1.01)	(1.00-1.01)	(1.00-1.01)	1.01)
	1.08**	1.08*	1.07*	1.08* (1.01-
Staffing	(1.02-1.15)	(1.02-1.15)	(1.01-1.14)	1.14)
Practice	0.87*	0.85*	0.86*	0.86* (0.76-
environment	(0.76-0.99)	(0.75-0.98)	(0.76-0.99)	0.99)

Table A.3. Hospital surgical care improvement project measure sensitivity analysis of the effects of hospital nursing on 30 day readmission (N = 112,018)

p <0.05*; p<0.01**; p<0.001***

Patient characteristics include: age, sex, comorbidities, type of surgery, number of procedures. Hospital characteristics include: bed size, teaching status, technology status. All of the models account for clustering of patients within hospitals. This analysis contains 112,018 patients in 495 hospitals.



APPENDIX D

Discussion: Sensitivity analysis

The sensitivity analyses revealed that including additional patient and hospital characteristics into the regression models did not substantively detract from the association between hospital nursing and readmissions. Controlling for discharge destination had the most pronounced effect on the association of both staffing and the practice environment. The effect of staffing was reduced by two percentage points (OR 1.06, 95% CI 1.00-1.12) and the practice environment was rendered insignificant (OR 0.89, 95% CI 0.80-1.101). These findings are not entirely unexpected in consideration of prior evidence (Bini et al, 2009; Schairer et al, 2014; Tsai et al, 2013; Zmistowski et al 2013). Given such robust evidence that discharge to a facility significantly increases the odds of readmission above and beyond patient acuity, it begs the question whether discharge destination should be risk adjusted for in readmission penalties. Currently, CMS does not risk adjust for discharge destination because it is believed to be associated with patient comorbidities, which is currently adjusted for in the readmission measure (Suter et al, 2014). Discharge destination is also related to the structure of the healthcare system, such as the availability of providers, which is generally beyond the scope of hospital control (Suter et al, 2014). These considerations suggest a valid argument for not risk adjusting for discharge destination in the readmission penalties.

Controlling for patient length of stay on the index admission, race/ethnicity, and socioeconomic status resulted in a greater effect of staffing on readmission. For example, once race/ethnicity was controlled for, the effect of staffing increased one percentage point. A likely explanation for this could be that non-white patients are in better staffed



hospitals yet have worse readmission outcomes. Additional analyses (Table A.4) revealed that non-white patients were, in fact, cared for in hospitals with better staffing than white patients (mean staffing: 4.94 vs. 4.69, p <0.001). Similar results were found for length of stay and socioeconomic status (Table A.5).

Although patient socioeconomic status was not significant in the unadjusted model, the effect of being in a good practice environment became even more pronounced as compared to when socioeconomic status was not included in the model (OR 0.84, 95% CI 0.74 - 0.96). Patients with low socioeconomic status were more likely to be in hospitals rated as having poor practice environments (Table A.6).

Nursing Characteristics	White : Hispanic N = 102	non- patients 2,887	Non-White patients N = 9,130		
	Ν	%	Ν	%	
Staffing					
<4	13,036	12.67	1,618	17.72	
4-<5	43,500	42.28	4,451	48.75	
5-<6	35,534	34.54	2,522	27.62	
6-<7	8,185	7.96	401	4.39	
7 +	2,632	2.56	138	1.51	

Table A.4.	Nursing	characteristics	by	patient race
			~ /	

Table A.5. Patient characteristics by patient race

Patient Characteristics	White non- Hispanic patients N = 102,887		Non-White patients N = 9,130	
	Ν	%	Ν	%
Length of stay (days) (mean, SD)	3.74	1.94	4.15	2.39
Socioeconomic status				
Low	23,136	23.43	4,061	46.50
Medium	50,542	51.19	3,184	36.46
High	25,061	25.38	1,489	17.05



Nursing Characteristics	Low socioeconomic status N = 27,197		Medium socioeconomic status N = 53,726		High socioeconomic status N = 26,550	
	Ν	%	Ν	%	Ν	%
Practice environment						
Poor	9,166	34.07	13,678	25.46	3,391	12.77
Mixed	13,352	49.09	27,467	51.12	13,529	50.96
Good	4,579	16.84	12,581	23.42	9,630	36.27

Table A.6. Nursing characteristics by patient socioeconomic status

Recently, there has been increased debate about whether socioeconomic status should be risk adjusted for in the readmission penalties. The complexity of this debate centers around the desire for performance measures to be based on fair comparisons across providers while not masking or perpetuating disparities that exist (NQF, 2014). Risk adjusting for socioeconomic status can mask disparities in health outcomes of vulnerable populations by effectually "adjusting away" the observed disparities (NQF, 2014; Suter et al, 2014). Doing so holds hospitals to different standards based on the socioeconomic status of their patient populations (Suter et al, 2014). This might lower the incentives for hospitals with a lower socioeconomic status population to improve readmission rates – effectively allowing those hospitals to perform worse on quality because that is the expectation (NQF, 2014).

Alternatively, there are valid concerns about the impact of not risk adjusting for socioeconomic status. Hospitals serving the neediest populations are penalized disproportionately, which makes improving care even more difficult in already under-resourced hospitals (NQF, 2014; Rau, 2013). This results in adverse feedback loops that could worsen, rather than improve, health disparities. Early evidence shows that safety-net hospitals tend to have smaller improvements in performance measures over time,



compared with non-safety-net hospitals (Werner, Goldman, & Dudley, 2008).

Additionally, with reimbursement at risk, providers have an incentive to "cherry pick" patients who are more likely to perform better on outcome measures (NQF, 2014; Werner, Asch, & Polsky, 2005). This phenomenon was observed when public reporting (unadjusted for race/ethnicity) for mortality rates following coronary artery bypass grafts (CABG) was introduced – fewer CABGs were performed on racial and ethnic minorities (Werner, Asch, & Polsky, 2005).

Evidence to date demonstrates that hospitals serving a larger portion of economically disadvantaged patients are more severely affected by readmission penalties. In the first year of implementing CMS penalties for hospitals with worse than expected readmission rates, 77% of safety-net hospitals were penalized, whereas only 36% of hospitals, which cared for the fewest low-income individuals, were penalized (Rau, 2013). Adjusting for socioeconomic status would likely equalize the expected readmission rates among hospitals that are safety-net and non-safety-net, alleviating the readmission penalty disparities between otherwise similar hospitals. Yet the debate still lingers as to whether the CMS penalties should account for hospitals' case-mix of patient socioeconomic status (Hu, Gonsahn, & Nerenz, 2014; Nagasako et al, 2014).

Hospital characteristics were also included in the sensitivity analysis as covariates that might confound the hospital nursing and readmission relationship. Controlling for geographic location of the hospital, patient caseload volume, ownership type, socioeconomic status profile, and whether or not the hospital was Magnet® designated, had no meaningful impact on the effect of staffing or the practice environment on readmission. Also, adjusting for hospital performance on surgical care improvement



project performance measures did not explain the relationship between hospital nursing and readmission. In sum, the relationship between hospital nursing and readmission was found to be robust, even after adjusting for potentially confounding patient and hospital characteristics.



REFERENCES

- Achté, K.A., & Apo, M. (1967). Schizophrenic patients in 1950-1952 and 1957-1959. *Psychiatric Quarterly*, 41(3), 422-441.
- AHA. (2014). AHA Annual Survey Database. Retrieved August 20, 2014 from http://www.ahadataviewer.com/book-cd-products/aha-survey/
- Ahmed, A. (2002). Quality and outcomes of heart failure care in older adults: Role of multidisciplinary disease-management programs. *Journal of American Geriatric Society*, 50(9), 1590-1593.
- Aiken, L.H., Smith, H.L., & Lake, E.T. (1994). Lower Medicare mortality among a set of hospitals known for good nursing care. *Medical Care*, *32*(8), 771-787.
- Aiken, L.H., Clarke, S.P., & Sloane, D.M. (2002). Hospital staffing, organization, and quality of care: Cross-national findings. [Research Support, Non-U.S. Gov't Research Support, U.S. Gov't, P.H.S.]. *Nursing outlook*, 50(5), 187-194.
- Aiken, L.H., Clarke, S.P., Sloane, D.M., Sochalski, J. & Silber, J.H. (2002). Hospital nurse staffing and patient mortality, nurse burnout, and job dissatisfaction. *The Journal of the American Medical Association*, 288(16), 1987-1993.
- Aiken, L.H., Clarke, S.P., Cheung, R.B., Sloane, D.M., & Silber, J.H. (2003). Educational levels of hospital nurses and surgical patient mortality. *The Journal* of the American Medical Association, 290(12), 1617-1623.
- Aiken, L.H. (2008). Economics of nursing. *Policy, Politics, and Nursing Practice*, 9(2), 73-79.
- Aiken, L.H., Sloane, D.M., Cimiotti, J.P., Clarke, S.P., Flynn, L., Seago, J.A., Spetz, J., & Smith, H.L. (2010). Implications of the California nurse staffing mandate for other states. *Health Services Research*, 45(A), 904-921.
- Aiken, L.H., Cimiotti, J.P., Sloane, D.M., Smith, H.L., Flynn, L., & Neff, D.F. (2011). The effects of nurse staffing and nurse education on patient deaths in hospitals with different nurse work environments. *Medical Care*, 49(12), 1047-1053.
- American Academy of Orthopedic Surgeons (AAOS). Hip and Thigh. Retrieved September 5, 2014 from <u>http://orthoinfo.aaos.org/</u>
- American Nurses Association (ANA). What is Nursing? Retrieved March 5, 2015 from <u>http://nursingworld.org/EspeciallyForYou/What-is-Nursing</u>



- American Nurses Credentialing Center (ANCC). (2014). ANCC Magnet Recognition Program. Retrieved October 1, 2014 from <u>http://www.nursecredentialing.org/magnet.aspx</u>
- Anderson, G.F., & Steinberg, E.P. (1984). Hospital readmissions in the Medicare population. *The New England Journal of Medicine*, *311*(21), 1349-1353.
- Ashton, C.M., Del Junco, D.J., Souchek, J., Wray, N.P., & Mansyur, C.L. (1997). The association between the quality of inpatient care and early readmission: a meta-analysis of the evidence. *Medical Care*, *35*(10), 1044-1059.
- Ayanian, J., & Weissman, J. (2002). Teaching hospitals and quality of care: A review of the literature. *Milbank Quarterly*, 80(3), 569–593.
- Bini, S.A., Fithian, D.C., Paxton, L.W., Khatod, M.X., Inacio, M.C., & Namba, R.S. (2009). Does discharge disposition after primary total joint arthroplasty affect readmission rates? *Journal of Arthroplasty*, 25(1), 114-117.
- Birkmeyer, N., Gu, N., Baser, O., Morris, A., & Birkmeyer, J. (2008). Socioeconomic status and surgical mortality in the elderly. *Medical Care*, 46(9), 893-899.
- Blegen, M.A., Goode, C.J., Park, S.H., Vaughn, T., & Spetz, J. (2013). Baccalaureate education in nursing and patient outcomes. *The Journal of Nursing Administration*, 43(2), 89-94.
- Blegen, M.A., Goode, C.J., Spetz, J., Vaughn, T., & Park, S.H. (2011). Nurse staffing effects on patient outcomes: Safety-net and non-safety-net hospitals. *Medical Care*, 49(4), 406-414.
- Boltz, M., Capezuti, E., Shabbat, N., & Hall, K. (2010). Going home better not worse: Older adults' views on physical function during hospitalization. *International Journal of Nursing Practice*, 16, 381-388.
- Boltz, M., Capezuti, E., & Shabbat, N. (2011). Nursing staff perceptions of physical function in hospitalized older adults. *Applied Nursing Research*, 24(4), 215-222.
- Boltz, M., Resnick, B., Capezuti, E., Shuluk, J., & Secic, M. (2012). Functional decline in hospitalized older adults: Can nursing make a difference? *Geriatric Nursing*, *33*(4), 272-279.
- Bozic, K.J., Rubash, H.E., Sculco, T.P., & Berry, D.J. (2008). An analysis of Medicare payment policy for total joint arthroplasty. *Journal of Arthroplasty*, 23(6 Suppl 1), 133-138.



- Brooks Carthon, J.M., Kutney-Lee, A., Jarrin, O., Sloane, D., & Aiken, L.H. (2012). Nurse staffing and postsurgical outcomes in black adults. *Journal of the American Geriatrics Society*, 60(6), 1078-1084.
- Brooks Carthon, J.M., Lasater, K.B., Sloane, D.M., & Kutney-Lee, A. (2015). The quality of hospital work environments and missed nursing care is linked to heart failure readmissions: A cross-sectional study of US hospitals. *British Medical Journal, Quality and Safety, 24*(4), 255-263.
- Burton, R. (2012). Health Policy Brief: Care Transitions. *Health Affairs*. Retrieved September 8, 2014 from rwjf.org/content/dam/farm/reports/issue briefs/2012/rwjf401314.
- CDC. (2014). Inpatient surgery. Retrieved September 10, 2014 from <u>http://www.cdc.gov/nchs/fastats/inpatient-surgery.html.</u>
- CMS. (2013). Medicare Provider Analysis and Review (MEDPAR) File. Retrieved August 28, 2014 from <u>http://www.cms.gov/Research-Statistics-Data-and-Systems/Files-for-Order/IdentifiableDataFiles/MedicareProviderAnalysisandReviewFile.html</u>.
- CMS. (2014). Readmissions Reduction Program. Retrieved September 9, 2014 from <u>http://www.cms.gov/Medicare/Medicare-Fee-for-Service-</u> Payment/AcuteInpatientPPS/Readmissions-Reduction-Program.html.
- Coleman, E.A., Parry, C., Chalmers, S., & Min, S.J. (2006). The care transitions intervention: results of a randomized controlled trial. *Archives of Internal Medicine*, *166*(17), 1822-1828.
- Congressional Budget Office. (2014, April 14). Medicare Baseline by Fiscal Year. Retrieved September 10, 2014 from <u>http://www.cbo.gov/sites/default/files/cbofiles/attachments/44205-2014-04-Medicare.pdf</u>
- Covinsky, K.E., Palmer, R.M., Fortinsky, R.H. et al. (2003). Loss of independence in activities of daily living in older adults hospitalized with medical illness: increased vulnerability with age. *Journal of the American Geriatrics Society*, *51*, 451-458.
- Cram, P., Lu, X., Kaboli, P.J., Vaughan-Sarrazin, M.S., Cai, X., Wolf, BR., & Li, Y. (2011). Clinical characteristics and outcomes of Medicare patients undergoing total hip arthroplasty, 1991-2008. *The Journal of the American Medical Association*, 305(15), 1560-1567.
- Cram, P., Lu, X., Kates, S.L., Singh, J.A., Li ,Y., & Wolf, B.R. (2012). Total knee arthroplasty volume, utilization, and outcomes among Medicare beneficiaries,



1991-2010. *The Journal of the American Medical Association*, 308(12), 1227-1236.

- Cullen, C., Johnson, D.S., & Cook, G. (2006). Readmission rates within 28 days of total hip replacement. *Annals of the Royal College of Surgeons of England*. 88(5), 475-478.
- Dharmarajan, K., Hsieh, A.F., Lin, Z., Bueno, H., Ross, J.S., Horwitz, L.I., ... Krumholz, H.M. (2013). Diagnoses and timing of 30-day readmissions after hospitalization for heart failure, acute myocardial infarction, or pneumonia. *The Journal of the American Medical Association*, 309(4), 355-363.
- Diez Roux, A.V., Merkin, S.S., Arnett, D., Chambless, L., Massing, M., Nieto, F.J., ... Watson, R.L. (2001). Neighborhood of residence and incidence of coronary heart disease. *The New England Journal of Medicine*, 345(2), 99-106.
- Dillman, D.A. (1978). Mail & telephone surveys: The total design method. New York: Wiley.
- Dillman, D.A. (2000). Mail and internet surveys the tailored design method. New York: John Wiley & Sons, Inc.
- Diya, L., Van den Heede, K., Sermeus, W., & Lesaffre, E. (2011). The relationship between in-hospital mortality, readmission into the intensive care nursing unit and/or operating theatre and nurse staffing levels. *Journal of Advanced Nursing*, 68(5), 1073-1081.
- Donabedian, A. (1966). Evaluating the quality of medical care. *The Milbank Quarterly*, 83(4), 691-729.
- Dudas, V., Bookwalter, T., Kerr, K.M., & Pantilat, S.V. (2001). The impact of follow-up telephone calls to patients after hospitalization. *The American Journal of Medicine*, 111(9B), 26S–30S.
- Estabrooks, C. A., Midodzi, W. K., Cummings, G. G., Ricker, K. L., & Giovannetti, P. (2005). The impact of hospital nursing characteristics on 30-day mortality. [Research Support, Non-U.S. Gov't Research Support, U.S. Gov't, P.H.S.]. Nursing Research, 54(2), 74-84.
- Fitzmaurice, G.M., Laird, N.M., & Ware, J.H. (2011). Applied longitudinal analysis (2nd Ed.). Hoboken, NJ: John Wiley & Sons Inc.
- Friese, C.R., Lake, E.T., Aiken, L.H., Silber, J.H., & Sochalski, J. (2008). Hospital nurse practice environments and outcomes for surgical oncology patients. *Health Services Research*, *43*(4), 1145-1163.



- Garcia, N., Fogel, S., Baker, C., Remine, S., & Jones, J. (2012). Should compliance with the Surgical Care Improvement Project (SCIP) process measures determine Medicare and Medicaid reimbursement rates? *The American Journal of Surgery*, 78(6), 653-656.
- Goodney, P.P., Stukel, T.A., Lucas, F.L., Finlayson, E.V., & Birkmeyer, J.D. (2003). Hospital volume, length of stay, and readmission rates in high-risk surgery. *The American Journal of Surgery*, 238(2), 161-167.
- Gorodeski, E.Z., Starling, R.C., & Blackstone, E.H. (2010). Are all readmissions bad readmissions? *The New England Journal of Medicine*, *363*(3), 297-298.
- Green, L.W. (1970). Manual for scoring socioeconomic status for research on health behavior. *Public Health Reports*, 85(9), 815-827.
- Hansen, L.O., Young, R.S., Hinami, K., Leung, A., & Williams, M.V. (2011). Interventions to reduce 30-day rehospitalization: a systematic review. *Annals of Internal Medicine*, 155(8), 520-528.
- HCUP. (2012). Clinical Classifications Software (CCS) for ICD-9-CM Fact Sheet. Retrieved August 28, 2014 from <u>http://www.hcup-us.ahrq.gov/toolssoftware/ccs/ccsfactsheet.jsp</u>.
- Heggestad, T. (2002). Do hospital length of stay and staffing ratio affect elderly patients' risk of readmission? A nation-wide study of Norwegian hospitals. *Health Services Research*, *37*(3), 647-665.
- Hosmer, D.W., & Lemeshow, S. (2000). *Applied Logistic Regression (2nd ed)*. New York, NY: John Wiley & Sons.
- Hu, J., Gonsahn, M.D., & Nerenz, D.R. (2014). Socioeconomic status and readmissions: evidence from an urban teaching hospital, *Health Affairs*, *33*(5), 778-785.
- Iezzoni, (2013). *Risk adjustment for measuring health care outcomes (4th ed)*. Chicago, IL: Health Administration Press.
- Jack, B., Chetty, V., Anthony, D., Greenwald, J., Sanchez, G., Johnson, A. ... Culpepper, L. (2009). A reengineered hospital discharge program to decrease rehospitalization: a randomized trial. *Annals of Internal Medicine*, 150(3), 178-187.
- Jain, N.B., Higgins, L.D., Ozumba, D., Guller, U., Cronin, M., Pietrobon, R., & Katz, J. (2005). Trends in epidemiology of knee arthroplasty in the United States, 1990– 2000. Arthritis & Rheumatism, 52(12), 3928–3933.



- James, J. (2012). Health Policy Brief: Pay-for-Performance. *Health Affairs*. Retrieved September 8, 2014 from http://www.healthaffairs.org/healthpolicybriefs/brief.php?brief_id=78
- Jencks, S.F., Williams, M.V., & Coleman, E.A. (2009). Rehospitalizations among patients in the Medicare fee-for-service program. *The New England Journal of Medicine*, 360(14), 1418-1428.
- Jenkins, R.L., Bermiss, E.L., & Lorr, M. (1953). Duration of hospitalization, readmission rate and stability of diagnoses in Veterans hospitalized with neuropsychiatric diagnoses. *The Psychiatric Quarterly*, 27(1-4), 59-72.
- Jha, A.K. (2013). Time to get serious about pay for performance. *The Journal of the American Medical Association, 309*(4) 347-348.
- Joynt, K.E. & Jha, A.K. (2011). Who has higher readmission rates for heart failure, and why? *Circulation: Cardiovascular Quality and Outcomes*, 4(1), 53-59.
- Joynt, K.E. & Jha, A.K. (2012). Thirty-day readmissions truth and consequences. *The New England Journal of Medicine, 366*(15), 1366-1369.
- Joynt, K.E. & Jha, A.K. (2013). Characteristics of hospitals receiving penalties under the hospital readmissions reduction program. *Journal of the American Medical Association*, 309(4), 342-343.
- Joynt, K.E., Orav, J.E., & Jha, A.K. (2011). Thirty-day readmission rates for Medicare beneficiaries by race and site of care. *Journal of the American Medical Association*, 305(7), 675-681.
- Joynt, K.E., Sarma, N., Epstein, A.M., Jha, A.K., & Weissman, J.S. (2014). Challenges in reducing readmissions: Lessons from leadership and frontline personnel at eight minority-serving hospitals. *Joint Commission Journal on Quality and Patient Safety*, 40(10), 435-443.
- Judge, A., Chard, J., Learmonth, I., & Dieppe, P. (2006). The effects of surgical volumes and training centre status on outcomes following total joint replacement: analysis of the Hospital Episode Statistics for England. *Journal of Public Health*, 28(2), 116-124.
- Kaiser Health News. (2014). Readmission Penalties by State: Year Three. Retrieved December 5, 2014 from <u>http://kaiserhealthnews.org/news/medicare-readmissions-penalties-by-state/</u>
- Kalisch, B.J. (2006). Missed nursing care: a qualitative study. *Journal of Nursing Care Quality*, 21(4), 306-313.



- Kalisch, B.J., Gay, L., & Williams, R.A. (2009). Missed nursing care: errors of omission. *Nursing Outlook*, 57(1), 3-9.
- Kane, R.L., Shamliyan, T.A., Mueller, C., Duval, S. & Wilt, T. (2007). The association of registered nurse staffing levels and patient outcomes: systematic review and metaanalysis. *Medical Care*, 45(12), 1195-1204.
- Kane, N.M., & Siegrist, R.B. (2002). Understanding rising hospital inpatient costs: Key components of cost and the impact of poor quality. Chicago, IL. Blue Cross and Blue Shield Association.
- Katz, J.N., Barrett, J., Mahomed, N.N., Baron, J.A., Wright, R.J., & Losina, E. (2004). Association between hospital and surgeon procedure volume and the outcomes of total knee replacement. *Journal of Bone and Joint Surgery*, 86-A(9), 1909–1916.
- Kazanjian, A., Green, C., Wong, J., & Reid, R. (2005). Effect of the hospital nursing environment on patient mortality: a systematic review. *Journal of Health Services Research and Policy*, 10(2), 111-117.
- Kendall-Gallagher, D., Aiken, L.H., Sloane, D.M., & Cimiotti, J.P. (2011). Nurse specialty certification, inpatient mortality, and failure to rescue. *Journal of Nursing Scholarship* 43(2), 188-194.
- Khan, M.A., Hossain, F.S., Dashti, Z., & Muthukumar, N. (2012). Causes and predictors of early re-admission after surgery for a fracture of the hip. *Journal of Bone and Joint Surgery British Volume*, *94*(5), 690-697.
- Kocher, R.P. & Adashi, E.Y. (2011). Hospital readmissions and the Affordable Care Act: paying for coordinated quality care. *Journal of the American Medical Association*, 306(16), 1794-1795.
- Kocher, K.E., Nallamothu, B.K., Birkmeyer, J.D., & Dimick, J.B. (2013). Emergency department visits after surgery are common for Medicare patients, suggesting opportunities to improve care. *Health Affairs*, *32*(9), 1600-1607.
- Kortebein, P., Symons, T.B., Ferrando, A., Paddon-Jones, D., Ronsen, O., & Protas, E. (2008). Functional impact of 10 days of bed rest in healthy older adults. *Journal of Gerontology: Medical Sciences*, 63A(10), 1076–1081.
- Kortebein, P. (2009). Rehabilitation for hospital-associated deconditioning. *American Journal of Physical Medicine and Rehabilitation*, 88(1), 66-77.
- Krumholz, H.M., Merrill, A.R., Schone, E.M., Schreiner, G.C., Chen, J., Bradley, E.H., Drye, E.E. (2009). Patterns of hospital performance in acute myocardial infarction and heart failure 30-day mortality and readmission. *Circulation: Cardiovascular Quality and Outcomes, 2*, 407–413.



- Kurtz, S., Mowat, F., Ong, K., Chan, N., Lau, E., & Halpern, M. (2005). Prevalence of primary and revision total hip and knee arthroplasty in the United States (1990-2002). *Journal of Bone and Joint Surgery - American Volume*, 87(7), 1487–1497.
- Kutney-Lee, A., Sloane, D.M., & Aiken, L.H. (2013). An increase in the number of nurses with baccalaureate degrees is linked to lower rates of postsurgery mortality. *Health Affairs*, *32*(3), 579-586.
- Lake, E.T. (2002). Development of the practice environment scale of the nursing work index. *Research in Nursing and Health*, 25(3), 176-188.
- Ma, C., McHugh, M., & Aiken, L. (2015). Organization of hospital nursing and 30-day readmissions in Medicare patients undergoing surgery. *Medical Care*, 53(1) 65-70.
- Ma, C., Shang, J., & Stone, P.W. (2014). Can nurse work environment influence readmission risk? – A systematic review. *Nursing: Research and Reviews*, 4, 91-101.
- Mahomed, N.N., Barrett, J.A., Katz, J.N., Phillips, C.B., Losina, E., Lew, R.A., ... Baron, J.A. (2003). Rates and outcomes of primary and revision total hip replacement in the United States Medicare population. *Journal of Bone & Joint Surgery -American Volume*, 85-A(1), 27-32.
- Mäkelä, K.T., Hakkinen, U., Peltola, M., Linna, M., Kroger, H., & Remes V. (2011). The effect of hospital volume on length of stay, re-admissions, and complications of total hip arthroplasty. *Acta Orthopaedica*, 82(1), 20-26.
- Martsolf, G. R., Auerbach, D.R., Benevent, R., Stocks, C., Jiang, H.J., Pearson, M.L., Ehrlich, E.D., & Gibson, T.B. (2014). Examining the value of inpatient nurse staffing. *Medical Care*, *52*(11), 982-988.
- Merkow, R.P., Ju, M.H., Chung, J.W., Hall, B.L., Cohen, M.E., Williams, M.V., Tsai, T.C., Ko, C.Y., & Bilimoria, K.Y. (2015). Underlying reasons associated with hospital readmission following surgery in the United States. *Journal of the American Medical Association*, 313(5), 483-495.
- McClure, M.L., Poulin, M.A., Sovie, M.D., & Wandelt, M.A. (1983). *Magnet Hospitals: Attraction and Retention of Professional Nurses*. Kansas City, MO: American Academy of Nursing.
- McHugh, M.D., Berez, J., & Small, D.S. (2014). Hospitals with higher nurse staffing had lower odds of readmissions penalties than hospitals with lower staffing. *Health Affairs*, *33*(3), 1740-1747.



- McHugh, M., Kang, R., & Hasnain-Wynia, R. (2009). Understanding the safety-net: Inpatient quality of care varies based on how one defines safety-net hospitals. *Medical Care Research and Review*, 66(5), 590–605.
- McHugh, M.D, & Ma, C. (2013). Hospital nursing and 30-day readmissions among Medicare patients with heart failure, acute myocardial infarction, and pneumonia. *Medical Care*, 52(1), 52-59.
- McHugh, M.D., & Witkoski Stimpfel, A. (2012). Nurse reported quality of care: A measure of hospital quality. *Research in Nursing and Health*, *35*, 566-575.
- Mitchell, P., Ferketich, S., & Jennings, B. (1998). Quality health outcomes model. *Journal of Nursing Scholarship*, 30(1), 43-46.
- Mitchell, P.H., & Lang, N.M. (2004). Framing the problem of measuring and improving healthcare quality: has the Quality Health Outcomes Model been useful? *Medical Care*, 42(2), II4-II11.
- Mnatzaganian, G., Ryan, P., Norman, P., Davidson, D., & Hiller, J. (2012). Length of stay in hospital and all-cause readmission following elective total joint replacement in elderly men. *Orthopedic Research and Reviews*, *4*, 43-51.
- Mor, V., Intrator, O., Feng, Z., & Grabowski, D.C. (2010). The revolving door of rehospitalization from skilled nursing facilities. *Health Affairs (Millwood)*, 29(1), 57-64.
- Nagasako, E.M., Reidhead, M., Waterman, B., & Dunagan, W.C. (2014). Adding socioeconomic data to hospital readmissions calculations may produce more useful results. *Health Affairs*, 33(5), 786-791.
- National Quality Forum. Risk Adjustment for Socioeconomic Status or Other Sociodemographic Factors: Technical Report, August 15, 2014.
- Naylor, M.D., Brooten, D.A., Campbell, R.L., Maislin, G., McCauley, K.M., & Schwartz, J. S. (2004). Transitional care of older adults hospitalized with heart failure: a randomized, controlled trial. *Journal of the American Geriatrics Society*, 52(5), 675-684.
- Naylor, M.D., Aiken, L.H., Kurtzman, E.T., Olds, D.M., & Hirschman, K.B. (2011). The importance of transitional care in achieving health reform. *Health Affairs*, *30*(4) 746-754.
- Needleman, J., Buerhaus, P.I., Stewart, M., Zelevinsky, K., & Mattke, S. (2006). Nurse staffing in hospitals: is there a business case for quality? *Health Affairs*, 25(2), 204–211.



- Needleman, J., Buerhaus, P., Pankratz, S., Leibson, C.L., & Stevens, M.S. (2011). Nurse staffing and inpatient hospital mortality. *The New England Journal of Medicine*, *364*, 1037-45.
- Ong, K.L., Mowat, F.S., Chan, N., Lau, E., Halpern, M.T., & Kurtz, S.M. (2006). Economic burden of revision hip and knee arthroplasty in Medicare enrollees. *Clinical Orthopaedics and Related Research*, 446, 22-8.

Patient Protection and Affordable Care Act, 42 U.S.C. § 18001 et seq. (2010).

- Ponzetto, M., Maero, B., Maina, P., Rosato, R., Ciccone, G., Merlettie, F., et al. (2003). Risk factors for early and late mortality in hospitalized older patients: The continuing importance of functional status. *Journals of Gerontology Series A-Biological Sciences & Medical Sciences*, 58(11), 1049–1054.
- Pope, G.C., Ellis, R.P., Ash, A.S., Ayanian, J.Z., Bates, D.W., Burstin, H., ... Wu, B. (2000). Diagnostic Cost Group Hierarchical Condition Category Models for Medicare Risk Adjustment: Final Report. Retrieved from <u>http://www.cms.hhs.gov/Reports/downloads/pope_2000_2.pdf</u>
- Press, M.J., Scanlon, D.P., Ryan, A.M., Zhu, J., Navathe, A.S., Mittler, J.N., & Volpp, K.G. (2013). Limits of readmission rates in measuring hospital quality suggest the need for added metrics. *Health Affairs*, 32(6), 1083-1091.
- Pugely, A.J., Callaghan, J.J., Martin, C.T., Cram, P., & Gau, Y. (2013). Incidence of and risk factors for 30-day readmission following elective primary total joint arthroplasty: Analysis From the ACS-NSQIP. *The Journal of Arthroplasty*, 28(9), 1499-1504.
- QualityNet. (2014a) Hip/Knee Readmission and Complication Measures. Retrieved September 4, 2014 from <u>https://www.qualitynet.org/dcs/ContentServer?c=Page&pagename=QnetPublic%</u> <u>2FPage%2FQnetTier4&cid=1228772504368</u>
- QualityNet. (2014b). Resources: Readmission Measures. Retrieved September 15, 2014 from <u>http://www.qualitynet.org/dcs/ContentServer?c=Page&pagename=QnetPublic%2</u> FPage%2FQnetTier4&cid=1228772783162
- Rau, J. (2013, August 2). Armed with bigger fines, Medicare to punish 2,225 hospitals for excess readmissions. Retrieved August 14, 2014 from http://www.kaiserhealthnews.org/Stories/2013/August/02/readmission-penaltiesmedicare-hospitals-year-two.aspx
- Rau, J. (2014, October 2). Medicare Fines 2,610 Hospitals In Third Round Of Readmission Penalties. Retrieved September 26, 2014 from



http://www.kaiserhealthnews.org/Stories/2014/October/02/Medicare-readmissions-penalties-2015.aspx

- ResDAC. (2013). *MedPAR RIF*. Retrieved September 29, 2014 from <u>http://www.resdac.org/cms-data/files/medpar-rif</u>.
- Riggs, R.V., Roberts, P.S., Aronow, H., & Younan, T. (2010). Joint replacement and hip fracture readmission rates: impact of discharge destination. *American Journal of Physical Medicine & Rehabilitation*, 2(9), 806-810.
- Robert Wood Johnson Foundation. (2013). The Revolving Door. Retrieved June 1, 2014 from <u>http://www.rwjf.org/content/dam/farm/reports/2013/rwjf404178</u>
- Ryan, A.M., Burgess, J.F., Pesko, M.F., Borden, W.B. & Dimick, J.B. (2015). The early effects of Medicare's mandatory hospital pay-for-performance program. *Health Services Research*, *50*(1), 81-97.
- Satagopan, J.M., Ben-Porat, L., Berwick, M., Robson, M., Kutler, D., & Auerbach, A.D. (2004). A note on competing risks in survival data analysis. *British Journal of Cancer*, 91(7), 1229-1235.
- Saucedo, J.M., Marecek, G.S., Wanke, T.R., Lee, J., Stulberg, S.D., & Puri, L. (2014). Understanding readmission after primary total hip and knee arthroplasty: Who's at risk? *The Journal of Arthroplasty*, 29(2), 256-260.
- Schairer, W.W., Sing, D.C., Vail, T.P., & Bozic, K.J. (2014). Causes and frequency of unplanned hospital readmission after total hip arthroplasty. *Clinical Orthopaedics* and Related Research, 472(2), 464-470.
- Schairer, W.W., Vail, T.P., & Bozic, K.J. (2014). What are the rates and causes of hospital readmission after total knee arthroplasty? *Clinical Orthopaedics and Related Research*, 472(1), 181-187.
- Shekelle, P.G. (2013). Nurse-patient ratios as a patient safety strategy: a systematic review. *Annals of Internal Medicine*, *158*(5 Pt 2), 404-409.
- Singh, J.A., Jensen, M.R., Harmsen, W.S., Gabriel, S.E., & Lewallen, D.G. (2011). Cardiac and thromboembolic complications and mortality in patients undergoing total hip and total knee arthroplasty. *Annals of Rheumatic Diseases*, 70(12), 2082-2088.
- Smith, H.L. (2008). A double sample to minimize bias due to non-response in a mail survey. In A. Ruiz-Gazen, P. Guilbert, D. Haziza, & Y. Tille (Eds.), Survey Methods: Applications to longitudinal studies, to health, to electoral studies and to studies in developing countries (334-339). Paris: Dunod.



- StataCorp. 2013. Stata Statistical Software: Release 13. College Station, TX: StataCorp LP.
- Strauss, M.E., Sirotkin, R.A., & Grisell, J. (1974). Length of hospitalization and rate of readmission of paranoid and nonparanoid schizophrenics. *Journal of Consulting* and Clinical Psychology, 42(1), 105-110.
- Suter, L.G., Parzynski, C.S., Searfoss, R., Dorsey, K.B., Grady, J.N., Keenan, M., ... Bernheim, S.M. 2014 Procedure-Specific Readmission Measures Updates and Specifications Report Elective Primary Total Hip Arthroplasty (THA) and/or Total Knee Arthroplasty (TKA) (Version 3.0). Yale New Haven Health Services Corporation/Center for Outcomes Research & Evaluation (YNHHSC/CORE), March 2014.
- Tourangeau, A.E., Cranley, L.A., & Jeffs, L. (2006). Impact of nursing on hospital patient mortality: a focused review and related policy implications. [Comparative Study Research Support, Non-U.S. Gov't Review]. Quality & Safety in Health Care, 15(1), 4-8.
- Tsai, T.C., Joynt, K.E., Orav, J., Gawande, A.A., & Jha, A.K. (2013). Variation in surgical-readmission rates and quality of hospital care. *The New England Journal of Medicine*, *369*, 1134-1142.
- United States Census Bureau (n.d.). *State and county quick facts*. Retrieved October 1, 2014, from <u>http://quickfacts.census.gov/qfd/index.html</u>
- Vaduganathan, M., Bonow, R.O., & Gheorghiade, M. (2013). Thirty-day readmissions: the clock is ticking. *Journal of the American Medical Association*, 309(4), 345-346.
- Vittinghoff, E., Glidden, D.V., Shiboski, S.C., & McCulloch, C.E. (2012). Regression methods in biostatistics (2nd Ed.). New York, NY: Springer.
- Vorhies, J.S., Wang, Y., Herndon, J., Maloney, W.J., & Huddleston, J.I. (2011). Readmission and length of stay after total hip arthroplasty in a national Medicare sample. *Journal of Arthroplasty*, 26(6 Suppl), 119-123.
- Weiss, M.E, Yakusheva, O., & Bobay, K.L. (2011). Quality and cost analysis of nurse staffing, discharge preparation, and postdischarge utilization. *Health Services Research*, 46(5), 1473-1494.
- Weissman, J.S., Stern, R.S., & Epstein, A.M. (1994). The impact of patient socioeconomic status and other social factors on readmission: a prospective study in four Massachusetts hospitals. *Inquiry*, *31*(2), 163-172.



- Werner, R.M., Asch, D.A., & Polsky, D. (2005). Racial profiling: the unintended consequences of coronary artery bypass graft report cards. *Circulation*, 111(10), 1257-1263.
- Werner, R. & Dudley, R.A. (2012). Medicare's new hospital value-based purchasing program is likely to have only a small impact on hospital payments. *Health Affairs*, 31(9), 1932-1940.
- Werner, R.M., Goldman, L.E., & Dudley, R.A. (2008). Comparison of change in quality of care between safety-net and non-safety-net hospitals. *Journal of the American Medical Association*, 299(18), 2180–2187.
- Werner, R., Kolstad, J.T., Stuart, E.A., & Polsky, D. (2011). The effect of pay-forperformance in hospitals: lessons for quality improvement. *Health Affairs*, *30*(4), 690-698.
- Wheeler, E.C., & Waterhouse, J.K. (2006). Telephone interventions by nursing students: improving outcomes for heart failure patients in the community. *Journal of Community Health Nursing*, 23(3), 137-146.
- Whittle, J., Steinberg, E.P., Anderson, G.F., Herbert, R., & Hochberg, M.C. (1993). Mortality after elective total hip arthroplasty in elderly Americans. Age, gender, and indication for surgery predict survival. *Clinical Orthopaedics and Related Research*, 295, 119–126.
- Wiltse Nicely, K.L., Sloane, D.M., & Aiken, L.H. (2012). Lower mortality for abdominal aortic aneurysm repair in high-volume hospitals is contingent upon nurse staffing, *Health Services Research*, 48(3) 972-991.
- Wing, J.K., Denham, J., & Munro, A.B. (1959). Duration of stay in hospital of patients suffering from schizophrenia. *British Journal of Preventative & Social Medicine*, 13(3), 145-148.
- Woodend, A.K., Sherrard, H., Fraser, M., Stuewe, L., Cheung, T., & Struthers, C. (2003). Telehome monitoring in patients with cardiac disease who are at high risk of readmission. *Heart and Lung*, 37(1), 36-45.
- Yakusheva, O., Lindrooth, R., & Weiss, M. (2014a). Economic evaluation of the 80% baccalaureate nurse workforce recommendation: a patient-level analysis. *Medical Care*, *52*(10), 864–869.
- Yakusheva, O., Lindrooth, R., & Weiss, M. (2014b). Nurse value-added and patient outcomes in acute care. *Health Services Research*, 49(6) 1767-1786.
- Zmistowski, B., Restrepo, C., Hess, J., Adibi, D., Cangoz, S., & Parvizi, J. (2013). Unplanned readmission after total joint arthroplasty: rates, reasons, and risk



factors. *Journal of Bone and Joint Surgery - American Volume*, 95(20), 1869-1876.

